•

CWI Tracts

Managing Editors

. .

K.R. Apt (CWI, Amsterdam) M. Hazewinkel (CWI, Amsterdam) J.K. Lenstra (Eindhoven University of Technology)

Editorial Board

W. Albers (Enschede)
P.C. Baayen (Amsterdam)
R.C. Backhouse (Eindhoven)
E.M. de Jager (Amsterdam)
M.A. Kaashoek (Amsterdam)
M.S. Keane (Delft)
H. Kwakernaak (Enschede)
J. van Leeuwen (Utrecht)
P.W.H. Lemmens (Utrecht)
M. van der Put (Groningen)
M. Rem (Eindhoven)
H.J. Sips (Delft)
M.N. Spijker (Leiden)
H.C. Tijms (Amsterdam)

CWI

P.O. Box 94079, 1090 GB Amsterdam, The Netherlands Telephone 31 - 20 592 9333, telex 12571 (mactr nl), telefax 31 - 20 592 4199

CWI is the nationally funded Dutch institute for research in Mathematics and Computer Science.

Sojourn times in feedback and processor sharing queues

.

J.L. van den Berg

1991 Mathematics Subject Classification: 60K25, 90B22. ISBN 90 6196 427 X NUGI-code: 811

.

Copyright $\ensuremath{\mathbb{C}}$ 1993, Stichting Mathematisch Centrum, Amsterdam Printed in the Netherlands

TABLE OF CONTENTS

1 INTRODUCTION

.

1.1	Background	1
1.2	Queueing models with feedback	3
1.3	Time sharing models; round robin and processor sharing queues	5
1.4	Models with permanent customers	8
1.5	Review of related literature	10
	1.5.1 Feedback queues	10
	1.5.2 Time sharing and processor sharing queues	11
1.6	Assumptions and notations	14
1.7	Overview of the contents of the next chapters	14

2 THE M/M/1 QUEUE WITH GENERAL FEEDBACK

2.1	Introduction	17
2.2	Model description	18
2.3	Main results	20
2.4	Sojourn time characteristics	25
2.5	Special cases: Bernoulli feedback and deterministic feedback	28
	2.5.1 Bernoulli feedback	28
	2.5.2 Deterministic feedback	31
2.6	Further extensions	33
	2.6.1 Derivation of a set of linear equations	33
	2.6.2 Special case: $M/G/1$ queue with deterministic feedback	35
	2.6.3 Special case: M/G/1 queue with Bernoulli feedback	36
App	pendix	- 39

3 THE M/G/1 PROCESSOR SHARING QUEUE AS A LIMITING MODEL OF THE M/M/1 FEEDBACK QUEUE

3.1	Introduction	42
3.2	Preliminary results	43

3.3	The M/G/1 processor sharing queue	46
	3.3.1 The limiting procedure	46
	3.3.2 The mean sojourn time	48
	3.3.3 The variance of the sojourn time	48
	3.3.4 The distribution of the sojourn time	53
3.4	The M/G/1 queue with generalized processor sharing	61
	3.4.1 Introduction	61
	3.4.2 Analysis	62
3.5	The M/G/1 processor sharing queue with feedback	65
	3.5.1 Introduction	65
	3.5.2 Model description and notations	65
	3.5.3 Analysis	67

.

4 SIMPLE APPROXIMATIONS FOR SECOND MOMENT CHA-RACTERISTICS OF THE SOJOURN TIME IN THE M/G/1 PROCESSOR SHARING QUEUE

4.1	Introduction	71
4.2	Notations and preliminary results	72
4.3	Approximation of $E{S^2(x)}$	76
4.4	Approximation of $E\{\mathbf{S}^2\}$	77
	4.4.1 Simple approximation	77
	4.4.2 Detailed approximation	78
4.5	Conclusions	82
Tat	bles	83

5 QUEUEING MODELS WITH ADDITIONAL PERMANENT CUSTOMERS

5.1	Introduction	90
5.2	Model description and notations	92
5.3	Mean queue lengths and sojourn times	93
5.4	Distributions of queue lengths and sojourn times	96
	5.4.1 The case $K = 1$	96
	5.4.2 The case $K \ge 1$	100

	5.4.3 Results for some special choices of the service time distribu- tions	104
	5.4.4 Generalizations; the M/G/1 queue with Bernoulli feedback and additional permanent customers	106
5.5	Relation with cyclic service models	109
5.6	The M/M/1 feedback queue with additional permanent customers	110
	5.6.1 Introduction	110
	5.6.2 Sojourn time distribution	111
	5.6.3 The $M/G/1$ PS queue with additional permanent customers	113
RE	FERENCES	117

. .

.

SUBJECT INDEX 122

.

Chapter 1

1

INTRODUCTION

1.1 BACKGROUND

Queueing theory is concerned with the mathematical analysis of the performance of systems that offer services for collective use, like telephone exchanges and computer networks. Due to the finite capacity of such systems queueing arises in many practical situations when too many users require access to the same facility at the same time. The mathematical study of queueing phenomena started with the investigation of telephone call congestion and delay about the beginning of this century. Later, queueing theory was successfully applied in operations research and management science, in particular for production planning; in the past decades it has become an almost indispensable tool for the performance prediction of complex computer communication systems.

A queueing model is usually described in terms of customers requiring service, service facilities providing service, and queues containing customers waiting for service. The present study is devoted to the analysis of queueing models where customers may repeatedly return to some service facility to obtain several phases of service before they finally depart from the system. Such *feedback* phenomena occur in a wide variety of processes arising in computer-communication and in production networks.

The basic queueing model representing the occurrence of feedback consists of a waiting room and a single service facility at which customers arrive according to a stochastic process; after having obtained a random amount of service a customer either returns to the queue of waiting customers to await another service or leaves the system, according to a probabilistic feedback scheme.

An example that illustrates the feedback phenomenon is found in manufacturing processes where quality control inspections are performed after the execution of an operation on a part or product, see Fig. 1.1. A part that does not meet the quality standards is sent back for reworking; this may happen several times until it finally passes the test and proceeds to the next phase of operation.



2

Fig. 1.1 Quality control of parts during a production process.

Another important example of the occurrence of feedback is encountered in a computer system which operates in a *time sharing* mode. In Fig. 1.2 a scheme of such a time sharing system is shown. In a time shared computer system each job is allocated a small time interval for uninterrupted processing at the CPU. If the total required processing time of a job exceeds the length of this interval it is fed back to a system of queues containing waiting jobs; here the job waits until it is permitted a second turn in the processing facility, according to a certain scheduling algorithm. This procedure is repeated until the job has obtained its required processing time and leaves the system.



Fig. 1.2 Principle of feedback in a time shared computer system.

The introduction of time sharing systems in the early sixties and the need to determine their performance has led to an extensive study of queueing models with feedback. On the other hand, theoretical investigations concerning networks of queues have also stimulated research on feedback queues. The research of J.R. Jackson (Jackson [1957,1963]) on queueing networks with exponential services and independent external Poisson arrival processes revealed that, under certain assumptions, a queue in such a network in steady state behaves just like an M/M/s queue in isolation. In the case of so called *feedforward* networks where customers never return to a queue they have once

left, one can indeed prove that the input process to each queue is a Poisson process. But when feedback is possible the input process is no longer a Poisson process - which makes the M/M/s behaviour all the more surprising. These observations revived the interest in single server queueing models with feedback of customers, like an M/M/1 queue with constant feedback probability. In particular, the stochastic aspects of customer flows in feedback queues were extensively studied (cf. the survey by Disney and König [1985]). Furthermore, the steady state queue length processes in Jackson networks and their generalizations, the so called BCMP networks (see Baskett et al. [1975]), appeared to be amenable to a detailed analysis; however, the analysis of other important characteristics such as the waiting time and sojourn time processes presented quite some difficulties. Again the case of feedforward Jackson networks was relatively simple - as long as each node contains only a single server; in this case the joint steady state distribution of the successive sojourn times of a particular customer could explicitly be obtained, cf. Lemoine [1979] and Walrand and Varaiya [1980]. But it readily became clear that the possibility of customers overtaking one another introduced considerable analytical difficulties. Once more the M/M/1 queue with feedback provided the simplest example to study this 'overtaking' phenomenon in isolation.

The aim of the present study is the analysis of sojourn times in single server queueing models with feedback: we shall derive the joint steady state distribution of the successive sojourn times of a customer in a feedback queue with a quite general feedback mechanism. As an important by-product, our study on feedback queues leads to new insights in the analysis of the well-known and widely used 'processor sharing' model for time sharing systems.

The remainder of this chapter is organized as follows. In Section 1.2 we shall first describe a basic feedback queueing model and discuss its main properties. Next, the central feedback model of this study is introduced. Time sharing systems and models are discussed in Section 1.3. In Section 1.4 we introduce an interesting variant of the 'standard' M/G/1 queueing model, viz. an M/G/1 queue with a fixed number of additional *permanent* customers; that are customers who reside permanently in the system, i.e. they are fed back after each service. Section 1.5 contains an extensive overview of the literature related with the models considered in the present study. Section 1.6 is concerned with assumptions about the notations and terminology used in this monograph. Finally, in Section 1.7 we give an overview of the contents of Chapters 2-5.

1.2 QUEUEING MODELS WITH FEEDBACK

The basic, in literature most frequently encountered, feedback queueing model is the M/G/1 queue with 'Bernoulli' feedback, see Fig. 1.3. The behaviour of the customers in this model is as follows. New customers, arriving according to a Poisson process, join the end of the queue. Immediately after his service



Δ

Fig. 1.3 The M/G/1 queue with Bernoulli feedback.

completion a customer returns to the end of the queue with probability p or leaves the system with probability 1-p, $0 \le p < 1$. It is assumed that all service times are independent, identically distributed, random variables. The customers in the queue are served according to the 'head-of-the-line' discipline.

It is readily seen that the M/G/1 queue with Bernoulli feedback has a stationary queue length process which has the same distribution as an equivalent M/G/1 queue without feedback, i.e. an M/G/1 system in which the service time distribution of a customer is equal to that of the total service time a customer obtains in the feedback model. Indeed, because the feedback probabilities are constant the queue length distribution is *independent* of the *order* in which the customers are served; so we may assume that they are served in one stretch with a service time equal to the total service time that they would have if they were served in the original manner.

A much more difficult task is the determination of the distribution of the total sojourn time of a customer. The problem is caused by the fact that the total sojourn time of a particular (tagged) customer is the sum of the (partial) sojourn times during his successive passes through the system, which are clearly not independent of each other. Moreover, for the analysis of the sojourn time of a tagged customer one has to take into account that new customers may arrive during the presence of the tagged customer and reside in the system during some passes (note that their services contribute to the tagged customer's total sojourn time), but leave the system before the tagged customer. The possibility that customers can overtake each other leads to dependencies which almost invariably makes the determination of the sojourn time distribution impossible (see e.g. the survey of Boxma and Daduna [1989] on sojourn times in queueing networks). For the M/G/1 queue with Bernoulli feedback, however, the sojourn time problem could be solved, see Takács [1963]. Takács obtained a recurrence relation for the (Laplace-Stieltjes transform and generating function of the) joint distribution of a tagged customer's total sojourn time and the number of customers present in the system after k services, k = 1, 2, ... The derivation is based on the observation that for a tagged customer the joint distribution of the duration of his (i + 1)-

th pass through the system (also called his (i + 1)-th sojourn time) and the number of customers present at the end of his (i + 1)-th service is completely determined by his *i*-th sojourn time and the number of customers present at the end of his *i*-th service, i = 1, 2, ... (i.e. the joint process of successive service completion epochs and queue length at these epochs is a *Markov renewal* process). In fact, this observation is the basis of the analysis of most of the feedback models discussed in the present study.

The feedback model investigated in this monograph is actually a generalization of the M/M/1 queue with Bernoulli feedback. It is an M/M/1 feedback model in which the probability that a customer is fed back after service completion depends on the number of times he has already been served: when a customer completes his *i*-th service he departs from the system with probability 1-p(i) and he recycles with probability p(i), i=1,2,... Obviously, taking $p(i)\equiv p$ this model reduces to the M/M/1 queue with Bernoulli feedback. In the sequel a customer who is visiting the queue for the *i*-th time will be called a 'type-*i* customer', i=1,2,...

It is important to note that the M/M/1 queue with general feedback as described above belongs to the well-known class of so called 'product form' networks (see e.g. Baskett et al. [1975] and Kelly [1979]), i.e. the (stationary) finite dimensional joint queue length distribution of the different types of customers is known and has a product form. It is noted here that due to the general feedback mechanism the distribution of the total number of customers in the system is *not* the same as the queue length distribution in the standard M/G/1 queue with service times equal to the total service time in the feedback queue, as is the case for the M/G/1 queue with Bernoulli feedback.

The main result of the present study of the M/M/1 queue with general feedback is a complete description of the *joint* distribution of the successive sojourn times of a particular customer.

1.3 TIME SHARING MODELS; ROUND ROBIN AND PROCESSOR SHARING QUEUES In Section 1.1 we already described the principle of time sharing in computer systems. Actually, the motivation for the introduction of time sharing computers has been to provide multiple users simultaneous and (almost) direct access to a single processor unit. In fact this is achieved by giving small jobs (for which the users expect small response times) preferential treatment at the expense of the longer ones (for which the users expect larger response times). It is desirable that this property is reflected by queueing models of time shared systems. Accordingly the performance measure most often used for time sharing models is the response time for a job *conditional* on its required service time. We shall discuss this performance measure for the time sharing model described below.

The M/G/1 queue with the so called 'round robin' (RR) service discipline is

the most frequently encountered queueing model for the time shared computer systems as described in Section 1.1. We have pictured this single queue model in Fig. 1.4.



Fig. 1.4 The M/G/1 queue with round robin service.

The customers are served as follows. New customers, arriving according to a Poisson process, join the end of the queue. The customers in the queue are served according to the head-of-the-line discipline receiving a (small) fixed quantum q of service. At the end of his service quantum a customer leaves the system if his total service requirement is met; if not he returns to the end of the queue with his remaining service requirement reduced by an amount q.

To overcome the mathematical problems that arise from the analysis of the RR model with fixed positive quantum size q it is often assumed that $q \rightarrow 0$ (an idea originally due to Kleinrock [1967]). RR models under the assumption $q \rightarrow 0$ are called *processor sharing* (PS) models and are of great interest; they have pleasing mathematical properties and they also accurately model the performance of many real systems.

In queueing literature the PS service discipline is often described as follows: when there are $n \ge 1$ customers present in the system then each customer receives service at a rate which is 1/n times the rate of service that a solitary customer in the system would receive. It is easily seen that this alternative description coincides with the original definition of PS as the limiting case of the RR service discipline.

The M/G/1 PS queue has some very interesting mathematical properties. First, the mean conditional sojourn time $ES^{PS}(x)$ of a customer with service demand $x \ge 0$ is *linear* in x and depends only on the first moment of the service time distribution: (see Kleinrock [1967], Sakata et al. [1971])

$$E\mathbf{S}^{PS}(x)=\frac{x}{1-\rho}\,,$$

where ρ denotes the offered load to the system. This formula shows in which

way the PS discipline provides preferential treatment to short jobs (customers): a job half as long as an other will spend on the average half as long in the system. The above formula for the mean conditional sojourn time in the M/G/1 PS queue should be compared with the result for the corresponding quantity in the M/G/1 first-come-first-served (FCFS) queue:

$$E\mathbf{S}^{FCFS}(x) = \frac{\lambda\beta_2}{2(1-\rho)} + x ,$$

with λ and β_2 denoting the arrival intensity and the second moment of the service time distribution, respectively.

Another important property of the M/G/1 PS queue is that the (stationary) queue length distribution is 'insensitive' to the character of the service time distribution, apart from its first moment:

$$Pr\{k \text{ customers in the system}\} = (1-\rho)\rho^k, \quad k=0,1,\dots$$

This insensitivity property also holds for the joint queue length distribution in networks of PS queues (cf. Baskett et al. [1975], Kelly [1979]).

The usefulness of the PS service discipline for modeling the performance of computer systems, and its mathematical properties, have strongly contributed to the extensive use of PS (network) models in present day performance analysis.

A very difficult mathematical problem for PS models is the determination of the sojourn time *distribution*. The difficulties are caused by the same phenomenon as occurring in the analysis of sojourn times in feedback queues: the PS service discipline allows customers to overtake each other. A solution of the sojourn time problem for the M/G/1 PS queue was first obtained by Yashkov [1983]. However, the derivation of his results is complicated and does not provide much insight into the behaviour of the main sojourn time characteristics.

In this monograph we present a new, more transparent method for the derivation of the (Laplace-Stieltjes transform of the) distribution of the sojourn time in the M/G/1 PS queue. The idea is to consider the M/G/1 PS queue as a limiting case of the M/M/1 queue with general feedback (see Section 1.2). The PS model is obtained by letting the feedback probabilities approach one and the mean service time at each loop approach zero, such that a customer's total required mean service time remains constant. Different choices of the feedback probabilities lead to different service time distributions in the PS queue. Application of this limiting procedure to the sojourn time results obtained for the M/M/1 feedback queue leads to results for the corresponding quantities in the PS queue.

The method described above exploits well-known product form results for

the feedback model and gives much insight into the occurrence of many basic sojourn time properties for the limiting PS model.

Many generalizations and variants of the PS (round robin) service discipline have been introduced, see e.g. the surveys by Jaiswal [1982] and Yashkov [1987]. For most of them the sojourn time distribution problem (even for the simplest M/M/1 case) is still unsolved. For one generalization, called 'generalized processor sharing' (GPS), we shall present in this monograph some new sojourn time results. Therefore, the GPS model will be discussed here in some more detail.

The GPS discipline generalizes the PS discipline as follows: when there are $n \ge 1$ customers present in the system then each customer is served with a rate equal to f(n) with $f(\cdot)$ an (almost) arbitrary positive function. Obviously, for f(n)=1/n the GPS discipline reduces to the PS discipline (assuming that the total capacity of the server is normalized to one).

Network models of GPS queues contain many interesting special cases such as the classical Erlang and Engset systems as well as many new processor sharing systems, see Cohen [1979]. The GPS discipline generalizes known results for classical networks: it preserves the product form and insensitivity property of the joint distribution of the queue lengths at the different nodes. At present most sojourn time results for GPS models are limited to the mean conditional sojourn time of a customer with given service demand. In general, sojourn time distributions are still unknown.

In this monograph we propose a new approach to the analysis of GPS queues. The idea is similar to that for the PS case: we consider the M/G/1 GPS queue as a limiting case of the M/M/1 queue with general feedback introduced in Section 1.2 but with *state dependent service rates*. Different choices of the service rates in the feedback model lead to different service rate functions $f(\cdot)$ in the GPS queue. We show that (known) results for the M/G/1 GPS queue can be very easily obtained from the analysis of this (product form) feedback model with state dependent service rates. (In fact, the analysis of the single node GPS model can be easily extended to the analysis of networks of GPS queues). For a special class of GPS disciplines this approach leads to new results for the sojourn time distribution.

1.4 MODELS WITH PERMANENT CUSTOMERS

An extreme case of a feedback queue is a *closed* queueing system, i.e. a queueing system with a fixed number of '*permanent*' customers. Closed queueing systems model the situation where the number of customers in the system is constant (once a customer has obtained his required service he is immediately replaced by another one with the same characteristics). A queue with additional permanent customers is a system where next to the ordinary customer stream(s) also permanent customers are processed. An interesting aspect in such a case is the interference of permanent customers with the other customer

streams, in particular the influence of the presence of the permanent customers on the queueing characteristics of the other customers. In the last chapter of this monograph, Chapter 5, we consider some single queue single server models with two types of customers: (i) ordinary customers who arrive according to a Poisson process, and (ii) a fixed number of permanent customers who immediately return to the end of the queue after having received a service. See Fig. 1.5 for the case of an M/G/1 queue with permanent customers. Our main goal is to present a study of the influence of the presence of additional permanent customers on queue lengths and sojourn times of the 'Poisson customers' for the standard M/G/1 queue and for the feedback and PS models discussed in the previous sections.



Fig. 1.5 The M/G/1 queue with additional permanent customers.

The main reason for studying these relatively simple models with permanent customers is that these models expose - stripped from all non-essential features - a structure that appears in many representations of computer and communication networks. For example, consider a telephone exchange to which two types of jobs are offered: call requests and operator tasks. To guarantee a certain quality of service to the call requests only a limited number, say K, of operator tasks (which are assumed to be always available) is allowed to be in the system at the same time. Obviously, it is important to know in which way the choice of the control parameter K influences the performance of the system with respect to the waiting times of the call requests and the throughput of the operator tasks.

Another reason for studying models with permanent customers is that there are several interesting relations with other important queueing models. For example, the M/G/1 queue with one additional permanent customer behaves exactly like a *vacation* queue, a queueing model where the server interrupts the service to a customer stream at certain epochs to take a vacation. Other related models are discussed in Chapter 5.

1.5 REVIEW OF RELATED LITERATURE

In this section we discuss some literature related to this study and indicate the place of the study within the literature. We restrict ourself mainly to literature concerning single server models where the (external) arrival process is a Poisson process. Work on multi resource systems (networks) and finite source models is not discussed.

1.5.1 Feedback queues

A pioneering study on feedback systems is Takács [1963]. He considered the M/G/1 FCFS queue with Bernoulli feedback. His main result is a recurrence relation for the Laplace-Stieltjes transform (LST) and generating function of the joint distribution of a customer's total sojourn time and the number of customers present in the system after k services, k = 1, 2, ... The key observation leading to this result is that for a tagged customer the joint process of successive service completion epochs and queue length at these epochs is a Markov renewal process, see Section 1.2. In fact, this observation is the basis of the analysis in most of the feedback studies discussed below.

Disney [1981], Disney et al. [1984] and Doshi and Kaufman [1988] have also studied queue length and sojourn time distributions in the M/G/1 Bernoulli feedback queue in some detail. In particular Doshi and Kaufman derive the LST of the joint distribution of the sojourn times of a customer on his successive passes through the system. Disney et al. [1980] is mainly concerned with a fundamental study of several traffic flow processes in the system and queue length distributions at different embedded stochastic epochs, see also Disney and König [1985]. They show for the M/M/1 queue with Bernoulli feedback that the input process (the successive epochs at which a customer (re)enters the queue) and the output process (the successive service completion epochs) are Markov renewal processes; for general service times the output process is also Markov renewal. It is shown that for positive feedback probabilities these processes are never renewal. Disney and König [1985] also present an overview of literature concerned with the analysis of Bernoulli feedback models.

Fontana and Diaz Berzosa [1984,1985] extend some results obtained for the M/G/1 model with Bernoulli feedback to a more general feedback model with non preemptive priorities. However, one should take care in applying their results because some of them do not agree with those in Disney et al. [1980] (e.g. Fontana and Diaz Berzosa [1984,1985] erroneously conclude that for the M/G/1 Bernoulli feedback queue the queue length distribution at output and arbitrary epochs are equal; Disney et al. [1980] prove that this does *not* hold).

Simon [1984] analyzes an M/G/1 feedback queue with multiple customer types and preemptive and non preemptive priority levels that may change after a service completion; the customers are fed back a fixed number of times. The main result of his paper is the derivation of a set of linear equations for the mean sojourn time of each visit.

The feedback model studied by Lam and Shankar [1981] is basically the same as the M/M/1 queue with general feedback analyzed in the present

study. They consider the model as a time sharing model with exponentially distributed service quanta. Lam and Shankar derive the total sojourn time distribution. This distribution is a special case of our result for the joint distribution of the sojourn times of a customer on his successive cycles. Nelson [1987] also considers queues with general feedback, but with varying service times, to study the effect that assigning increasing service times to customers has on mean sojourn times.

Ali and Neuts [1984] consider an M/G/1 Bernoulli feedback queue with a waiting room and a service room. Newly arriving customers and customers who have been fed back join the waiting room. Whenever the service room becomes empty all customers from the waiting room, together with a random number of overhead customers, are transferred to it. Ali and Neuts determine the stationary distribution of queue lengths at various embedded random epochs and the distribution of a customer's waiting time until his first service.

Hunter [1989] considers single server queues with state dependent feedback and finite waiting room. In particular, he studies an appropriately constructed Markov renewal process which describes the behaviour of the system starting at the arrival of a tagged customer; the sojourn time of the tagged customer relates to a first passage time in this process. For some special cases (e.g. the M/M/1/2 queue with Bernoulli feedback) this approach leads to the derivation of explicit expressions for the LST of the distribution of the total sojourn time. Mean sojourn times are obtained for the M/M/1/N ($N \ge 1$) queue with Bernoulli feedback. Hunter also gives a brief survey of the literature on sojourn times in feedback models.

1.5.2 Time sharing and processor sharing queues

The first queueing model for a time shared computer system was presented by Kleinrock [1964]. He studied a single queue single server system under the round robin discipline as a discrete time Markovian model. Since then until the early seventies many papers on (variants of) this model have been published, see e.g. Schrage [1967], Kleinrock [1967], Coffman and Kleinrock [1968], Adiri and Avi-Itzhak [1969A,1969B], Sakata et al. [1971], Adiri [1972] and Muntz [1972]. All these papers are concerned with the derivation of *mean* queue lengths and sojourn time distribution for an M/M/1 queue under the RR discipline with fixed quantum size and overhead due to switching between customers. An extensive discussion on time sharing systems and models, with many references, is given by Kleinrock [1976, Chapter 4] and Jaiswal [1982].

To simplify the mathematical analysis of RR models Kleinrock [1967] proposed to study the case that the size of the service quanta shrinks to zero, thus obtaining the *processor sharing* discipline. In particular the derivation of queue length distributions and mean (conditional) sojourn times appeared to be much easier for the PS model than for the corresponding RR system with positive service quanta. Kleinrock [1967] showed that for the M/M/1 PS queue the

mean conditional sojourn time is linear in the required service time (cf. the formula in Section 1.3). Sakata et al. [1971] obtained the same result for the case of general service times by letting in their RR model the quantum size tend to zero. O'Donovan [1974] and Asare and Foster [1983] derived the mean conditional sojourn time directly from the behaviour of the M/G/1 PS queue. A very recent paper on this subject is Foley and Klutke [1989]. Foley and Klutke show that the (non stationary) queue length process starting at the arrival of a tagged customer is stochastically strictly increasing during the presence of the tagged customer, which makes it quite surprising that the mean conditional sojourn time is a linear function of the required service time. Their approach, based on introducing different time scales for different processes, provides insight into this somewhat paradoxical property. In this monograph we shall give an explanation of it which is based on our 'feedback approach' to PS queues.

The derivation of the insensitivity property of the distribution of the queue length in the M/G/1 PS queue (given in Section 1.3) is originally due to Sakata et al. [1969]. In fact, they obtain their result as a special case of the queue length distribution in a multi server processor sharing model.

In contrast to the derivation of queue length distributions it appeared to be much more difficult to derive sojourn time distributions for PS queues. Coffman et al. [1970] obtained the distribution of the conditional sojourn time for the M/M/1 PS queue; for general service times it remained an unsolved problem until the early eighties. The LST of the distribution of the conditional sojourn time for the M/G/1 PS queue was obtained by Yashkov [1983], Ott [1984] and Schassberger [1984]. The approaches used by Yashkov and Ott are quite similar. The essence is a decomposition of the sojourn time of a (tagged) customer as the sum of 'time delays' which are induced by the customers present in the system at the arrival of the tagged customer (and by the tagged customer himself); these time delays include the influence of customers who arrive *during* the sojourn time of the tagged customer. It is shown that the time delays can be interpreted as lifetimes of some terminating branching process. The dynamics of the time delays is described by some integrodifferential equations derived by using ideas from branching theory. Schassberger [1984] obtained his result by means of the analysis of a discrete time queue under a slight variation of the standard RR discipline in which a newly arriving customer immediately receives a quantum of service and only then joins the tail of the queue. Using known sojourn time results for this RR model (obtained in Schassberger [1981]) and letting the quantum size shrink to zero he finds results for the corresponding sojourn times in the M/G/1 PS queue. Schassberger [1984] also gives the theoretical background of the weak convergence of the sojourn time distribution for the discrete time RR model to the distribution of the sojourn time in the PS queue.

The analysis of the sojourn time distribution in the M/G/1 PS queue proposed in this monograph is in some sense similar to the method used by Schassberger [1984]. We first analyze a kind of RR (feedback) queue with exponentially distributed service quanta and then take appropriate limits such

that the behaviour of the system approaches that of the M/G/1 PS queue. The advantage of this method is that it exploits well-known product form results for the feedback model which give much insight into the behaviour of the sojourn time in the limiting PS model. The proof that the distribution of the sojourn time in the feedback model converges weakly to that in the PS model has been given recently by Resing et al. [1989].

From the results obtained by Yashkov [1983], Ott [1984] and Schassberger [1984] expressions for the second and higher moments of the sojourn time distribution can be derived. The resulting formulas are very complex and their (numerical) evaluation requires quite some effort. As far as we know no attention has been paid in the processor sharing literature to the derivation of approximations or asymptotic formulas which are useful for practical evaluation, apart from a paper by Yashkov [1986]. He derived some asymptotic estimates for the conditional sojourn time variance for customers with very small or very large service demands. In this monograph simple approximations for the second moment of the sojourn time in the M/G/1 PS queue are presented. The derivations of these approximations are mainly based on new asymptotic results (e.g. heavy traffic and an extension of Yashkov's results) and on exact expressions for specific service time distributions.

A fundamental study of the generalized processor sharing service discipline is given by Cohen [1979]. Cohen studies the class of GPS disciplines in a very general model of closed and open networks with multiple customer types. This model contains as special cases the classical Erlang and Engset systems, the multi server M/G/s PS queue as well as many new PS systems. He obtains generalizations of known results for classical networks such as the product form and insensitivity property of the joint distribution of the queue lengths at the nodes and the mean conditional sojourn time of a customer with given service demand; sojourn time distributions are not studied.

In the present study it is shown that most of Cohen's GPS results can be obtained by using an approach similar to the one used for the analysis of the PS case, see the discussion at the end of Section 1.3. In addition, we shall derive the LST of the distribution of the sojourn time in the M/G/1 GPS queue for a special class of GPS disciplines.

Next to the GPS discipline there exist many other generalizations and variants of the PS discipline. For an overview of the various models and a discussion of the results we refer to the surveys by Jaiswal [1982] and Yashkov [1987]. Recently some studies on two interesting variants of the M/G/1 PS queue appeared, which are not covered by Jaiswal [1982] and Yashkov [1987]. Avi-Itzhak and Halfin [1988] consider an M/M/1 PS queue with a limited number (r) of service positions and preemption when there are at least r customers in the system upon the arrival of a new customer. (The case without preemption is treated in Avi-Itzhak and Halfin [1989A]). They present methods for calculating the LST and the moments of the (conditional) sojourn time distribution; it is shown that for $r \rightarrow \infty$ the results coincide with the results obtained by Coffman et al. [1970]. In Rege and Sengupta [1989] a 'gated' M/M/1 PS queue is studied. In this model a gate controlling the access to the service facility opens when the server becomes idle, admitting at most $m \ge 1$ waiting customers, and then closes again. Here, the case m = 1 corresponds to the M/M/1 FCFS queue. (Avi-Itzhak and Halfin [1989B] consider the case $m = \infty$ for general service times). The authors derive the LST of the waiting time distribution, the LST of the distribution of the time in service conditional on the required amount of service and the mean conditional sojourn time. Both PS variants can be used for modeling the performance of certain multiprogrammed (time shared) computer systems in which the degree of multiprogramming is limited to some maximum due to the constraints of finite memory.

1.6 Assumptions and notations

Throughout this study we assume that all the systems considered are stable (in statistical equilibrium), i.e. we assume that all involved stochastic processes (e.g. the queue length process and the sojourn time process) are stationary. For the systems studied in this monograph necessary and sufficient conditions for stability are well-known or else can be easily obtained from existing results for related models. So, when we refer to the 'sojourn time of an arbitrary customer' then we mean an independent copy of the sojourn time of the *n*th (newly) arriving customer, for some $n \ge 1$. Similarly, the 'queue length at an arbitrary epoch' refers to an independent copy of the queue length at some time *t*.

Throughout, random variables representing service times are indicated by bold printed Greek letters tau (τ); all other random variables are indicated by capitals also printed in bold type. Sections, formulas, theorems, figures, etc. are referred to by a numeral indicating the chapter in which they originally occur followed by their number within that chapter.

1.7 Overview of the contents of the next chapters

Chapter 2 is concerned with a fundamental analysis of sojourn times in the M/M/1 queue with general feedback. We first derive, in the form of Laplace-Stieltjes transforms and generating functions, a recursive expression for the joint steady state distribution of the successive sojourn times and the number of customers present in the system at each service completion of a tagged customer who has been fed back k times, $k = 0, 1, \ldots$. Using this result it is shown that the successive sojourn times have the same marginal distribution, which is negative exponential. We also derive some other sojourn time after k services and the correlation coefficient of the *i*-th and the *j*-th sojourn time of a

tagged customer, i < j. In particular, we prove the intuitively appealing properties that the latter quantity is positive and that it decreases if j-i grows. It is shown that for some interesting special choices of the feedback probabilities (e.g. Bernoulli feedback) the general expressions reduce to simple, explicit formulas. Chapter 2 concludes with the analysis of an M/G/1 queue with general feedback where the service time of a customer at each service depends on the number of times that he has been fed back. The results for this model are restricted to *mean* queue lengths and sojourn times. For the special case of deterministic feedback (i.e. each customer is fed back a fixed number of times) and all mean service times of a tagged customer are all equal.

In Chapter 3 the sojourn times in the M/G/1 PS queue are analyzed by taking appropriate limits in the M/M/1 queue with general feedback. We first formulate the limiting procedure and then show how this procedure can be applied to the sojourn time formulas for the M/M/1 feedback model obtained in Chapter 2. It appears that some well-known results for the M/G/1 PS queue (e.g. the mean conditional sojourn time) follow immediately from the product form properties of the joint queue length distribution in the feedback model. Next to the mean sojourn time we also derive the variance of the sojourn time and the (LST of the) sojourn time distribution. In particular, a new asymptotic result for the variance of the conditional sojourn time for customers with a very small service demand is obtained. Subsequently it is pointed out how the analysis of the M/G/1 PS queue can be extended to the analysis of the M/G/1 queue with generalized processor sharing by applying a similar limiting procedure to the M/M/1 queue with general feedback and state dependent service rates. Using known product form results for the latter model we present a new, simple, derivation of the queue length distribution and mean conditional sojourn time in the M/G/1 GPS queue. The last section of Chapter 3 is devoted to the analysis of sojourn times in the M/G/1 PS queue with Bernoulli feedback. From the results for the M/M/1 queue with general feedback obtained in Chapter 2 and application of the limiting procedure we derive new results for the correlation coefficients of the successive sojourn times of a tagged customer in the PS feedback model.

In Chapter 4 we develop some simple approximation formulas for the second moment of the conditional and unconditional sojourn time in the M/G/1 PS queue. The main reason for the development of these approximations is that the exact expressions can in general only be evaluated numerically and require perfect information about the service time distribution (which is almost never available in practical situations). The approximations depend on the service time distribution only through its first and second moment. They are mainly based on new asymptotic results (e.g. heavy traffic) and on simple exact expressions for some specific service time distributions. The many numerical examples show that these simple two-moment approximations are sufficiently accurate for many practical purposes. A refinement of the approximations is obtained by taking the third moment of the service time distribution into account.

As mentioned before, Chapter 5 is concerned with the study of some queueing models with additional permanent customers. First a detailed study of the basic model, the M/G/1 queue with permanent customers, is presented. The analysis is largely based on a decomposition of the queue length as a sum of independent random variables; the distributions of these random variables are obtained from known results for a related M/G/1 model with vacations. We derive queue length and sojourn time distributions for the Poisson customers and the permanent customers and we obtain simple explicit expressions for their first moments. Next, the M/M/1 queue with general feedback and additional permanent customers is studied. We obtain the rather remarkable result that for the case with $K \ge 1$ additional permanent customers the sojourn time distribution is the (K + 1)-fold convolution of the sojourn time distribution in the original system (i.e. without permanent customers). Application of the limiting procedure as described in Chapter 3 leads to a similar result for the M/G/1 PS queue with additional permanent customers.

Most results presented in this monograph are new except from that concerning the distribution of the sojourn time in the M/G/1 PS queue; actually the analysis of PS queues as presented in this study is new and our approach provides much more insight into the main sojourn time and queue length characteristics than previous methods.

The results of Chapter 2 are based on Van den Berg et al. [1989] and Van den Berg and Boxma [1989A]. The first three sections of Chapter 3 are mainly based on Van den Berg and Boxma [1989B]. Chapter 4 is based on Van den Berg [1989].

Chapter 2

THE M/M/1 QUEUE WITH GENERAL FEEDBACK

2.1 INTRODUCTION

In this chapter we consider an M/M/1 queue with a very general feedback mechanism. When a newly arriving customer, to be called a type-1 customer, has received his service, he departs from the system with probability 1-p(1) and is fed back to the end of the queue with probability p(1); in the latter case he becomes a type-2 customer. When he has received his *i*-th service, he leaves with probability 1-p(i) and he recycles with probability p(i), in the latter case becoming a type-(i + 1) customer. The service times of each customer at all visits are independent, identically, negative exponentially distributed random variables. The resulting queueing model has the property that the joint queue-length distribution of type-*i* customers, i = 1, 2, ..., is of product-form type. This property will be exploited to analyze the sojourn time process. In particular, we present a complete description of the joint distribution of the sojourn times of a customer on his successive cycles.

In the queueing literature, research on feedback queues has been mainly restricted to single server queues with Bernoulli feedback (see Takács [1963], Disney [1981], Disney et al. [1984] and Doshi and Kaufman [1988]). The Bernoulli feedback mechanism is a special case of the one in the present study: take $p(i) \equiv p$ in the general model. Lam and Shankar [1981] have studied a feedback model with basically the same feedback procedure as described above; they derive the total sojourn time distribution. This distribution comes out as a special case of our result for the joint distribution of a customer's successive sojourn times.

The organization of this chapter is as follows. In Section 2.2 the model is described in detail and some preliminary results are given. Section 2.3 contains our main result. We derive a formula for (the transform of) the joint distribution of the successive sojourn times of a tagged customer in the system and the numbers of customers of the various types present at his successive departure epochs. In Section 2.4 it is shown that the sojourn times in all individual cycles are identically, negative exponentially, distributed. Also, the correlation between the sojourn times of the *j*-th and *k*-th cycle of the tagged customer is calculated; furthermore, the distribution of the total sojourn time is derived. In Section 2.5 two special feedback mechanisms are studied:

Bernoulli feedback (Subsection 2.5.1) and deterministic feedback (Subsection 2.5.2). Finally, in Section 2.6 a similar model with *generally* distributed service times at each visit is considered. We derive a set of linear equations from which the mean sojourn time per visit can be calculated. Sections 2.2 through 2.5 are based on Van den Berg and Boxma [1989A]; Section 2.6 is based on Van den Berg et al. [1989].

2.2 MODEL DESCRIPTION

We consider a single server queueing system with infinite waiting room, see Fig. 2.1. Customers arrive at the system according to a Poisson process with intensity $\lambda > 0$. After having received a service, a customer may either leave the system or be fed back. When a customer has completed his *i*-th service, he departs from the system with probability 1-p(i) and is fed back with probability p(i). Fed back customers return instantaneously, joining the end of the queue. A customer who is visiting the queue for the *i*-th time will be called a type-*i* customer. To avoid the problems that occur in dealing with an infinite number of different customer types, it is assumed that after a certain number of services the feedback probabilities of a customer remain constant. Thus p(i) = p(N) := p, i = N, N + 1,... for some $N \ge 1$. The service discipline is first-come-first-served (FCFS).



Fig. 2.1 The M/M/1 queue with general feedback.

It is assumed that the successive service times of a customer are independent, negative exponentially distributed, random variables, with mean β . These service times are also independent of the service times of other customers. Introduce

$$q(0) := 1,$$

$$q(i) := \prod_{j=0}^{i-1} p(j), \quad i = 1, ..., N-1,$$

$$q(N) := \sum_{m=N}^{\infty} \prod_{j=0}^{m-1} p(j) = q(N-1)p(N-1)/(1-p),$$
(2.1)

with

$$p(0) := 1.$$

Note that $\lambda q(i)$ is the arrival rate of type-*i* customers, i = 1, ..., N. The total offered load to the queue per unit of time, denoted by ρ , is given by

$$\rho = \lambda \beta \sum_{i=1}^{N} q(i).$$
(2.2)

For stability it is required that $\rho < 1$.

We are interested in the following steady-state quantities:

- X_i : number of type-*i* customers in the system at an arbitrary epoch, i = 1,...,N;
- $\mathbf{X}_{i}^{(j)}$: number of type-*i* customers in the system at the *j*-th service completion of a customer, i = 1, ..., N, j = 1, 2, ...;
- $\mathbf{X}_{i}^{(0)}$: number of type-*i* customers in the system at the arrival of a new customer, i = 1, ..., N;
- S_j : time required for the *j*-th pass through the system (*j*-th sojourn time), j=1,2,...;

-
$$\mathbf{S}^{(k)}$$
: total sojourn time after k services: $\mathbf{S}^{(k)} = \sum_{j=1}^{k} \mathbf{S}_{j}, \ k = 1, 2...$

It is important to note that the system described above can be considered as a queueing network consisting of one queue with N types of customers. Type-*i* customers are fed back with probability p(i) after service, and then change into type-(i+1) customers, i=1,...,N-1. Type-N customers are fed back with probability p after service, and do not change their type. Because the service times are assumed to be independent exponentially distributed, the joint distribution of the number of type-*i* customers in the system at an arbitrary epoch, i=1,2,...,N, is of product-form type, see Baskett et al. [1975]. It is found that, for $x_1,...,x_N = 0,1,...,$

$$P(x_1,...,x_N) := Pr\{X_1 = x_1,...,X_N = x_N\} =$$
(2.3)

$$(1-\rho) \frac{(x_1 + \cdots + x_N)!}{x_1 ! \cdots x_N !} \prod_{i=1}^N (\lambda \beta q(i))^{x_i}.$$

It is convenient to have at our disposal the generating function of the joint queue length distribution. We have, for $|z_i| \leq 1, i = 1, ..., N$,

$$E\{z_{1}^{\mathbf{X}_{1}}\cdots z_{N}^{\mathbf{X}_{N}}\} = \sum_{x_{1}=0}^{\infty}\cdots\sum_{x_{N}=0}^{\infty} z_{1}^{\mathbf{X}_{1}}\cdots z_{N}^{\mathbf{X}_{N}} P(x_{1},\cdots,x_{N}) = (2.4)$$

$$(1-\rho)\sum_{m=0}^{\infty}\sum_{x_{1}}\cdots\sum_{x_{N}}\frac{m!}{x_{1}!\cdots x_{N}!}\prod_{i=1}^{N}(\lambda\beta q(i)z_{i})^{x_{i}} = (1-\rho)\sum_{m=0}^{\infty}\left[\sum_{i=1}^{N}\lambda\beta q(i)z_{i}\right]^{m} = \frac{1-\rho}{1-\sum_{i=1}^{N}\lambda\beta q(i)z_{i}}.$$

The distribution of the total number of customers in the system coincides with the queue length distribution in an ordinary M/M/1 model:

$$E\{z^{\mathbf{X}_{1}+...+\mathbf{X}_{N}}\} = \frac{1-\rho}{1-\rho z}, \quad |z| \leq 1,$$

$$Pr\{\mathbf{X}_{1}+...+\mathbf{X}_{N}=j\} = (1-\rho)\rho^{j}, \quad j=0,1,....$$
(2.5)

We shall use these results in the next section.

2.3 MAIN RESULTS

In this section we present, in the form of Laplace-Stieltjes transforms and generating functions, an expression for the joint steady-state distribution of the successive sojourn times \mathbf{S}_{j} , j = 1,...,k, and the number of type-*i* customers, $\mathbf{X}_{i}^{(j)}$, i = 1,...,N, present at the *j*-th service completion of a customer who is fed back at least k - 1 times, k = 1,2,....

Let us follow a tagged customer from the moment he arrives as a type-1 customer until he completes his k-th service. Obviously, the k successive sojourn times of the tagged customer depend on the number of customers of each type present in the system upon his arrival, the behaviour of these customers and the behaviour of subsequent arrivals. The PASTA property (Wolff [1982]) implies the equality of the joint queue length distribution at the epoch of a new arrival and at an arbitrary epoch:

$$Pr\{\mathbf{X}_{1}^{(0)}=x_{1},\ldots,\mathbf{X}_{N}^{(0)}=x_{N}\}=P(x_{1},\ldots,x_{N}), \quad x_{1},\ldots,x_{N}=0,1,\ldots.$$
(2.6)

Hence, for Re $\omega_i \ge 0$, $|z_{i,j}| \le 1$, i = 1, ..., N, j = 0, ..., k,

$$E\{e^{-(\omega_{1}\mathbf{S}_{1}+...+\omega_{k}\mathbf{S}_{k})}(z_{1,0}^{\mathbf{X}_{1}^{(0)}}\cdots z_{N,0}^{\mathbf{X}_{N}^{(0)}})\cdots (z_{1,k}^{\mathbf{X}_{1}^{(k)}}\cdots z_{N,k}^{\mathbf{X}_{N}^{(k)}})\} =$$
(2.7)

20

i.e.

$$\sum_{x_1=0}^{\infty} \cdots \sum_{x_N=0}^{\infty} P(x_1, \ldots, x_N) \times E\{e^{-(\omega_1 \mathbf{S}_1 + \ldots + \omega_k \mathbf{S}_k)}(z_{1,0}^{\mathbf{X}_1^{(0)}} \cdots z_{N,0}^{\mathbf{X}_N^{(0)}}) \cdots (z_{1,k}^{\mathbf{X}_1^{(k)}} \cdots z_{N,k}^{\mathbf{X}_N^{(k)}}) | \mathbf{X}_1^{(0)} = x_1, \ldots, \mathbf{X}_N^{(0)} = x_N\}.$$

The conditional expectation in the RHS of (2.7) can be evaluated by using the following property, which is easily seen to hold: $(\mathbf{X}_{1}^{(i+1)}, \dots, \mathbf{X}_{N}^{(i+1)})$, which determines the distribution of \mathbf{S}_{i+2} , is conditionally independent of $\{(\mathbf{X}_{1}^{(i)}, \dots, \mathbf{X}_{N}^{(i)}), j = 0, ..., i-1; \mathbf{S}_{1}, \dots, \mathbf{S}_{i}\}$ given $\{(\mathbf{X}_{1}^{(i)}, \dots, \mathbf{X}_{N}^{(i)}); \mathbf{S}_{i+1}\}$, i = 1, ..., k - 1, i.e. the joint process of successive service completion epochs and queue length vector at these service completion epochs is a *Markov renewal* process (cf. Çinlar [1975, Ch. 10]). The calculations, which are very lengthy, are omitted here; they can be found in Appendix 2.1 at the end of this chapter. There it is shown that

$$E\{e^{-(\omega_{1}\mathbf{S}_{1}+...+\omega_{k}\mathbf{S}_{k})}(z_{1,0}^{\mathbf{X}_{1}^{(0)}}\cdots z_{N,0}^{\mathbf{X}_{N}^{(0)}})\cdots (z_{1,k}^{\mathbf{X}_{1}^{(k)}}\cdots z_{N,k}^{\mathbf{X}_{N}^{(k)}}) | \mathbf{X}_{1}^{(0)} = x_{1}, \ldots, \mathbf{X}_{N}^{(0)} = x_{N}\}$$

$$=\prod_{k}^{k} A_{k}^{N}(j,\omega,z) \prod_{i=1}^{N} (z_{i,0}f_{k}^{N}(i,\omega,z))^{x_{i}}, \qquad (2.8)$$

$$j=1$$
 $i=1$
with $\omega:=(\omega_1,\ldots,\omega_k), \ z:=((z_{1,0},\ldots,z_{N,0}),\ldots,(z_{1,k},\ldots,z_{N,k})),$ and

$$A_k^N(1,\omega,z) := [1 + \beta \{\omega_k + \lambda(1 - z_{1,k})\}]^{-1},$$
(2.9)

$$A_{k}^{N}(2,\omega,z) := [1 + \beta \{\omega_{k-1} + \lambda - \lambda z_{1,k-1}A_{k}^{N}(1,\omega,z)[p(1)z_{2,k} + 1 - p(1)]\}]^{-1},$$

$$A_{k}^{N}(i,\omega,z) := [1+\beta\{\omega_{k-i+1}+\lambda-\lambda z_{1,k-i+1}A_{k}^{N}(i-1,\omega,z)[A_{k}^{N}(i-2,\omega,z)[\cdots [A_{k}^{N}(2,\omega,z)[A_{k}^{N}(1,\omega,z)[p(i-1)z_{i,k}+1-p(i-1)]]$$

$$p(i-2)z_{i-1,k-1}+1-p(i-2)]p(i-3)z_{i-2,k-2}+1-p(i-3)]\cdots]$$

$$p(1)z_{2,k-i+2}+1-p(1)]\}]^{-1}, \quad i=3,...,k,$$

$$f_{k}^{N}(i,\omega,z) := A_{k}^{N}(k,\omega,z)[A_{k}^{N}(k-1,\omega,z)[\cdots [A_{k}^{N}(2,\omega,z)]A_{k}^{N}(1,\omega,z) \qquad (2.10)$$

$$[p(k+i-1)z_{k+i,k}+1-p(k+i-1)]p(k+i-2)z_{k+i-1,k-1}+$$

$$1-p(k+i-2)]p(k+i-3)z_{k+i-2,k-2}+1-p(k+i-3)]\cdots]$$

$$p(i)z_{i+1,1}+1-p(i)], \quad i=1,...,N.$$

Here we have defined

•

Ċ,

$$z_{i,j}:=z_{N,j}, \quad i=N+1,...,N+k$$

Remark 2.1

From the calculations in Appendix 2.1 it is seen that the factor $(z_{i,0}f_k^N(i,\omega,z))^{x_i}$ in the RHS of (2.8) is due to the contribution to $\{(\mathbf{X}_1^{(j)},\ldots,\mathbf{X}_k^{(j)}), j=0,...,k; \mathbf{S}_1,\ldots,\mathbf{S}_k\}$ induced by the x_i type-*i* customers present in the system upon the first arrival of the tagged customer, i=1,...,N; the factor $\prod_{j=1}^k A_k^N(j,\omega,z)$ is due to the contribution induced by the tagged customer himself. These contributions are all independent, cf. (2.8).

Substituting (2.8) and (2.3) into (2.7) and evaluating the summations (use (2.4)) we obtain our main result:

THEOREM 2.1

$$E\{e^{-(\omega_{1}\mathbf{S}_{1}+...+\omega_{k}\mathbf{S}_{k})}(z_{1,0}^{\mathbf{X}_{1}^{(0)}}\cdots z_{N,0}^{\mathbf{X}_{N}^{(0)}})\cdots (z_{1,k}^{\mathbf{X}_{k}^{(i)}}\cdots z_{N,k}^{\mathbf{X}_{N}^{(i)}})\} =$$
(2.11)
$$\frac{(1-\rho)\prod_{j=1}^{k}A_{k}^{N}(j,\omega,z)}{1-\lambda\beta\sum_{i=1}^{N}q(i)z_{i,0}f_{k}^{N}(i,\omega,z)}, \quad \text{Re } \omega_{j} \ge 0, \ |z_{i,j}| \le 1, \ i=1,...,N, \ j=0,...,k.$$

COROLLARY 2.1

The Laplace-Stieltjes transform of the joint distribution of the first k successive sojourn times of a customer, who is fed back at least k-1 times, is given by

$$E\{e^{-(\omega_{1}\mathbf{S}_{1}+...+\omega_{k}\mathbf{S}_{k})}\} = \frac{(1-\rho)\prod_{j=1}^{k}A_{k}^{N}(j,\omega)}{1-\lambda\beta\sum_{i=1}^{N}q(i)f_{k}^{N}(i,\omega)},$$
(2.12)

with,

$$A_k^N(1,\omega) := [1 + \beta \omega_k]^{-1}, \tag{2.13}$$

$$A_k^N(2,\omega) := [1 + \beta \{\omega_{k-1} + \lambda - \lambda A_k^N(1,\omega)\}]^{-1},$$

$$A_{k}^{N}(i,\omega) := [1+\beta\{\omega_{k-i+1}+\lambda-\lambda A_{k}^{N}(i-1,\omega)[A_{k}^{N}(i-2,\omega)[\cdots [A_{k}^{N}(2,\omega)]A_{k}^{N}(1,\omega) \\ p(i-2)+1-p(i-2)]p(i-3)+1-p(i-3)]\cdots]p(1)+1-p(1)]\}]^{-1},$$

$$i=3,...,k,$$

$$f_{k}^{N}(i,\omega) := A_{k}^{N}(k,\omega)[A_{k}^{N}(k-1,\omega)[\cdots]$$

$$[A_{k}^{N}(2,\omega)[A_{k}^{N}(1,\omega)p(k+i-2)+1-p(k+i-2)]$$

$$p(k+i-3)+1-p(k+i-3)[\cdots]p(i)+1-p(i)], \quad i=1,...,N.$$
(2.14)

Proof

Substitute $z_{i,j} = 1$ into (2.9)-(2.11), i = 1, ..., N, j = 0, ..., k.

COROLLARY 2.2

.

The joint distribution of the number of type-i customers, i = 1,...,N, present in the queue at the end of the j-th service of a tagged customer is independent of j and given by

$$E\{z_{1}^{\mathbf{X}_{1}^{(j)}}\cdots z_{N}^{\mathbf{X}_{N}^{(j)}}\} = E\{z_{1}^{\mathbf{X}_{1}}\cdots z_{N}^{\mathbf{X}_{N}}\} = \frac{1-\rho}{1-\lambda\beta\sum_{i=1}^{N}q(i)z_{i}}, \quad (2.15)$$
$$|z_{i}| \leq 1, \ i=1,...,N, \ j=0,1,...,k.$$

Proof

It follows from (2.3) and (2.6) that (2.15) holds for j=0. If (2.15) also holds for j=1, then it clearly holds for all j=0,1,.... The validity of (2.15) for j=1 follows by a simple calculation.

Remark 2.2

Corollary 2.2 is, in a more general context, known as the 'arrival theorem' for product form networks, see e.g. Walrand [1988, Section 4.4]. This theorem implies that an arriving type-*i* customer (who has just completed his (i - 1)-th service) 'sees' the system as at an arbitrary epoch.

The Laplace-Stieltjes transform of the joint sojourn time distribution ((2.12)-(2.14)) can be presented in a form which is more suitable for obtaining sojourn time moments. For this purpose we first rewrite (2.13) and (2.14):

$$A_{k}^{N}(i,\omega) := [1 + \beta \omega_{k-i+1} + \lambda \beta \{1 - \tilde{q}(i-1) \prod_{j=1}^{i-1} A_{k}^{N}(j,\omega) - (2.16)$$

$$\sum_{l=2}^{i-1} \tilde{q}(i-l)(1 - p(i-l)) \prod_{j=l}^{i-1} A_{k}^{N}(j,\omega) \}]^{-1}, \quad i = 1, ..., k,$$

$$f_{k}^{N}(i,\omega) := \frac{1}{\tilde{q}(i)} [\tilde{q}(k+i-1) \prod_{j=1}^{k} A_{k}^{N}(j,\omega) +$$
(2.17)

$$\sum_{l=2}^{k} \tilde{q}(k+i-l)(1-p(k+i-l))\prod_{j=l}^{k} A_{k}^{N}(j,\omega)], \quad i=1,...,N,$$

with

$$\tilde{q}(0) := 1,$$

 $\tilde{q}(i) := \prod_{j=0}^{i-1} p(j), \quad i = 1, 2, ...,$

and an empty product being one by definition. From (2.17) it is found that the summation in the denominator of (2.12) can be written as

$$\lambda \beta \sum_{i=1}^{N} q(i) f_k^N(i,\omega) = (\rho - \lambda \beta \sum_{i=1}^{k-1} \tilde{q}(i)) \prod_{j=1}^{k} A_k^N(j,\omega) + \lambda \beta \sum_{l=2}^{k} \tilde{q}(k-l+1) \prod_{j=l}^{k} A_k^N(j,\omega) .$$

Substituting this into (2.12) and introducing

$$\begin{split} M_k(i,\omega) &:= \prod_{j=1}^i \frac{1}{A_k^N(j,\omega)}, \quad i=1,...,k, \\ M_k(0,\omega) &:= 1, \end{split}$$

we obtain

$$E\{e^{-(\omega_1\mathbf{S}_1+\ldots+\omega_k\mathbf{S}_k)}\} =$$
(2.18)

$$\frac{1-\rho}{M_k(k,\omega) - \lambda\beta \sum_{l=2}^k \tilde{q}(k-l+1)M_k(l-1,\omega) - (\rho-\lambda\beta \sum_{i=1}^{k-1} \tilde{q}(i))},$$

with, from (2.16),

$$M_{k}(i,\omega) = (1+\beta\omega_{k-i+1})M_{k}(i-1,\omega) + \lambda\beta \left[M_{k}(i-1,\omega) - \tilde{q}(i-1) - (2.19)\right]$$
$$\sum_{j=2}^{i-1} \tilde{q}(i-j)(1-p(i-j))M_{k}(j-1,\omega), \quad i = 1,...,k.$$

Note that (2.18) and (2.19) are *independent* of N - the number of different customer types in the system. Hence, this result for the joint sojourn time distribution is also valid without the assumption made in Section 2.2 that the feedback probabilities remain constant after a finite number N of services. Moreover, it follows from (2.18) and (2.19) that the joint distribution of the first k successive sojourn times of a particular customer depends on the feedback

probabilities only through $p(1), \ldots, p(k-3)$ and ρ (which reflects the influence of p(k-2), p(k-1), ..., cf. (2.1), (2.2)). In the next section we shall use (2.18) and (2.19) to derive some important sojourn time characteristics.

2.4 SOJOURN TIME CHARACTERISTICS

In this section we derive expressions for some important sojourn time characteristics such as the marginal distribution of the successive sojourn times, the correlation coefficient of the *i*-th and the *j*-th sojourn time of a particular customer, and the mean and variance of the total sojourn time after k services. As an example we study the case that a customer receives exactly 2 services; for this case simple explicit results for the above mentioned sojourn time characteristics are obtained.

The fact that the joint queue length distribution at the arrival of a customer and after each of his passes is the same (cf. Corollary 2.2), implies that the sojourn times S_j , j=1,...,k have the same marginal distribution. S_1 can easily be obtained from (2.18) and (2.19) by taking k=1. It is found that the sojourn times are negative exponentially distributed with mean $\beta/(1-\rho)$:

$$E\{e^{-\omega_{j}\mathbf{S}_{j}}\} = \frac{1-\rho}{1-\rho+\beta\omega_{j}}, \quad j=1,...,k.$$
(2.20)

Note that this coincides with the sojourn time transform in an ordinary M/M/1 queue with mean service time β and arrival rate $\lambda \sum_{i=1}^{N} q(i)$, cf. (2.5).

In order to investigate the dependence between the *i*-th and *j*-th sojourn times we have computed the Laplace-Stieltjes transform of the joint distribution of S_i and S_j , $1 \le i < j \le k$. It is found from (2.18) and (2.19) that

$$E\{e^{-(\omega_i \mathbf{S}_i + \omega_j \mathbf{S}_j)}\} = \frac{1 - \rho}{1 - \rho + \beta \omega_i + \beta \omega_j + \beta^2 \omega_i \omega_j C_{j-i}}, \quad 1 \le i < j \le k, \quad (2.21)$$

where C_{i-i} is determined by

$$C_{1} = 1,$$

$$C_{n} = (1+\lambda\beta)C_{n-1} - \lambda\beta \sum_{l=2}^{n-1} \tilde{q}(n-l)(1-p(n-l))C_{l-1}, \quad n = 2,...,k-1.$$
(2.22)

Note that $E\{e^{-(\omega_i \mathbf{S}_i + \omega_j \mathbf{S}_j)}\}$ only depends on *i* and *j* through the difference j - i. This property might also have been derived from Corollary 2.2.

Remark 2.3

It was pointed out by Prof. J.W. Cohen that the two-dimensional Laplace-Stieltjes transform given by (2.21) is of a type for which the corresponding joint probability density function, $f_{i,j}(\cdot, \cdot)$, is known. From the formula given in entry 8 of Table B in Voelker and Doetsch [1950, p. 208] it is found that, for $1 \le i < j \le k$,

$$f_{i,j}(x,y) =$$

$$\frac{1-\rho}{\beta^2 C_{j-i}} e^{-(x+y)/(\beta C_{j-i})} \sum_{m=0}^{\infty} \left[\frac{-xy}{\beta^2 C_{j-i}} \right]^m (1-\rho+1/C_{j-i})^m (1/m!)^2, \quad x,y \ge 0.$$
(2.23)

From (2.21) the correlation coefficient, $corr(\mathbf{S}_i, \mathbf{S}_i)$, can easily be obtained:

$$corr(\mathbf{S}_i, \mathbf{S}_j) = 1 - C_{j-i}(1-\rho), \quad 1 \le i < j \le k.$$
 (2.24)

It follows from (2.22) and (2.24) that $corr(\mathbf{S}_i, \mathbf{S}_j)$ as a function of i and j only depends on j-i. Noting that in (2.22) $\sum_{l=2}^{n-1} \tilde{q}(n-l)(1-p(n-l)) \leq 1$ (remember that $\tilde{q}(n-l)(1-p(n-l))$ is the probability that a customer receives exactly n-l services) it follows by induction that the row $\{C_n, n=1,2,...\}$ is monotonically increasing. Hence, from (2.24), $corr(\mathbf{S}_i, \mathbf{S}_j)$ decreases if j-i grows. In particular it can be proven that $\lim_{n\to\infty} C_n = 1/(1-\rho)$, see Chapter 3, yielding $\lim_{j=i\to\infty} corr(\mathbf{S}_i, \mathbf{S}_j) = 0$. For j-i=1, $corr(\mathbf{S}_i, \mathbf{S}_j) = \rho$. So, the successive sojourn times of a tagged customer are always correlated positively.

The Laplace-Stieltjes transform of the distribution of a customer's total time spent in the system until the end of his k-th pass $\mathbf{S}^{(k)} := \mathbf{S}_1 + ... + \mathbf{S}_k$, can be obtained from (2.18) by substituting $\omega_j = \omega_0$, j = 1, ..., k. From (2.20) it follows immediately that $E\{\mathbf{S}^{(k)}\}$ is linear in k:

$$E\{\mathbf{S}^{(k)}\} = \sum_{i=1}^{k} E\{\mathbf{S}_i\} = k \frac{\beta}{1-\rho} .$$
(2.25)

To derive an expression for the variance of this sojourn time, $var(S^{(k)})$, it is convenient to use the formula

$$var(\mathbf{S}^{(k)}) = \sum_{i=1}^{k} var(\mathbf{S}_i) + 2\sum_{i=1}^{k} \sum_{j=i+1}^{k} cov(\mathbf{S}_i, \mathbf{S}_j).$$

Hence, from (2.20),

$$var(\mathbf{S}^{(k)}) = k \ var(\mathbf{S}_1) + 2\sum_{i=1}^k \sum_{j=i+1}^k cov(\mathbf{S}_i, \mathbf{S}_j).$$

The covariance of S_i and S_j , $cov(S_i, S_j)$, and $var(S_1)$ can easily be obtained from the results (2.20) and (2.24). It is found that

$$var(\mathbf{S}^{(k)}) = \left[\frac{\beta}{1-\rho}\right]^2 \left[k^2 - 2(1-\rho)\sum_{j=1}^{k-1} jC_{k-j}\right],$$
(2.26)

with $C_1, ..., C_{k-1}$ given by (2.22).

.

The Laplace-Stieltjes transform of the distribution of the total sojourn time S of an arbitrary customer is now given by

$$E\{e^{-\omega_0 \mathbf{S}}\} = \sum_{k=1}^{\infty} \tilde{q}(k)(1-p(k))E\{e^{-\omega_0 \mathbf{S}^{(k)}}\}.$$
(2.27)

In an example, we shall examine the case k=2 for which explicit closed form results can easily be obtained.

EXAMPLE 2.1 (*The case* k = 2) From (2.21) and (2.22) it follows that for the case k = 2,

$$E\{e^{-(\omega_{1}\mathbf{S}_{1}+\omega_{2}\mathbf{S}_{2})}\} = \frac{1-\rho}{1-\rho+\beta\omega_{1}+\beta\omega_{2}+\beta^{2}\omega_{1}\omega_{2}}.$$
(2.28)

Note that the feedback probabilities p(i), i = 1,...,N, enter into the joint distribution of S_1 and S_2 only via the offered load ρ . Thus, as long as ρ remains constant, the joint distribution of S_1 and S_2 is independent of the individual values of p(i), i = 1,...,N. (Consequently, this also holds for S_i and S_{i+1} , i = 1,2,..., cf. (2.21)). Doshi and Kaufman [1988] derived (2.28) for the (special) case of Bernoulli feedback ($p(i) \equiv p$).

From (2.24) it follows that

$$corr(\mathbf{S}_1, \mathbf{S}_2) = \rho. \tag{2.29}$$

Let $F_2(t)$ denote the distribution function of the sojourn time until the end of the second pass:

$$F_2(t) := Pr\{\mathbf{S}_1 + \mathbf{S}_2 < t\}, \quad t \ge 0.$$

From (2.28) we find

$$E\{e^{-\omega_0(\mathbf{S}_1+\mathbf{S}_2)}\}=\frac{1+\sqrt{\rho}}{2\sqrt{\rho}}\frac{1-\sqrt{\rho}}{1-\sqrt{\rho}+\beta\omega_0}-\frac{1-\sqrt{\rho}}{2\sqrt{\rho}}\frac{1+\sqrt{\rho}}{1+\sqrt{\rho}+\beta\omega_0}.$$

Hence

$$F_2(t) = \frac{1 + \sqrt{\rho}}{2\sqrt{\rho}} (1 - e^{-t(1 - \sqrt{\rho})/\beta}) - \frac{1 - \sqrt{\rho}}{2\sqrt{\rho}} (1 - e^{-t(1 + \sqrt{\rho})/\beta}), \quad t \ge 0.$$
(2.30)

In Doshi and Kaufman [1988], $F_2(\cdot)$ is compared with the distribution of S_1+S_2 that results when one assumes that S_1 and S_2 are independent. Due to the positive correlation between S_1 and S_2 (cf. (2.29)), it is found that $F_2(\cdot)$ has a longer "tail" than this approximate distribution.

Finally, the variance of $S_1 + S_2$ is obtained from (2.26):

$$var(\mathbf{S}_1 + \mathbf{S}_2) = 2(1+\rho) \left[\frac{\beta}{1-\rho}\right]^2.$$
(2.31)

2.5 Special cases: Bernoulli feedback and deterministic feedback

In this section we study two feedback systems which are special cases of the general model described in Section 2.2, viz., Bernoulli feedback (Subsection 2.5.1) and deterministic feedback (Subsection 2.5.2). For these models we obtain simple, explicit expressions for most of the quantities analyzed in Section 2.3. The results for the deterministic feedback model have been published before in Van den Berg et al. [1989]. The Laplace-Stieltjes transform of the joint sojourn time distribution in the Bernoulli feedback model has also been derived by Doshi and Kaufman [1988].

2.5.1 Bernoulli feedback

The Bernoulli feedback model is obtained from the general model by taking $p(i) \equiv p$: when a customer completes his service he departs from the system with probability 1-p and is fed back with probability p. Obviously

$$\rho = \frac{\lambda\beta}{1-p}.$$

The Laplace-Stieltjes transform of the joint distribution of the successive sojourn times S_1, \ldots, S_k can be obtained from (2.12)-(2.14) (or from (2.18) and (2.19)) by substituting $p(i) \equiv p$. The expression that results from (2.12)-
(2.14) has also been derived by Doshi and Kaufman [1988].

To obtain explicit expressions for $E\{e^{-(\omega_i \mathbf{S}_i + \omega_j \mathbf{S}_j)}\}$, $corr(\mathbf{S}_i, \mathbf{S}_j)$, and $var(\mathbf{S}^{(k)})$, (see (2.21), (2.24) and (2.26)) we have to derive C_n , n = 1, ..., k - 1, from the set of difference equations (2.22). After the substitution $\tilde{q}(j) = p^{j-1}$, j = 1, 2, ..., (2.22) reduces to

$$C_{1} = 1,$$

$$C_{n} = (1+\lambda\beta)C_{n-1} - \lambda\beta \sum_{l=2}^{n-1} (p^{n-l-1} - p^{n-l})C_{l-1}, \quad n = 2, ..., k-1.$$
(2.32)

From (2.32) it follows that

. •

$$C_{1} = 1,$$

$$C_{2} = 1 + \lambda \beta,$$

$$C_{n} - pC_{n-1} = (1 + \lambda \beta)C_{n-1} - p(1 + \lambda \beta)C_{n-2} - \lambda \beta(1-p)C_{n-2},$$

$$n = 3, ..., k - 1.$$

Hence

$$C_{1} = 1,$$

$$C_{2} = 1 + \lambda \beta,$$

$$C_{n} = (1 + \lambda \beta + p)C_{n-1} - (\lambda \beta + p)C_{n-2}, \quad n = 3, ..., k - 1.$$
(2.33)

The general solution of (2.33) is given by

$$C_n = U_1 y_1^n + U_2 y_2^n,$$

where $y_1 = 1$ and $y_2 = \lambda \beta + p$ are the roots of

$$y^2 - (1 + \lambda\beta + p)y + (\lambda\beta + p) = 0,$$

and U_1 and U_2 are determined by

$$U_1y_1 + U_2y_2 = 1,$$

$$U_1y_1^2 + U_2y_2^2 = 1 + \lambda\beta.$$

After some calculations it is found that

$$C_n = \frac{1 - \rho (\lambda \beta + p)^{n-1}}{1 - \rho}, \quad n = 1, ..., k - 1.$$
(2.34)

Substitution of (2.34) in (2.21), (2.24) and (2.26) yields, respectively

$$E\{e^{-(\omega_i \mathbf{S}_i + \omega_j \mathbf{S}_j)}\} = \frac{1-\rho}{1-\rho+\beta\omega_i+\beta\omega_j+\beta^2\omega_i\omega_j\frac{1-\rho(\lambda\beta+p)^{j-i-1}}{1-\rho}},$$
(2.35)

 $1 \leq i < j \leq k$, Re $\omega_i, \omega_j \geq 0$,

$$corr(\mathbf{S}_i, \mathbf{S}_j) = \rho(\lambda \beta + p)^{j-i-1}, \quad 1 \le i < j \le k,$$
(2.36)

$$var(\mathbf{S}^{(k)}) = \tag{2.37}$$

$$\left[\frac{\beta}{1-\rho}\right]^{2} \left[k+2\frac{\rho}{1-p}\left\{\frac{k}{1-\rho}-\frac{1}{1-p}\frac{1}{(1-\rho)^{2}}(1-(p+\lambda\beta)^{k})\right\}\right],\ k=1,2,\dots.$$

It follows from (2.36) that $\lim_{j \to i\uparrow\infty} corr(\mathbf{S}_i, \mathbf{S}_j) = 0$ (cf. Section 2.3). It is also seen that $corr(\mathbf{S}_i, \mathbf{S}_j)$ is an increasing function of $\lambda\beta$ for fixed *i* and *j*. These intuitively appealing properties are illustrated in Fig. 2.2.

The Laplace-Stieltjes transform of the distribution of $S^{(k)}$ can be obtained from (2.18) and (2.19) by substituting $\omega_j = \omega_{0, j} = 1, ..., k$. The resulting set of difference equations (2.19) can be solved in the same way as (2.33). After extensive but straightforward calculations it is found that

$$E\{e^{-\omega_0 \mathbf{S}^{(k)}}\} = \frac{1 - p - \lambda \beta}{Q_1(1 - x_2)x_1^k + Q_2(1 - x_1)x_2^k}, \quad \text{Re } \omega_0 \ge 0, \quad k = 1, 2, \dots, \quad (2.38)$$

where,

$$\begin{aligned} x_1 &= \frac{1 + \beta \omega_0 + \lambda \beta + p + \sqrt{(1 + \beta \omega_0 + \lambda \beta + p)^2 - 4(p + p\beta \omega_0 + \lambda \beta)}}{2}, \\ x_2 &= \frac{1 + \beta \omega_0 + \lambda \beta + p - \sqrt{(1 + \beta \omega_0 + \lambda \beta + p)^2 - 4(p + p\beta \omega_0 + \lambda \beta)}}{2}, \\ Q_1 &= \frac{x_2 - (1 + \beta \omega_0)}{x_2 - x_1}, \end{aligned}$$

$$Q_2 = \frac{x_1 - (1 + \beta \omega_0)}{x_1 - x_2}.$$



Fig. 2.2 corr $(\mathbf{S}_i, \mathbf{S}_j)$ as a function of offered load $\rho = \frac{\lambda \beta}{1-p}$, with p = 0.5.

2.5.2 Deterministic feedback

Taking p(i)=1, i=1,...,N-1, p(N)=p=0, we obtain the deterministic feedback model in which each customer is fed back exactly N-1 times and leaves the system after the N-th service. Obviously

oviously

$$\rho = N\lambda\beta.$$

Noting that

$$q(j) = \tilde{q}(j) = 1, \quad 0 \leq j \leq N,$$
$$= 0, \quad j > N,$$

it is easily seen from (2.18) and (2.19) that

$$E\{e^{-(\omega_1\mathbf{S}_1+\ldots+\omega_k\mathbf{S}_k)}\} = \frac{1-\rho}{M_k(k,\omega) - \lambda\beta \sum_{i=0}^{k-1} M_k(i,\omega) - (N-k)\lambda\beta}, \quad (2.39)$$

with

32

$$M_k(0,\omega) = 1,$$

$$M_k(i,\omega) = (1+\lambda\beta+\beta\omega_{k-i+1})M_k(i-1,\omega)-\lambda\beta, \quad i=1,...,k, \quad k \le N.$$
(2.40)

At the end of this subsection we shall use (2.39) and (2.40) to obtain an explicit expression for the Laplace-Stieltjes transform of the total sojourn time distribution.

As in Subsection 2.5.1 we solve the set of difference equations (2.22) to obtain explicit expressions for $E\{e^{-(\omega_i \mathbf{S}_i + \omega_j \mathbf{S}_j)}\}$, $corr(\mathbf{S}_i, \mathbf{S}_j)$ and $var(\mathbf{S}^{(k)})$ from the general formulas (2.21), (2.24) and (2.26). Substituting in (2.22) $p(i) = \tilde{q}(i) = 1, i = 1, ..., N - 1$, we get

$$C_1 = 1,$$

 $C_n = (1 + \lambda \beta) C_{n-1}, \quad n = 2, ..., N - 1.$

Hence

.

$$C_n = (1 + \lambda \beta)^{n-1}, \quad n = 1, ..., N-1.$$

Now it follows from (2.21), (2.24) and (2.26) that

$$E\{e^{-(\omega_i \mathbf{S}_i + \omega_j \mathbf{S}_j)}\} = \frac{1 - \rho}{1 - \rho + \beta \omega_i + \beta \omega_j + \beta^2 \omega_i \omega_j (1 + \lambda \beta)^{j - i - 1}},$$
(2.41)

 $1 \le i < j \le N$,

$$corr(\mathbf{S}_i, \mathbf{S}_j) = 1 - (1 - \rho)(1 + \lambda \beta)^{j - i - 1}, \quad 1 \le i < j \le N,$$
(2.42)

$$var(\mathbf{S}^{(k)}) = \left[\frac{\beta}{1-\rho}\right]^2 \left[k^2 - 2(1-\rho)\frac{(1+\lambda\beta)^k - k\lambda\beta - 1}{(\lambda\beta)^2}\right], \quad k \le N.$$
(2.43)

The Laplace-Stieltjes transform of the distribution of the total sojourn time after k services is obtained from (2.39) and (2.40) by substituting $\omega_j = \omega_0, j = 1, ..., k$. The resulting set of relations yields

$$M_{k}(i,\omega) = (1+\lambda\beta+\beta\omega_{0})^{i} - \lambda\beta \sum_{j=0}^{i-1} (1+\lambda\beta+\beta\omega_{0})^{j} =$$

$$\frac{\omega_{0}}{\lambda+\omega_{0}} (1+\lambda\beta+\beta\omega_{0})^{i} + \frac{\lambda}{\lambda+\omega_{0}}, \quad i=0,...,k, \quad k \leq N.$$
(2.44)

Using (2.44) it follows from (2.39) that

$$E\{e^{-\omega_0 S^{(k)}}\} =$$
 (2.45)

$$\frac{(1-\rho)(\lambda+\omega_0)^2}{\omega_0^2(1+\lambda\beta+\beta\omega_0)^k+\lambda(\lambda+\omega_0)(1-\rho-(N-k)\beta\omega_0)+\lambda\omega_0}, \quad \text{Re } \omega_0 \ge 0, \ k \le N.$$

Remark 2.4

The simple explicit formula (2.45) for the LST of the total sojourn time distribution has been obtained under the assumption that $k \leq N$. Unfortunately, this result can not be extended to the sojourn time of a (special) customer who is fed back k > N times. The problem is that, for k > N, the set of difference equations (2.19) for the $M_k(i, \omega)$'s can not be explicitly solved (cf. (2.44) for the case $k \leq N$). The same holds for the solution of the C_n 's from (2.22) for the analysis of the sojourn time variance.

2.6 Further extensions

In this section we consider the feedback model described in Section 2.2 with the following extension: the successive service times of a customer are generally distributed and may depend on the number of times he has already been fed back. We also omit the assumption made in Section 2.2 that the feedback probabilities remain constant after a finite number (N) of services (cf. the discussion below (2.19)). For this extended model the joint stationary distribution of the number of type-*i* customers in the system is no longer of product form type. In fact no results concerning the distribution of the queue length are available. Consequently, it can not be expected that we are able to obtain sojourn time distributions. Therefore, in this section, we restrict ourself to the derivation of mean queue lengths and mean sojourn times. First, as in Simon [1984] (cf. Subsection 1.5.1), we derive a set of linear equations from which the mean sojourn time per visit can be calculated. Next, we show that for the special case of deterministic feedback with all mean service times equal (but not necessarily negative exponentially distributed), this set of linear equations can be solved explicitly. It appears that from the second visit on, all mean sojourn times are equal. Finally, explicit results are obtained for the case of Bernoulli feedback.

2.6.1 Derivation of a set of linear equations

We consider the case that the service time distribution of a customer who has been fed back i-1 times is given by $B_i(\cdot)$, with mean β_i and second moment $\beta_i^{(2)}$, i=1,2,... The definitions of type-*i* customers and their characteristic quantities, as given in Section 2.2, are extended in an obvious way. Denote by $\rho_i := \lambda q(i)\beta_i$ the offered traffic due to type-*i* customers. Obviously the stability condition for this system is that $\rho = \sum_{i=1}^{\infty} \rho_i < 1$. We start by obtaining a relation for ES_1 . Note that a newly arriving customer is a Poisson arrival and hence PASTA (see Wolff [1982]) applies. Consider the mean amount of work that has to be handled before this newly arriving customer (in the following: the tagged customer) receives his first service. This quantity consists of two components:

1. the mean amount of waiting work found upon his arrival that is handled before his first service, given by: $\sum_{i=1}^{\infty} \beta_i E \mathbf{X}_i^w$;

2. the mean amount of work currently in service:
$$\sum_{i=1}^{\infty} \rho_i \frac{\beta_i^{(2)}}{2\beta_i}$$
;

where \mathbf{X}_{i}^{w} denotes the number of *waiting* type-*i* customers. The expression for the second component follows by noting that, at an arbitrary epoch, a type-*i* customer is being served with probability ρ_{i} , while his residual service time has mean $\beta_{i}^{(2)}/2\beta_{i}$. It may now be seen that,

$$E\mathbf{S}_1 = \sum_{i=1}^{\infty} \boldsymbol{\beta}_i E\mathbf{X}_i^w + \sum_{i=1}^{\infty} \rho_i \frac{\boldsymbol{\beta}_i^{(2)}}{2\boldsymbol{\beta}_i} + \boldsymbol{\beta}_1 .$$

With $E\mathbf{X}_i^w = E\mathbf{X}_i - \rho_i$ we obtain:

$$E\mathbf{S}_{1} = \sum_{j=1}^{\infty} \boldsymbol{\beta}_{j} E\mathbf{X}_{j} + \sum_{j=1}^{\infty} (\frac{\boldsymbol{\beta}_{j}^{(2)}}{2\boldsymbol{\beta}_{j}} - \boldsymbol{\beta}_{j})\boldsymbol{\rho}_{j} + \boldsymbol{\beta}_{1} .$$
(2.46)

 ES_{i+1} is composed of mean service times of "old" customers (customers who were already present at the first arrival of the tagged customer) and of customers who have arrived during the first *i* sojourn times. It is easily seen that the mean number of old type-*j* customers still present in the queue (as type*j*+*i* customers) immediately after the *i*-th service of the tagged customer is given by $\frac{q(j+i)}{q(j)}EX_j$. The mean number of customers that arrived during the tagged customer's *j*-th sojourn time and that are still present (as type-(i-j+1) customers) at the end of his *i*-th service is $\lambda q(i-j+1)ES_j$. Hence

$$E\mathbf{S}_{i+1} = \sum_{j=1}^{\infty} \frac{q(j+i)}{q(j)} \beta_{j+i} E\mathbf{X}_j + \lambda \sum_{j=1}^{i} q(i-j+1)\beta_{i-j+1} E\mathbf{S}_j + \beta_{i+1} , \quad (2.47)$$
$$i = 1, 2, \dots$$

The mean number of type-*j* customers in the system and the *j*-th sojourn time

can be related to each other by Little's formula (see e.g. Kleinrock [1975]):

$$E\mathbf{X}_{j} = \lambda q(j) E\mathbf{S}_{j}.$$

Substituting this into (2.46) and (2.47) leads to

$$E\mathbf{S}_{1} = \sum_{j=1}^{\infty} \rho_{j} E\mathbf{S}_{j} + \sum_{j=1}^{\infty} (\frac{\beta_{j}^{(2)}}{2\beta_{j}} - \beta_{j})\rho_{j} + \beta_{1}. \qquad (2.48)$$

$$E\mathbf{S}_{i+1} = \sum_{j=1}^{\infty} \rho_{j+i} E\mathbf{S}_j + \sum_{j=1}^{i} \rho_{i-j+1} E\mathbf{S}_j + \beta_{i+1}, \quad i = 1, 2, \dots .$$
(2.49)

Formulas (2.48) and (2.49) represent an infinite set of linear equations in ES_1, ES_2, \ldots . For some special cases this set of equations can be easily solved explicitly. In the next subsections we shall consider two cases which yield interesting results for the successive mean sojourn times of a customer.

2.6.2 Special case: M/G/1 queue with deterministic feedback

In this subsection we assume that the customers require exactly N services and that all service time distributions are the same, i.e. $B_i(.) \equiv B(.)$ and p(i)=1, i=1,...,N-1, p(N)=0, in the general model. Let S denote the total sojourn time after N services. The equations (2.48) and (2.49) now become:

$$E\mathbf{S}_{1} = \lambda\beta E\mathbf{S} + \frac{\lambda}{2}N(\beta^{(2)} - 2\beta^{2}) + \beta, \qquad (2.50)$$

$$E\mathbf{S}_{i+1} = \lambda \beta \sum_{j=1}^{N-i} E\mathbf{S}_j + \lambda \beta \sum_{j=1}^{i} E\mathbf{S}_j + \beta, \quad i = 1, ..., N-1.$$
(2.51)

Due to the symmetry in (2.51) we have that

$$ES_{i+1} = ES_{N-i+1}, \quad i=1,...,N-1.$$

Subtracting ES_i from ES_{i+1} , we obtain

$$E\mathbf{S}_{i+1} - E\mathbf{S}_i = -\lambda\beta E\mathbf{S}_{N-i+1} + \lambda\beta E\mathbf{S}_i = -\lambda\beta(E\mathbf{S}_{i+1} - E\mathbf{S}_i),$$

$$i = 2, ..., N - 1.$$

Hence, $ES_i = ES_{i+1}$, i = 2, ..., N - 1, and, interestingly, we have

$$E\mathbf{S}_2 = E\mathbf{S}_3 = \cdots = E\mathbf{S}_N. \tag{2.52}$$

Now from (2.50) and (2.52):

$$E\mathbf{S}_1 = \lambda\beta E\mathbf{S}_1 + (N-1)\lambda\beta E\mathbf{S}_2 + \frac{\lambda}{2}N(\beta^{(2)}-2\beta^2) + \beta.$$
(2.53)

And from (2.51) and (2.52):

$$E\mathbf{S}_2 = 2\lambda\beta E\mathbf{S}_1 + (N-2)\lambda\beta E\mathbf{S}_2 + \beta.$$
(2.54)

Solving equations (2.53) and (2.54) yields:

$$E\mathbf{S}_{1} = \frac{\beta}{1 - N\lambda\beta} + \frac{(1 - (N - 2)\lambda\beta)\frac{\lambda}{2}N(\beta^{(2)} - 2\beta^{2})}{(1 + \lambda\beta)(1 - N\lambda\beta)}, \qquad (2.55)$$

$$E\mathbf{S}_{2} = E\mathbf{S}_{3} = \cdots = E\mathbf{S}_{N} = \frac{\beta}{1-N\lambda\beta} + \frac{\lambda^{2}\beta N(\beta^{(2)}-2\beta^{2})}{(1+\lambda\beta)(1-N\lambda\beta)} \cdot (2.56)$$

Hence

$$E\mathbf{S} = \frac{N\beta}{1 - N\lambda\beta} + \frac{1 + N\lambda\beta}{1 - N\lambda\beta} \frac{\lambda}{2} \frac{N(\beta^{(2)} - 2\beta^2)}{1 + \lambda\beta}.$$
(2.57)

For N=1 (2.57) gives the well-known formula for the mean sojourn time in the standard M/G/1 queue, (see e.g. Cohen [1982])

$$E\mathbf{S} = \frac{\lambda\beta^{(2)}}{2(1-\lambda\beta)} + \beta, \qquad (2.58)$$

as could be expected.

Observe from (2.55) and (2.56) that $E\mathbf{S}_1 = E\mathbf{S}_2$ if the service times are negative exponentially distributed (i.e. $\beta^{(2)} = 2\beta^2$), cf. Section 2.3. Noting that $(1-(N-2)\lambda\beta)\lambda N/2 > \lambda^2\beta N$ (use $N\lambda\beta = \rho < 1$) it follows from (2.55) and (2.56) that $E\mathbf{S}_1 < E\mathbf{S}_2$ if $\beta^{(2)} < 2\beta^2$ and $E\mathbf{S}_1 > E\mathbf{S}_2$ if $\beta^{(2)} > 2\beta^2$.

Remark 2.5

Note that, in fact, for the derivation of (2.52) it suffices to assume that all mean sojourn times are equal, i.e. $\beta_i = \beta$, i = 1, ..., N.

2.6.3 Special case: M/G/1 queue with Bernoulli feedback

In this subsection we consider the M/G/1 queue with Bernoulli feedback, i.e. $B_i(\cdot) \equiv B(\cdot)$ and $p(i) \equiv p$ in the general model. For this case the set of equations (2.48) and (2.49) reads

$$E\mathbf{S}_{1} = \lambda \beta \sum_{j=1}^{\infty} p^{j-1} E\mathbf{S}_{j} + \frac{\lambda}{2} (\beta^{(2)} - 2\beta^{2}) \sum_{j=1}^{\infty} p^{j-1} + \beta.$$
(2.59)

$$E\mathbf{S}_{i+1} = \lambda \beta \sum_{j=1}^{\infty} p^{j+i-1} E\mathbf{S}_j + \lambda \beta \sum_{j=1}^{i} p^{i-j} E\mathbf{S}_j + \beta , \quad i = 1, 2, \dots .$$
(2.60)

Introducing

.

$$M := \frac{\lambda}{2} (\beta^{(2)} - 2\beta^2) \frac{1}{1-p},$$

$$M_j := \frac{1}{M} \left[E \mathbf{S}_j - \frac{\beta}{1 - \lambda \beta / (1-p)} \right], \quad j = 1, 2, ...,$$

we can rewrite (2.59) and (2.60) into

$$M_1 = \lambda \beta \sum_{j=1}^{\infty} p^{j-1} M_j + 1, \qquad (2.61)$$

$$M_{i+1} = \lambda \beta \sum_{j=0}^{i-1} p^j M_{i-j} + \lambda \beta \sum_{j=1}^{\infty} p^{i+j-1} M_j, \quad i = 1, 2, \dots.$$
(2.62)

From (2.62),

$$M_{i+2} = (\lambda \beta + p)M_{i+1} = \cdots = (\lambda \beta + p)^i M_2, \quad i = 0, 1, \dots$$
 (2.63)

Substitution of (2.63) into (2.61) and (2.62) leads to a set of two linear equations with two unknowns M_1 and M_2 ; these equations yield

$$M_{1} = \frac{1 - p - \lambda \beta p}{1 - p - \lambda \beta},$$
$$M_{2} = \lambda \beta \frac{1 - p (\lambda \beta + p)}{1 - p - \lambda \beta},$$

so

$$E\mathbf{S}_{1} = \frac{\beta}{1-\lambda\beta/(1-p)} + \frac{\lambda}{2}(\beta^{(2)}-2\beta^{2})\frac{1}{1-p}\frac{1-p-\lambda\beta p}{1-p-\lambda\beta},$$
(2.64)

$$E\mathbf{S}_{k} = \frac{\beta}{1-\lambda\beta/(1-p)} +$$

$$\frac{\lambda}{2}(\beta^{(2)}-2\beta^{2})\frac{1}{1-p}\lambda\beta\frac{1-p(\lambda\beta+p)}{1-p-\lambda\beta}(\lambda\beta+p)^{k-2}, \quad k = 2,3,\dots.$$
(2.65)

From (2.64) and (2.65) it follows that the successive mean sojourn times are all equal if the service times are exponentially distributed ($\beta^{(2)}=2\beta^2$), cf. Section 2.3. Using $\lambda\beta/(1-p)=\rho<1$ it can be shown that $ES_1 < ES_2$ if $\beta^{(2)} < 2\beta^2$ and $ES_1 > ES_2$ if $\beta^{(2)} > 2\beta^2$. In the previous subsection we have observed similar properties for the first two sojourn times in the M/G/1 queue with deterministic feedback. Apparently, the difference between ES_1 and ES_2 is due to the fact that a customer's first sojourn time may contain a residual service (of mean length $\beta^{(2)}/(2\beta)$) while the second sojourn time only consists of complete service times, cf. (2.50), (2.51) and (2.59), (2.60). From (2.65) it is seen that, for $k \rightarrow \infty$,

$$E\mathbf{S}_k \to \frac{\beta}{1 - \lambda\beta/(1-p)}, \qquad (2.66)$$

which is the mean sojourn time per visit in the case of a negative exponential service time distribution, cf. (2.20).

Finally, the mean total sojourn time, ES, is given by

$$ES = \sum_{i=1}^{\infty} p^{i-1} ES_i = \frac{\beta}{1-p-\lambda\beta} + \frac{\lambda}{2} (\beta^{(2)} - 2\beta^2) \frac{1}{1-p-\lambda\beta}.$$
 (2.67)

This result has been obtained before by Takács [1963].

Appendix 2.1

In this appendix we derive Formula (2.8). For ease of notation we introduce the service time distribution function $B(t):=1-e^{-t/\beta}$; the *n*-fold convolution of $B(\cdot)$ is denoted by $B(\cdot)^{n^*}$, n=1,2,...

The derivation of (2.8) is based on the fact that $(\mathbf{X}_{1}^{(i+1)}, \ldots, \mathbf{X}_{N}^{(i+1)})$, which determines the distribution of \mathbf{S}_{i+2} , is conditionally independent of $\{(\mathbf{X}_{1}^{(0)}, \ldots, \mathbf{X}_{N}^{(0)}), \ldots, (\mathbf{X}_{1}^{(i-1)}, \ldots, \mathbf{X}_{N}^{(i-1)}); \mathbf{S}_{1}, \ldots, \mathbf{S}_{i}\}$ given $\{(\mathbf{X}_{1}^{(i)}, \ldots, \mathbf{X}_{N}^{(i)}); \mathbf{S}_{i+1}\}$, $i=1,\ldots,k-1$. Using this property it is easily seen that, conditioning on the number of type-*j* arrivals, $n_{j}^{(m)}$, during the *m*-th sojourn time,

$$E\{e^{-(\omega_{1}\mathbf{S}_{1}+...+\omega_{k}\mathbf{S}_{k})}(z_{1,0}^{\mathbf{X}_{1}^{(0)}}\cdots z_{N,0}^{\mathbf{X}_{N}^{(0)}})\cdots (z_{1,k}^{\mathbf{X}_{1}^{(k)}}\cdots z_{N,k}^{\mathbf{X}_{N}^{(k)}}) | \mathbf{X}_{1}^{(0)}=x_{1},\ldots,\mathbf{X}_{N}^{(0)}=x_{N}\}=$$

$$z_{1,0}^{x_{1}} \cdots z_{N,0}^{x_{N}} \int_{t_{1}=0}^{\infty} e^{-\omega_{1}t_{1}} \int_{t_{2}=0}^{\infty} e^{-\omega_{2}t_{2}} \cdots \int_{t_{k-1}=0}^{\infty} e^{-\omega_{k-1}t_{k-1}} \int_{t_{k}=0}^{\infty} e^{-\omega_{k}t_{k}}$$

$$\sum_{n_{1}^{(0)}=0}^{\infty} e^{-\lambda t_{1}} \frac{(\lambda t_{1})^{n_{1}^{(0)}}}{n_{1}^{(1)}!} z_{1,1}^{n_{1}^{(1)}} \left[\prod_{j=1}^{N} \sum_{n_{j+1}^{(0)}=0}^{x_{j}} \left[\frac{x_{j}}{n_{j+1}^{(1)}} \right] p(j)^{n_{j+1}^{(1)}} (1-p(j))^{x_{j}-n_{j+1}^{(1)}} z_{j+1,1}^{n_{j+1}^{(1)}} \right]$$

$$\prod_{m=2}^{k} \left\{ \sum_{n_{1}^{(m)}=0}^{\infty} e^{-\lambda t_{m}} \frac{(\lambda t_{m})^{n_{1}^{(m)}}}{n_{1}^{(m)}!} z_{1,m}^{n_{1}^{(m)}} \left[\prod_{j=1}^{N+m-1} \sum_{n_{j+1}^{(m)}=0}^{n_{m}^{(m-1)}} \left[\frac{n_{j}^{(m-1)}}{n_{j+1}^{(m)}} \right] p(j)^{n_{j+1}^{(m)}} (1-p(j))^{n_{j}^{(m-1)}-n_{j+1}^{(m)}} z_{j+1,m}^{n_{j+1}^{(m)}} \right] dB(t_{k})^{(1+n_{1}^{(k-1)}+\dots+n_{N+k-1}^{(k-1)})^{*}} dB(t_{k-1})^{(1+n_{1}^{(k-2)}+\dots+n_{N+k-2}^{(k-2)})^{*}} \cdots dB(t_{2})^{(1+n_{1}^{(1)}+\dots+n_{N+1}^{(1)})^{*}} dB(t_{1})^{(1+x_{1}+\dots+x_{N})^{*}}.$$

Note that by definition $z_{i,j}:=z_{N,j}$, i=N+1,...,N+k, j=1,...,k. We first evaluate the integral with respect to t_k , obtaining

$$E\{e^{-(\omega_1\mathbf{S}_1+\ldots+\omega_k\mathbf{S}_k)}(z_{1,0}^{\mathbf{X}_1^{(0)}}\cdots z_{N,0}^{\mathbf{X}_N^{(0)}})\cdots (z_{1,k}^{\mathbf{X}_1^{(k)}}\cdots z_{N,k}^{\mathbf{X}_N^{(k)}}) | \mathbf{X}_1^{(0)} = x_1,\ldots,\mathbf{X}_N^{(0)} = x_N\} =$$

$$z_{1,0}^{x_{1}} \cdots z_{N,0}^{x_{N}} \int_{t_{1}=0}^{\infty} e^{-\omega_{1}t_{1}} \int_{t_{2}=0}^{\infty} e^{-\omega_{2}t_{2}} \cdots \int_{t_{k-1}=0}^{\infty} e^{-\omega_{k-1}t_{k-1}}$$

$$\sum_{n_{1}^{(0)}=0}^{\infty} e^{-\lambda t_{1}} \frac{(\lambda t_{1})^{n_{1}^{(0)}}}{n_{1}^{(1)}!} z_{1,1}^{n_{1}^{(0)}} \left[\prod_{j=1}^{N} \sum_{n_{j+1}^{(0)}=0}^{x_{j}} \left[\frac{x_{j}}{n_{j+1}^{(1)}} \right] p(j)^{n_{j+1}^{(0)}} (1-p(j))^{x_{j}-n_{j+1}^{(0)}} z_{j+1,1}^{n_{j+1}^{(0)}} \right]$$

$$\prod_{m=2}^{k-1} \left\{ \sum_{n_{1}^{(m)}=0}^{\infty} e^{-\lambda t_{m}} \frac{(\lambda t_{m})^{n_{1}^{(m)}}}{n_{1}^{(m)}!} z_{1,m}^{n_{1}^{(m)}} \left[\prod_{j=1}^{N+m-1} \sum_{n_{j+1}^{(m)}=0}^{n_{j}^{(m-1)}} \left[n_{j+1}^{(m-1)} \right] p(j)^{n_{j+1}^{(m)}} (1-p(j))^{n_{j}^{(m-1)}-n_{j+1}^{(m)}} z_{j+1,m}^{n_{j+1}} \right] \right\}$$

$$[1+\beta\{\omega_{k}+\lambda(1-z_{1,k})\}]^{-(1+n_{1}^{(k-1)}+\ldots+n_{N+k-1}^{(k-1)})}\prod_{j=1}^{N+k-1}(p(j)z_{j+1,k}+1-p(j))^{n_{j}^{(k-1)}}$$

$$dB(t_{k-1})^{(1+n_{1}^{(k-2)}+\ldots+n_{N+k-2}^{(k-2)})^{*}}\cdots dB(t_{2})^{(1+n_{1}^{(1)}+\ldots+n_{N+1}^{(1)})^{*}}dB(t_{1})^{(1+x_{1}+\ldots+x_{N})^{*}}=$$

$$z_{1,0}^{x_{1}} \cdots z_{N,0}^{x_{N}} A_{k}^{N}(1,\omega,z) \int_{t_{1}=0}^{\infty} e^{-\omega_{1}t_{1}} \int_{t_{2}=0}^{\infty} e^{-\omega_{2}t_{2}} \cdots \int_{t_{k-1}=0}^{\infty} e^{-\omega_{k-1}t_{k-1}}$$

$$\sum_{n_{1}^{(1)}=0}^{\infty} e^{-\lambda t_{1}} \frac{(\lambda t_{1})^{n_{1}^{(1)}}}{n_{1}^{(1)}!} z_{1,1}^{n_{1}^{(1)}} \left[\prod_{j=1}^{N} \sum_{n_{j+1}^{(1)}=0}^{x_{j}} \left[\frac{x_{j}}{n_{j+1}^{(1)}+1} \right] p(j)^{n_{j+1}^{(1)}} (1-p(j))^{x_{j}-n_{j+1}^{(1)}} z_{j+1,1}^{n_{j+1}^{(1)}} \right]$$

$$\prod_{m=2}^{k-2} \left\{ \sum_{n_{1}^{(m)}=0}^{\infty} e^{-\lambda t_{m}} \frac{(\lambda t_{m})^{n_{1}^{(m)}}}{n_{1}^{(m)}!} z_{1,m}^{n_{1}^{(m)}} \left[\prod_{j=1}^{N+m-1} \sum_{n_{j+1}^{(m)}=0}^{n_{j+1}^{(m-1)}} \left[n_{j+1}^{(m-1)} \right] p(j)^{n_{j+1}^{(m)}} (1-p(j))^{n_{j}^{(m-1)}-n_{j+1}^{(m)}} z_{j+1,m}^{n_{j+1}^{(m)}} \right] \right\}$$

$$e^{-\lambda t_{k-1}(1-z_{1,k-1}A_{k}^{N}(1,\omega,z)[p(1)z_{2,k}+1-p(1)])}$$

$$\prod_{j=1}^{N+k-2} (A_{k}^{N}(1,\omega,z)[p(j+1)z_{j+2,k}+1-p(j+1)]p(j)z_{j+1,k-1}+1-p(j)]^{n_{j}^{(k-2)}}$$

$$dB(t_{k-1})^{(1+n_{1}^{(k-2)}+\ldots+n_{N+k-2}^{(k-2)}*} \cdots dB(t_{2})^{(1+n_{1}^{(1)}+\ldots+n_{N+1}^{(1)})*} dB(t_{1})^{(1+x_{1}+\ldots+x_{N})*}.$$

Next the integral with respect to t_{k-1} is evaluated, yielding

$$E\{e^{-(\omega_{1}\mathbf{S}_{1}+\ldots+\omega_{k}\mathbf{S}_{k})}(z_{1,0}^{\mathbf{X}_{1}^{(0)}}\cdots z_{N,0}^{\mathbf{X}_{N}^{(0)}})\cdots (z_{1,k}^{\mathbf{X}_{1}^{(k)}}\cdots z_{N,k}^{\mathbf{X}_{N}^{(k)}}) | \mathbf{X}_{1}^{(0)}=x_{1},\ldots,\mathbf{X}_{N}^{(0)}=x_{N}\} =$$

$$z_{1,0}^{x_{1}} \cdots z_{N,0}^{x_{N}} A_{k}^{N}(1,\omega,z) \int_{t_{1}=0}^{\infty} e^{-\omega_{1}t_{1}} \int_{t_{2}=0}^{\infty} e^{-\omega_{2}t_{2}} \cdots \int_{t_{k-2}=0}^{\infty} e^{-\omega_{k-2}t_{k-2}}$$

$$\sum_{n_{1}^{(0)}=0}^{\infty} e^{-\lambda t_{1}} \frac{(\lambda t_{1})^{n_{1}^{(0)}}}{n_{1}^{(1)}!} z_{1,1}^{n_{1}^{(0)}} \left[\prod_{j=1}^{N} \sum_{n_{j+1}^{(0)}=0}^{x_{j}} \left[\frac{x_{j}}{n_{j+1}^{(1)}} \right] p(j)^{n_{j+1}^{(1)}} (1-p(j))^{x_{j}-n_{j+1}^{(0)}} z_{j+1,1}^{n_{j+1}^{(0)}} \right]$$

$$\prod_{m=2}^{k-2} \left\{ \sum_{n_{1}^{(m)}=0}^{\infty} e^{-\lambda t_{m}} \frac{(\lambda t_{m})^{n_{1}^{(m)}}}{n_{1}^{(m)}!} z_{1,m}^{n_{1}^{(m)}} \left[\prod_{j=1}^{N+m-1} \sum_{n_{j+1}^{(m)}=0}^{n_{j}^{(m-1)}} \left[n_{j+1}^{(m-1)} \right] p(j)^{n_{j+1}^{(m)}} (1-p(j))^{n_{j}^{(m-1)}-n_{j+1}^{(m)}} z_{j+1,m}^{n_{j+1}} \right] \right\}$$

$$[1+\beta\{\omega_{k-1}+\lambda-\lambda z_{1,k-1}A_{k}^{N}(1,\omega,z)[p(1)z_{2,k}+1-p(1)]\}]^{-(1+n_{1}^{(k-2)}+\ldots+n_{N+k-2}^{(k-2)})}$$

$$\sum_{j=1}^{N+k-2} (A_{k}^{N}(1,\omega,z)[p(j+1)z_{j+2,k}+1-p(j+1)]p(j)z_{j+1,k-1}+1-p(j)]^{n_{j}^{(k-2)}}$$

•

$$dB(t_{k-2})^{(1+n_{1}^{(k-3)}+...+n_{N+k-3}^{(k-3)})^{*}} \cdots dB(t_{2})^{(1+n_{1}^{(1)}+...+n_{N+1}^{(1)})^{*}} dB(t_{1})^{(1+x_{1}+...+x_{N})^{*}} =$$

$$z_{1,0}^{x_{1,0}} \cdots z_{N,0}^{x_{N}} A_{k}^{N}(1,\omega,z) A_{k}^{N}(2,\omega,z) \int_{t_{1}=0}^{\infty} e^{-\omega_{1}t_{1}} \int_{t_{2}=0}^{\infty} e^{-\omega_{2}t_{2}} \cdots \int_{t_{k-2}=0}^{\infty} e^{-\omega_{k-2}t_{k-2}}$$

$$\sum_{n_{1}^{(0)}=0}^{\infty} e^{-\lambda t_{1}} \frac{(\lambda t_{1})^{n_{1}^{(0)}}}{n_{1}^{(1)}!} z_{1,1}^{n_{1}^{(0)}} \left[\prod_{j=1}^{N} \sum_{n_{j+1}^{(0)}=0}^{x_{j}} \left[n_{j+1}^{x_{j}}\right] p(j)^{n_{j+1}^{(0)}} (1-p(j))^{x_{j}-n_{j+1}^{(0)}} z_{j+1,1}^{n_{j+1}^{(0)}}\right]$$

$$\prod_{m=2}^{k-2} \left\{\sum_{n_{1}^{(m)}=0}^{\infty} e^{-\lambda t_{m}} \frac{(\lambda t_{m})^{n_{1}^{(m)}}}{n_{1}^{(m)}!} z_{1,m}^{n_{1}^{(m)}} \left[\prod_{j=1}^{N+m-1} \sum_{n_{j+1}^{(m)}=0}^{n_{m-1}^{(m)}} \left[n_{j+1}^{(m-1)}\right] p(j)^{n_{j+1}^{(m)}} (1-p(j))^{n_{j}^{(m-1)}-n_{j+1}^{(m)}} z_{j+1,m}^{n_{j+1}^{(m)}}\right]\right\}$$

$$\prod_{j=1}^{N+k-2} \left(A_{k}^{N}(2,\omega,z) [A_{k}^{N}(1,\omega,z)] p(j+1) z_{j+2,k} + 1 - p(j+1)] p(j) z_{j+1,k-1} + 1 - p(j)] \right)^{n_{j}^{(k-2)}}$$

$$dB(t_{k-2})^{(1+n_{1}^{(k-3)}+...+n_{N+k-3}^{(k-3)})^{*}} \cdots dB(t_{2})^{(1+n_{1}^{(1)}+...+n_{N+1}^{(m)})^{*}} dB(t_{1})^{(1+x_{1}+...+x_{N})^{*}}.$$

We now sum over $n_1^{(k-2)}, \ldots, n_{N+k-2}^{(k-2)}$ and subsequently integrate with respect to t_{k-2} , thus obtaining $A_k^N(3,\omega,z)$ terms; etc.; finally the summations over $n_1^{(1)}, \ldots, n_N^{(1)}$ and the integration over t_1 are performed, which gives rise to the $(f_k^N(i,\omega,z))^{x_i}$ contribution in Formula (2.8).

Chapter 3

THE M/G/1 PROCESSOR SHARING QUEUE AS A LIMITING MODEL OF THE M/M/1 FEEDBACK QUEUE

3.1 INTRODUCTION

In the previous chapter we have regarded the feedback model as an M/M/1 queue in which after each service it is decided whether or not the customer is fed back. In this chapter we consider the same model from another point of view, viz. as a round robin (time sharing) model in which a customer's service demand requires a stochastic number of exponentially distributed service quanta with mean length β . Obviously, the service requirements are completely determined by the feedback probabilities $p(1), p(2), \dots$, as defined in Chapter 2. From this point of view it is intuitively clear that if the mean service time β shrinks to zero while the feedback probabilities go to one such that a customer's total required service time remains unchanged, the behaviour of the feedback queue approaches that of the M/G/1 processor sharing (PS) queue. Different choices of the feedback probabilities lead to different service time distributions in the PS queue.

The queue length process in a round robin type of queue is usually less amenable to mathematical analysis than the queue length process in its limiting case, a PS queue. This has been the main reason for the queueing analysis of processor sharing, see Kleinrock [1976]. Sakata et al. [1969] showed that the distribution of the queue length, X^{PS} , in the M/G/1 PS queue is independent of the distribution of the required service time apart from its first moment:

$$Pr\{\mathbf{X}^{PS}=j\} = (1-\rho)\rho^{j}, \quad j=0,1,2,\dots,$$
(3.1)

with ρ the offered load per unit of time. The determination of the *sojourn time* distribution in a PS queue has turned out to be a much harder problem. Only recently the sojourn time distribution in the M/G/1 PS queue has been derived, cf. Yashkov [1983], Ott [1984], Schassberger [1984], and the survey of Yashkov [1987]. We refer to Section 1.5 for a brief description of the approaches used by these authors. The approach presented in this chapter is

new: via a limiting procedure we obtain sojourn time results for the M/G/1 PS queue from known sojourn time results (obtained in Chapter 2) for the M/M/1 queue with general feedback.

The limiting procedure described above was first proposed by Van den Berg et al. [1989A]. In that paper it is shown how the distribution of the sojourn time in the M/D/1 PS queue follows immediately (by taking appropriate limits) from the sojourn time distribution in the M/M/1 queue with deterministic feedback. In Van den Berg and Boxma [1989B] this method has been extended to the analysis of the processor sharing queue with general service times. In these papers the authors concluded on intuitive grounds that the performance measures such as the sojourn time in the feedback model converge to the corresponding performance measures in the processor sharing queue. Only very recently a formal proof of this convergence has been given by Resing et al. [1989]. They present a probabilistic coupling between the M/G/1 PS queue and the approximating sequence of M/M/1 feedback queues, which shows that the sojourn time of the *n*-th customer in the feedback model converges almost surely to the corresponding quantity in the PS model. From this result they conclude the distributional convergence of the steady state sojourn times. The proof partially follows the same line of thought as Schassberger [1984].

The organization of the rest of this chapter is as follows. Section 3.2 contains some definitions and restates those results of Chapter 2 that are essential for the analysis in Section 3.3. In the latter section we study sojourn times in the M/G/1 PS queue, by taking appropriate limits in the M/M/1 queue with feedback. We first derive the mean sojourn time (Subsection 3.3.2) and the sojourn time variance (Subsection 3.3.3). Next, in Subsection 3.3.4, it is shown how the LST of the distribution of the sojourn time in the M/G/1 PS queue can be obtained. Section 3.2, Subsection 3.3.2 and Subsection 3.3.3 are mainly based on Van den Berg and Boxma [1989B]. In Section 3.4 we consider the same feedback model as in Chapter 2 but with state dependent service rates. It is shown how a similar limiting procedure leads to the analysis of the M/G/1 queue with the so called 'generalized processor sharing' service discipline. This section is restricted to the derivation of mean sojourn times. In the last section of this chapter, Section 3.5, we analyze sojourn times in the M/G/1 PS queue with feedback. Using the (M/M/1 FCFS) feedback results obtained in Chapter 2 and applying the limiting procedure we derive new results for the correlation coefficients of the successive sojourn times of a tagged customer.

3.2 Preliminary results

We consider the M/M/1 feedback queue introduced and analyzed in Chapter 2. In Section 2.2 we made the assumption that the feedback probabilities of a customer remain constant after a finite number of services. However, from the ultimate result for the joint sojourn time distribution ((2.18)) it appeared that this assumption is not needed, see the discussion below (2.19). In the

following no restrictions will be put on the structure of the feedback probabilities. So, we rewrite Definition (2.1):

$$q(1) := 1,$$

$$q(i) := \prod_{j=1}^{i-1} p(j), \quad i = 2, 3, \dots.$$
(3.2)

Obviously, for stability it is required that the q(i)'s satisfy:

$$\lambda \beta \sum_{i=1}^{\infty} q(i) =: \rho < 1.$$

For future reference we introduce the generating function of the probabilities of visiting the queue exactly *i* times, i = 1, 2, ...:

$$Q(z) := \sum_{i=1}^{\infty} q(i)(1-p(i))z^{i}, \quad |z| \le 1.$$
(3.3)

We now recall those feedback results obtained in Chapter 2 which are essential for the analysis in the next sections. Some of the expressions will be rewritten such that they are more suitable for analyzing sojourn times in the M/G/1 PS queue.

First, we slightly rewrite formula (2.18) and use it to obtain a convenient expression for the LST of the total sojourn time after k services. Replacing the term $M_k(k, \omega)$ in the denominator of (2.18) by the RHS of (2.19) (with i = k) and substituting $\omega_j = \omega_0$, j = 1, ..., k, it is easily seen that (using the notation introduced in (3.2)), for Re $\omega_0 \ge 0$, k = 1, 2, ...,

(3.4)

$$E\{e^{-\omega_0\mathbf{S}^{(\kappa)}}\} =$$

$$\frac{1-\rho}{(1+\beta\omega_0)M_{k-1} - \lambda\beta\sum_{j=1}^{k-2}q(k-j-1)M_j - (\rho-\lambda\beta\sum_{i=1}^{k-2}q(i))},$$

where,

$$M_{0} := 1,$$

$$M_{n} := (1 + \beta \omega_{0} + \lambda \beta) M_{n-1} - \lambda \beta \left[q(n-1) + \sum_{l=2}^{n-1} q(n-l)(1 - p(n-l)) M_{l-1} \right],$$

$$n = 1, 2, \dots.$$
(3.5)

(q(0) := 1)

In Chapter 2 it has been found that the mean total sojourn time after k services is linear in k, cf. (2.25):

$$E\{\mathbf{S}^{(k)}\} = k \frac{\beta}{1-\rho}, \quad k = 1, 2, \dots.$$
 (3.6)

Formula (2.26) gives the variance of the total sojourn time after k services:

$$war(\mathbf{S}^{(k)}) = \left(\frac{\beta}{1-\rho}\right)^2 [k^2 - 2(1-\rho) \sum_{j=1}^{k-1} j C_{k-j}], \quad k = 1, 2, ...,$$
(3.7)

where C_1, \ldots, C_{k-1} can be successively obtained from (2.22). This recurrence relation for the C_n can be simplified in the following way. Noting that q(n-l)(1-p(n-l))=q(n-l)-q(n-l+1), and splitting the sum in (2.22) we obtain

$$C_n - \lambda \beta \sum_{l=2}^n q(n-l+1)C_{l-1} = C_{n-1} - \lambda \beta \sum_{l=2}^{n-1} q(n-l)C_{l-1}.$$

Now, using $C_1 = 1$, it is easily seen that

,

.

$$C_{1} = 1 , \qquad (3.8)$$

$$C_{n} = 1 + \lambda \beta \sum_{l=1}^{n-1} q(n-l)C_{l} , \quad n = 2, 3... .$$

Taking generating functions and using (3.3) leads to

$$C(z) := \sum_{n=1}^{\infty} C_n z^n = \frac{z}{(1-z)(1-\lambda\beta\frac{z}{1-z}(1-Q(z)))}, \quad |z| < 1.$$
(3.9)

The sequence $C_1, C_2,...$ is non-decreasing and, cf. (2.24), limited from above. Hence $\lim_{n\to\infty} C_n$ exists; an Abelian theorem now implies that (cf. Titchmarsh [1952])

$$\lim_{n \to \infty} C_n = \lim_{z \to 1} (1 - z)C(z) = \frac{1}{1 - \rho}.$$
(3.10)

For future use we also introduce the generating function of the M_n 's. From (3.5) it follows that

$$M(z) := \sum_{n=1}^{\infty} M_n z^n = z \frac{1 + \beta \omega_0 - \lambda \beta \frac{z}{1 - z} (1 - Q(z))}{1 - z (1 + \beta \omega_0 + \lambda \beta) + \lambda \beta z Q(z)}, \quad |z| < 1.$$
(3.11)

3.3 The M/G/1 processor sharing queue

3.3.1 The limiting procedure

In this section we show how the feedback results collected in Section 3.2 can be used to analyze the sojourn time in the M/G/1 PS queue. We apply a limiting procedure, in which $\beta \rightarrow 0$ while the feedback probabilities approach one in such a way that the mean total required service time, $\hat{\beta}$, remains a positive constant. We restrict ourself to those service times, τ^{PS} , in the PS queue which are composed of negative exponentially distributed stages:

$$E\{\exp(-\omega_0 \tau^{PS})\} = \sum_{j=1}^{m} \alpha_j \prod_{i=1}^{r_j} \frac{1}{1 + \hat{\beta}_{ij}\omega_0},$$
(3.12)

with $\alpha_1, \ldots, \alpha_m > 0$, $\sum_{j=1}^m \alpha_j = 1, r_1, \ldots, r_m$ positive integers (cf. Kleinrock [1975], p. 145); note that this class of distributions contains the Erlang, hyperexponential and Coxian distributions, and that arbitrary probability distributions of nonnegative random variables can be arbitrarily closely approximated by distributions from this class (cf. Tijms [1986], p. 398). This choice of service time distribution for the PS queue enables us to choose the feedback probabilities (hence Q(z)) such that τ^{PS} and the total required service time τ^{FB} in the feedback queue have exactly the same distribution - not just in the limit $\beta \rightarrow 0$, but for a wide range of values of β . Observe that, cf. (3.3),

$$E\{\exp(-\omega_0 \boldsymbol{\tau}^{FB})\} = \sum_{i=1}^{\infty} q(i)(1-p(i))(\frac{1}{1+\beta\omega_0})^i = Q(\frac{1}{1+\beta\omega_0}), \quad (3.13)$$

Re $\omega_0 \ge 0$.

Now choose

$$Q(z) = \sum_{j=1}^{m} \alpha_j \prod_{i=1}^{r_j} \frac{(1-p_{ij})z}{1-p_{ij}z},$$
(3.14)

with

$$p_{ij} = 1 - \beta / \hat{\beta}_{ij} > 0, \quad i = 1, ..., r_j, \ j = 1, ..., m.$$
 (3.15)

Then

$$E\{\exp(-\omega_0 \boldsymbol{\tau}^{FB})\} = \sum_{j=1}^{m} \alpha_j \prod_{i=1}^{r_j} \frac{\beta/\hat{\beta}_{ij}}{1+\beta\omega_0 - (1-\beta/\hat{\beta}_{ij})} =$$
(3.16)

$$\sum_{j=1}^m \alpha_j \prod_{i=1}^{r_j} \frac{1}{1+\hat{\beta}_{ij}\omega_0} = E\{\exp(-\omega_0 \tau^{PS})\}.$$

As an example, consider the case of Bernoulli feedback:

Q(z) = (1-p)z/(1-pz). In this case,

$$E\{\exp(-\omega_0 \tau^{PS})\} = E\{\exp(-\omega_0 \tau^{FB})\} = \frac{1}{1 + (\beta/(1-p))\omega_0} = \frac{1}{1 + \hat{\beta}\omega_0}.$$
 (3.17)

Hence the total required service times in both the feedback queue and the PS queue are negative exponentially distributed with mean $\hat{\beta} = \beta/(1-p)$.

When $\beta \rightarrow 0$, performance measures in the feedback queue clearly approach corresponding performance measures in the PS queue. Resing et al. [1989] give a formal proof of the convergence of the sojourn time. Note that the queue length distribution in both models is the same for the *whole* range of possible β values, cf. (3.1) and (2.5). Below we shall focus mainly on sojourn times. In particular we are interested in the sojourn time of a customer *conditioned* on his required service time. This is an important performance measure for time sharing systems like PS queues, cf. Kleinrock [1976]. We define for the PS queue

- $S^{PS}(x)$: conditional sojourn time of a customer with service demand x;

- S^{PS} : sojourn time of an arbitrary customer.

Obviously,

$$Pr\{\mathbf{S}^{PS} < s\} = \int_{x=0}^{\infty} Pr\{\mathbf{S}^{PS}(x) < s\} dPr\{\tau^{PS} < x\}, \quad s \ge 0.$$
(3.18)

The conditional sojourn time $\mathbf{S}^{PS}(x)$ can be derived from the total sojourn time after k services, $\mathbf{S}^{(k)}$, in the feedback queue as follows. Choose Q(z) for the feedback queue as in (3.14), (3.15), and consider a newly arriving customer, say C, who requires exactly k services. Then take $\beta = x/k$ and let $k \to \infty$. It is easily seen that the total required service time of C approaches the constant x. Indeed, the LST of C's total required service time equals $(1+\beta\omega_0)^{-k} = (1+x\omega_0/k)^{-k} \to e^{-x\omega_0}$. Hence, for $k\to\infty$, C can be viewed as a customer with service request x in the M/G/1 PS queue with service time distribution characterized by (3.12).

The limiting procedure described above will be applied below. We shall obtain results for the mean, the variance and the LST of the sojourn time in the PS queue from $E\{\mathbf{S}^{PS}(x)\} = \lim_{k \to \infty} E\{\mathbf{S}^{(k)}\}, var(\mathbf{S}^{PS}(x)) = \lim_{k \to \infty} var(\mathbf{S}^{(k)})$ and $E\{e^{-\omega_0 \mathbf{S}^{rs}(x)}\} = \lim_{k \to \infty} E\{e^{-\omega_0 \mathbf{S}^{(k)}}\}$ respectively. The results to be presented for the mean and the variance of the sojourn time are more general and more detailed than the results for the LST.

3.3.2 The mean sojourn time

In the M/G/1 PS queue, the mean sojourn time of a customer with service demand x is linear in x (cf. Kleinrock [1976]):

$$E\{\mathbf{S}^{PS}(x)\} = \frac{x}{1-\rho}.$$
(3.19)

We now show how this well known result can be easily obtained from the feedback results collected in Section 3.2. The mean total sojourn time $E\{\mathbf{S}^{(k)}\}$ of a customer who requires k services is linear in k, see (3.6). Apply the limiting procedure described in Subsection 3.3.1, taking $\beta = x/k$ and letting $k \to \infty$. Formula (3.19) now immediately follows from (3.6).

3.3.3 The variance of the sojourn time

The sojourn time variance for a customer with service request x in the M/G/1 PS queue, $var(\mathbf{S}^{PS}(x))$, can be obtained by applying the limiting procedure to (3.7). First, as an example, we derive $var(\mathbf{S}^{PS}(x))$ for the M/M/1 PS queue. Next the analysis is extended to the PS queue with general service times. This leads to a simple explicit expression for the asymptotic behaviour of $var(\mathbf{S}^{PS}(x))$ for very large $(x \to \infty)$ and very small $(x \to 0)$ service requests.

The M/M/1 PS queue

As observed in (3.17), the choice Q(z) = (1-p)z/(1-pz) leads, in the feedback queue as well as the PS queue, to a negative exponentially distributed total service time with mean $\beta/(1-p) = \hat{\beta}$. To obtain an explicit expression for $var(\mathbf{S}^{(k)})$, see (3.7), we derive C_n , n = 1, 2, ..., from (3.9). Substituting Q(z) = (1-p)z/(1-pz) into (3.9) yields

$$C(z) = z \frac{1 - pz}{(1 - z)(1 - (\lambda\beta + p)z)}.$$
(3.20)

Rewriting the right-hand side of (3.20) as

$$z(U_1\frac{1}{1-z} + U_2\frac{1}{1-(\lambda\beta+p)z}),$$

it follows that

$$C_n = U_1 + U_2 x_2^{n-1}, \quad n = 1, 2, ...,$$
 (3.21)

with $U_1=1/(1-\rho)$, $U_2=-\rho/(1-\rho)$, $x_2=\lambda\beta+p$. Substituting (3.21) into (3.7) yields

$$var(\mathbf{S}^{(k)}) = (\frac{\beta}{1-\rho})^2 [k - 2(1-\rho)U_2 \frac{x_2^k + k(1-x_2) - 1}{(1-x_2)^2}] = (3.22)$$

$$(\frac{\beta}{1-\rho})^2 [k + \frac{2\rho}{1-\rho}(\frac{k}{1-\rho} - \frac{1-(\lambda\beta+p)^k}{(1-\rho)(1-\rho)^2})].$$

Let $\hat{\beta}$ be the mean service time for the M/M/1 PS queue and let x be the service time of a tagged customer (cf. Subsection 3.3.2). Substitute $\beta = x/k$ and $p = 1 - x/k\hat{\beta}$ into (3.22). Letting $k \to \infty$ leads to $var(\mathbf{S}^{PS}(x))$:

$$var(\mathbf{S}^{PS}(x)) = \lim_{k \to \infty} var(\mathbf{S}^{(k)}) = \frac{2\rho\hat{\beta}x}{(1-\rho)^3} - \frac{2\rho\hat{\beta}^2}{(1-\rho)^4} [1 - e^{-x(1-\rho)/\hat{\beta}}], \quad (3.23)$$

a result previously obtained by Ott [1984]. Note that the sojourn time variance depends linearly on x for $x \rightarrow \infty$:

$$var(\mathbf{S}^{PS}(x)) \sim \frac{2\rho\hat{\beta}}{(1-\rho)^3} x - \frac{2\rho\hat{\beta}^2}{(1-\rho)^4}, \quad x \to \infty,$$
 (3.24)

(see also Kleinrock [1976], p. 170), whereas it depends quadratically on x for $x \rightarrow 0$:

$$var(\mathbf{S}^{PS}(x)) \sim \frac{\rho}{(1-\rho)^2} x^2 - \frac{\lambda}{3(1-\rho)} x^3, \quad x \to 0.$$
 (3.25)

The M/G/1 PS queue

.

We now derive an expression for $var(S^{PS}(x))$ for the M/G/1 PS queue, in particular showing that the above asymptotic properties hold for general service time distributions. We consider service time distributions with LST as in (3.12), by choosing Q(z) as in (3.14), (3.15):

$$Q(z) = \sum_{j=1}^{m} \alpha_j \prod_{i=1}^{r_j} \frac{(1-p_{ij})z}{1-p_{ij}z} = \sum_{j=1}^{m} \alpha_j \prod_{i=1}^{r_j} \frac{\beta z / \hat{\beta}_{ij}}{1-(1-\beta/\hat{\beta}_{ij})z}.$$
 (3.26)

Analogously to the M/M/1 case analyzed above, (3.9) and (3.26) lead to:

$$C_n = U_1 + U_2 x_2^{n-1} + \cdots + U_L x_L^{n-1}, \ n = 1, 2, ...,$$
 (3.27)

where $1/x_2, \ldots, 1/x_L$ are the roots of

$$1 - \lambda \beta \frac{z}{1 - z} (1 - Q(z)) = 0.$$
(3.28)

 U_1, \ldots, U_L are determined by

$$\frac{U_1 z}{1 - x_1 z} + \dots + \frac{U_L z}{1 - x_L z} = C(z).$$
(3.29)

Note that in (3.27)-(3.29) we have used the following assumption:

Assumption 3.1

It is assumed that the roots $1/x_2, \ldots, 1/x_L$ of (3.28) are all distinct.

Remark 3.1

Assumption 3.1 can be easily proved to hold for the case of Erlang and hyperexponential service times; we have found no example for which the roots are *not* distinct.

Remark 3.2

 $1/x_2, \ldots, 1/x_L$ are the roots of a polynomial of degree $L-1 \leq \sum_{j=1}^m r_j$, see (3.26), (3.28); for example, for *m*-stage hyperexponential and *m*-stage Erlang service time distributions L = m + 1. Note that (3.29) leads to a set of L linear equations from which U_1, \ldots, U_L can be obtained.

We now prove some properties of x_i and U_i that will be used in the sequel.

LEMMA 3.1 (i) $|x_i| < 1$, i = 2,...,L; (ii) x_i can be written as

 $x_i = 1 - \beta a_i, \tag{3.30}$

with a_i independent of β , and Re $a_i > 0$, i = 2,...,L; (iii) U_i is independent of β , i = 1,...,L, and $U_1 = 1/(1-\rho)$.

PROOF

Noting that (see (3.9)),

$$1-\lambda\beta\frac{z}{1-z}(1-Q(z)) = 1-\lambda\beta\sum_{i=1}^{\infty}q(i)z^{i},$$

and $\lambda \beta \sum_{i=1}^{\infty} q(i) = \rho < 1$, it follows immediately that $|x_i| < 1$, i = 2, ..., L. To prove (ii), substitute (3.26) into (3.28) and replace z by $1/(1-\beta \tilde{z})$. Then (3.28) reduces to

$$1 + \frac{\lambda}{\tilde{z}} - \frac{\lambda}{\tilde{z}_{j}} \sum_{j=1}^{m} \alpha_{j} \prod_{i=1}^{r_{j}} \frac{1}{1 - \hat{\beta}_{ij} \tilde{z}} = 0.$$
(3.31)

Since $1/x_i$ is a root of (3.28), $(1-x_i)/\beta = a_i$ is a root of (3.31). The fact that

 β does not occur in the left-hand side of (3.31) implies that $1-x_i$ depends linearly on β . The statement concerning Re $a_i > 0$ now follows from (i). It follows from (3.29) that

$$U_{i}\frac{1}{x_{i}} = \lim_{z \to 1/x_{i}} (1 - zx_{i})C(z) = \lim_{\tilde{z} \to a_{i}} (1 - \frac{x_{i}}{1 - \beta \tilde{z}})C(\frac{1}{1 - \beta \tilde{z}}).$$
(3.32)

Observing that $\beta C(\frac{1}{1-\beta \tilde{z}})$ is independent of β , it is found that

$$U_i = \lim_{\tilde{z} \to a_i} x_i (1 - \frac{x_i}{1 - \beta \tilde{z}}) C(\frac{1}{1 - \beta \tilde{z}})$$

is independent of β .

Finally, $U_1 = 1/(1-\rho)$ follows from (3.10), (3.27) and (i).

Substituting (3.27) into (3.7) yields (cf. (3.22))

$$var(\mathbf{S}^{(k)}) = (\frac{\beta}{1-\rho})^2 [k - 2(1-\rho) \sum_{j=2}^{L} U_j \frac{x_j^k + k(1-x_j) - 1}{(1-x_j)^2}].$$
 (3.33)

Now, let x be the service time of a tagged customer, and take $\beta = x/k$. For $k \to \infty$, $var(\mathbf{S}^{PS}(x))$ follows from (3.33) and (i) of Lemma 3.1; integrating $E\{(\mathbf{S}^{PS}(x))^2\} = var(\mathbf{S}^{PS}(x)) + x^2/(1-\rho)^2$ over x and subtracting $(E\{\mathbf{S}^{PS}\})^2 = \beta^2/(1-\rho)^2$ yields the unconditional sojourn time variance. We collect these results in

THEOREM 3.1

In the M/G/1 PS queue with service time LST given by (3.12),

$$var(\mathbf{S}^{PS}(x)) = \frac{2}{1-\rho_j} \sum_{j=2}^{L} (1/a_j)^2 U_j [1-xa_j - e^{-xa_j}], \qquad (3.34)$$

$$var(\mathbf{S}^{PS}) = \frac{2}{1-\rho} \sum_{j=2}^{L} (1/a_j)^2 U_j [1-\hat{\beta}a_j - E\{e^{-a_j \tau^{PS}}\}] + \frac{E\{(\tau^{PS})^2\} - \hat{\beta}^2}{(1-\rho)^2}, \quad (3.35)$$

with a_2, \ldots, a_L the roots of (3.31) and U_2, \ldots, U_L determined by (3.29), cf. Remark 3.2; a_j and U_j are independent of x, j = 2, ..., L.

Formula (3.34) shows that $var(\mathbf{S}^{PS}(x))$ depends on the required service time x in a very simple way. It is convenient to use this formula for the analysis of

the behaviour of the sojourn time variance when x varies. Below we shall derive asymptotic results for $x \rightarrow \infty$ and $x \rightarrow 0$.

From (ii) of Lemma 3.1 it follows that

$$\lim_{x \to \infty} \frac{2}{1 - \rho_j} \sum_{j=2}^{L} (1/a_j)^2 U_j e^{-xa_j} = 0$$

Hence, the sojourn time variance is asymptotically linear in x:

$$var(\mathbf{S}^{PS}(x)) \sim \frac{2}{1-\rho_j} \sum_{j=2}^{L} (1/a_j)^2 U_j(1-xa_j), \quad x \to \infty.$$
 (3.36)

From (3.27) and (3.30),

$$\sum_{j=2}^{L} (1/a_j) U_j = \beta \sum_{j=2}^{L} U_j \frac{1}{1-x_j} = \beta \sum_{n=1}^{\infty} (C_n - \frac{1}{1-\rho}),$$

and

$$\sum_{j=2}^{L} (1/a_j)^2 U_j = \beta^2 \sum_{j=2}^{L} U_j \frac{1}{(1-x_j)^2} = \beta^2 \sum_{n=1}^{\infty} n(C_n - \frac{1}{1-\rho}) .$$

It can be derived from (3.9) that

$$\beta \sum_{n=1}^{\infty} (C_n - \frac{1}{1-\rho}) = -\frac{\lambda \hat{\beta}_2}{2(1-\rho)^2} ,$$

and

$$\beta_{n=1}^{2} \sum_{n=1}^{\infty} n(C_{n} - \frac{1}{1-\rho}) = -\frac{1}{2(1-\rho)^{3}} \left[\frac{\lambda}{3}\hat{\beta}_{3} + \lambda^{2} \left(\frac{1}{2}\hat{\beta}_{2}^{2} - \frac{1}{3}\hat{\beta}\hat{\beta}_{3}\right)\right],$$

with $\hat{\beta}_i := E\{(\tau^{PS})^i\}$, i = 2,3. Hence, from (3.36),

$$var(\mathbf{S}^{PS}(x)) \sim \frac{\lambda \hat{\beta}_2}{(1-\rho)^3} x - \frac{1}{(1-\rho)^4} [\frac{\lambda}{3} \hat{\beta}_3 + \lambda^2 (\frac{1}{2} \hat{\beta}_2^2 - \frac{1}{3} \hat{\beta} \hat{\beta}_3)], \quad (3.37)$$

(Note that, formally, this asymptotic result should have been written as $var(\mathbf{S}^{PS}(x)) - (\lambda \hat{\beta}_2 / (1-\rho)^3)x \sim -[\frac{\lambda}{3}\hat{\beta}_3 + \lambda^2(\frac{1}{2}\hat{\beta}_2^2 - \frac{1}{3}\hat{\beta}\hat{\beta}_3)]/(1-\rho)^4, x \to \infty$).

Noting that, in (3.34),

$$1 - xa_j - e^{-xa_j} = -\sum_{i=2}^{\infty} \frac{(-xa_j)^i}{i!}, \quad j = 2, ..., L,$$

and using

$$\sum_{j=2}^{L} U_j = C_1 - \frac{1}{1-\rho} = -\frac{\rho}{1-\rho}, \qquad \sum_{j=2}^{L} U_j x_j = C_2 - \frac{1}{1-\rho} = \lambda \beta - \frac{\rho}{1-\rho},$$

it is found that (cf. the remark below (3.37))

$$var(\mathbf{S}^{PS}(x)) \sim \frac{\rho}{(1-\rho)^2} x^2 - \frac{\lambda}{3(1-\rho)} x^3, \quad x \to 0.$$
 (3.38)

This expression appears to be independent of the service time distribution, apart from its first moment (cf. also (3.25)). The quadratic behaviour of $var(\mathbf{S}^{PS}(x))$ for small service requests x should be contrasted with the linear behaviour for large x.

Remark 3.3

Formula (3.38) slightly generalizes Theorem 1 of Yashkov [1986]. Formula (3.37) is contained in Theorem 2 of the same paper; but for the service time distributions defined by (3.12), Yashkov's theorem follows immediately from (3.34).

3.3.4 The distribution of the sojourn time

Application of the limiting procedure to (3.4) yields the LST of the distribution of the sojourn time in the M/G/1 PS queue. The analysis can be performed along the same lines as the analysis of the sojourn time variance. It appears that the M_n 's in (3.4) have similar properties as the C_n 's in the previous subsection. However, there are some difficulties which did not arise in the analysis of the variance. These problems are due to the presence of the individual feedback probabilities contained in the q(n)'s in the denominator of (3.4). In general the q(n)'s are given by very complicated expressions and can not be explicitly determined for the whole class of service time distributions given by (3.12) (cf. (3.14), (3.15) and (3.16)). Therefore, we shall restrict ourself below to a subclass of these service times, viz. mixtures of Erlang distributions: (cf. (3.12))

$$E\{e^{-\omega_0\tau^{FS}}\} = \sum_{j=1}^{m} \alpha_j \left(\frac{1}{1+\hat{\beta}_j\omega_0}\right)^{r_j},$$
(3.39)

with $\alpha_1, \ldots, \alpha_m \ge 0$, $\sum_{j=1}^m \alpha_j = 1$, r_1, \ldots, r_m positive integers. The corresponding feedback probabilities are determined by (cf. (3.14), (3.15))

$$Q(z) = \sum_{j=1}^{m} \alpha_j \left[\frac{(1-p_j)z}{1-p_j z} \right]^{r_j}, \qquad (3.40)$$

with $p_j = 1 - \beta / \hat{\beta}_j > 0$.

From (3.40) we find

$$q(l)(1-p(l)) = \sum_{j=1}^{m} \alpha_j (1-p_j)^{r_j} {l-1 \brack r_j - 1} p_j^{l-r_j}, \qquad (3.41)$$

from which the q(n)'s can be obtained via:

$$q(n) = \sum_{l=n}^{\infty} q(l)(1-p(l)), \quad n = 1, 2, \dots$$
(3.42)

Note that the (sub)class of distribution functions determined by (3.39) is still large enough to approximate the distribution of any nonnegative random variable arbitrarily closely (cf. Tijms [1986], p. 398).

We shall start the analysis with a lemma that states some properties of the M_n 's given by (3.5) (see also (3.11)). Then, as an example, we consider the M/M/1 PS queue and show how these properties can be exploited to derive from (3.4) the LST of the sojourn time distribution. Next, the general case is treated. Finally, we consider the M/D/1 PS queue. Although the deterministic distribution is not contained in the class of service time distributions determined by (3.39) it appears that this case can be (partially) analyzed and yields simple expressions.

Let $1/y_1, \ldots, 1/y_L$ be the zeros of the denominator,

$$1 - z(1 + \beta \omega_0 + \lambda \beta) + \lambda \beta z Q(z), \qquad (3.43)$$

of the generating function M(z) of the M_n 's, cf. (3.11). To obtain closed expressions for the M_n 's we introduce the following assumption: (cf. Assumption 3.1)

Assumption 3.2

We assume that the zeros $1/y_1, \ldots, 1/y_L$ of (3.43) are all distinct.

Under this assumption it is easily seen that we can write, cf. (3.27),

$$M_n = A_1 y_1^n + \cdots + A_L y_L^n$$
, $n = 1, 2, ...,$ (3.44)

with A_1, \ldots, A_L determined by

$$\frac{A_{1}y_{1}}{1-y_{1}z} + \cdots + \frac{A_{L}y_{L}}{1-y_{L}z} = M(z).$$
(3.45)

Remark 3.4

· . · ·

 $1/y_1, \ldots, 1/y_L$ are the roots of a polynomial of degree $L \le 1 + \sum_{j=1}^m r_j$, see (3.11), (3.40); for example, for *m*-stage hyperexponential and *m*-stage Erlang service time distributions L = m + 1. Note that (3.45) leads to a set of L linear equations from which A_1, \ldots, A_L can be obtained (cf. Remark 3.2).

Analogously to the proof of (ii) and (iii) of Lemma 3.1 it can be shown that

Lemma 3.2

(i) y_i can be written as

$$y_i = 1 - \beta d_i, \tag{3.46}$$

with d_i independent of β , i = 1, ..., L;

(ii) A_i is independent of β , i = 1, ..., L.

Note that, in fact, $d_i = (1-y_i)/\beta$, i = 1, ..., L are the roots of: (cf. (3.43) and the derivation of (3.31))

$$\tilde{z} + \omega_0 + \lambda - \lambda \sum_{j=1}^m \alpha_j \left(\frac{1}{1 - \hat{\beta}_j \tilde{z}} \right)^{r_j} = 0.$$
(3.47)

The properties stated in Lemma 3.2 will be used below. Before treating the general case we first give an example.

The M/M/1 PS queue

For exponential service times (Q(z)=(1-p)z/(1-pz)), with $p=1-\beta/\hat{\beta}$

$$M(z) = z \frac{1 + \beta \omega_0 - \lambda \beta \frac{z}{1 - z} (1 - (1 - p)z/(1 - pz))}{1 - z (1 + \beta \omega_0 + \lambda \beta) + \lambda \beta z (1 - p)z/(1 - pz)} .$$
(3.48)

It is easily seen that the zeros $1/y_1$ and $1/y_2$ of the denominator of (3.48) are given by

$$y_{1} = \frac{1}{2} \left[1 + \beta \omega_{0} + \lambda \beta + p + \sqrt{(1 + \beta \omega_{0} + \lambda \beta + p)^{2} - 4(p + p \beta \omega_{0} + \lambda \beta)} \right]$$
$$= 1 + \frac{1}{2} \beta \left[\omega_{0} + \lambda - 1/\hat{\beta} + \sqrt{(\omega_{0} + \lambda - 1/\hat{\beta})^{2} + 4\omega_{0}/\hat{\beta}} \right],$$

$$y_{2} = \frac{1}{2} \left[1 + \beta \omega_{0} + \lambda \beta + p - \sqrt{(1 + \beta \omega_{0} + \lambda \beta + p)^{2} - 4(p + p \beta \omega_{0} + \lambda \beta)} \right]$$
$$= 1 + \frac{1}{2} \beta \left[\omega_{0} + \lambda - 1/\hat{\beta} - \sqrt{(\omega_{0} + \lambda - 1/\hat{\beta})^{2} + 4\omega_{0}/\hat{\beta}} \right].$$

We can write (cf. (3.44))

$$M_n = A_1 y_1^n + A_2 y_2^n , \quad n = 1, 2, \dots .$$
(3.49)

For the determination of A_1 and A_2 it is more convenient to use (3.5) instead of (3.45):

$$A_1 + A_2 = M_0 = 1$$
,
 $A_1 y_1 + A_2 y_2 = M_1 = 1 + \beta \omega_0$.

Hence

$$A_1 = \frac{y_2 - (1 + \beta \omega_0)}{y_2 - y_1}$$
, $A_2 = \frac{y_1 - (1 + \beta \omega_0)}{y_1 - y_2}$.

Now substitute (3.49) into (3.4) and evaluate the summations in the denominator (take $q(i)=p^{i-1}$). Taking in the resulting expressions $y_i=1-\beta d_i$, i=1,2, $p=1-\beta/\beta$, $\beta=x/k$ and using that d_i is independent of β it is easily seen that

$$\lim_{k \to \infty} (1+\beta\omega_0)M_{k-1} = \sum_{h=1}^2 A_h e^{-xd_h} ,$$

$$\lim_{k \to \infty} \lambda \beta \sum_{j=1}^{k-2} q(k-j-1)M_j = \lambda \sum_{h=1}^2 A_h \hat{\beta} \frac{1}{1-\hat{\beta}d_h} [e^{-xd_h} - e^{-x/\hat{\beta}}] ,$$

$$\lim_{k \to \infty} \lambda \beta \sum_{i=1}^{k-2} q(i) = \lambda \hat{\beta} (1-e^{-x/\hat{\beta}}) .$$

Hence, cf. (3.4),

$$E\{e^{-\omega_0 \mathbf{S}^{t^s}(x)}\} = \lim_{k \to \infty} E\{e^{-\omega_0 \mathbf{S}^{(k)}}\} =$$
(3.50)

$$\frac{1-\rho}{\sum_{h=1}^{2}A_{h}e^{-xd_{h}}-\rho e^{-x/\hat{\beta}}-\lambda\sum_{h=1}^{2}A_{h}\hat{\beta}\frac{1}{1-\hat{\beta}d_{h}}[e^{-xd_{h}}-e^{-x/\hat{\beta}}]}.$$

It is easily shown that this result coincides with the result obtained in Coffman

et al. [1970] (formula (30) on page 128). Note that formula (30) of that paper represents the LST of the distribution of the total *delay* of a customer with a specific service demand. To match it with our result it has to be multiplied by the LST of the required service time (given by $e^{-\omega_0 x}$).

The M/G/1 PS queue

Now we shall treat the general case, i.e. the case that the service times are determined by (3.39). Consider in the corresponding feedback queue the total sojourn time after k services given by (3.4). As in the M/M/1 case, we evaluate the terms $(1+\beta\omega_0)M_{k-1}$, $\lambda\beta\sum_{j=1}^{k-2}q(k-j-1)M_j$ and $\lambda\beta\sum_{i=1}^{k-2}q(i)$ in the denominator and take the limit $k\to\infty$ independently for each term. The first term is simple: from (3.44) and (3.46) it is easily seen that

$$\lim_{k \to \infty} (1 + \beta \omega_0) M_{k-1} = \lim_{k \to \infty} (1 + \frac{x}{k} \omega_0) \sum_{h=1}^{L} A_h (1 - \frac{x}{k} d_h)^{k-1} = \sum_{h=1}^{L} A_h e^{-x d_h} . \quad (3.51)$$

The second one needs more effort. Using (3.41)-(3.44) and (3.46) it is found after extensive calculations that

$$\begin{split} \lambda \beta \sum_{j=1}^{k-2} q(k-j-1)M_{j} &= \\ \lambda \beta \sum_{j=1}^{k-2} M_{j} \sum_{n=1}^{m} \alpha_{n} \sum_{i=0}^{r_{n}-1} \left[\binom{k-j-2}{r_{n}-1-i} (1-p_{n})^{r_{n}-1-i} p_{n}^{k-j-2-(r_{n}-1-i)} \right] \\ \lambda \beta \sum_{h=1}^{L} A_{h} \sum_{n=1}^{m} \alpha_{n} \sum_{i=0}^{r_{n}-1} \sum_{j=1}^{k-2} \left[\binom{k-j-2}{r_{n}-1-i} (1-p_{n})^{r_{n}-1-i} p_{n}^{k-j-2-(r_{n}-1-i)} y_{h}^{i} \right] \\ \lambda \sum_{h=1}^{L} A_{h} \sum_{n=1}^{m} \alpha_{n} \sum_{i=0}^{r_{n}-1} \frac{(1-x/(k\hat{\beta}_{n}))^{k-2-(r_{n}-1-i)}}{(r_{n}-1-i)!} \times \\ \\ \lambda \sum_{j=1}^{L} A_{h} \sum_{n=1}^{m} \alpha_{n} \sum_{i=0}^{r_{n}-1} \frac{(1-x/(k\hat{\beta}_{n}))^{k-2-(r_{n}-1-i)}}{(r_{n}-1-i)!} \times \\ \end{split}$$

The last equality is obtained by substituting $p_n = 1 - \beta / \hat{\beta}_n$, $\beta = x/k$ and noting that $\begin{pmatrix} k-j-2\\ r_n-1-i \end{pmatrix} = 0$ if $k-j-2 < r_n-1-i$.

Using that, actually by definition,

$$\lim_{k \to \infty} \sum_{j=1}^{k-r_n-1+i} \frac{x}{k} (k-j-2)(k-j-3) \cdots (k-j-2-(r_n-2-i)) \times$$
(3.52)

58

$$(x/(k\hat{\beta}_n))^{r_n-1-i}\left(\frac{1-xd_h/k}{1-x/(k\hat{\beta}_n)}\right)^j =$$

$$\int_{s=0}^x (\frac{x}{\hat{\beta}_n} - \frac{s}{\hat{\beta}_n})^{r_n-1-i} e^{-s(d_h-1/\hat{\beta}_n)} ds ,$$

we obtain

$$\lim_{k \to \infty} \lambda \beta \sum_{j=1}^{k-2} q(k-j-1)M_j =$$

$$\lambda \sum_{h=1}^{L} A_h \sum_{n=1}^{m} \alpha_n \sum_{i=0}^{r_n-1} \frac{e^{-x/\hat{\beta}_n}}{(r_n-1-i)!} \int_{s=0}^{x} (\frac{x}{\hat{\beta}_n} - \frac{s}{\hat{\beta}_n})^{r_n-1-i} e^{-s(d_n-1/\hat{\beta}_n)} ds .$$
(3.53)

The evaluation of the third term is analogous to that of the second term:

$$\lambda \beta \sum_{i=1}^{k-2} q(i) = \lambda \beta \sum_{i=1}^{k-2} \sum_{n=1}^{m} \alpha_n \sum_{j=0}^{r_n-1} \left[\binom{i-1}{r_n-1-j} (1-p_n)^{r_n-1-j} p_n^{i-1-(r_n-1-j)} \right] = \lambda \sum_{n=1}^{m} \alpha_n \sum_{j=0}^{r_n-1} \frac{(1-x/(k\hat{\beta}_n))^{-(r_n-j)}}{(r_n-1-j)!} \times \sum_{i=r_n-j}^{k-2} \frac{x}{k} (i-1) \cdots (i-1-(r_n-2-j)) (x/(k\hat{\beta}_n))^{r_n-1-j} (1-x/(k\hat{\beta}_n))^i.$$

Hence, cf. (3.52),

$$\lim_{k \to \infty} \lambda \beta \sum_{i=1}^{k-2} q(i) = \lambda \sum_{n=1}^{m} \alpha_n \sum_{j=0}^{r_n-1} \frac{1}{(r_n-1-j)!} \int_{s=0}^{s} (\frac{s}{\beta_n})^{r_n-1-j} e^{-s/\hat{\beta}_n} ds .$$
(3.54)

(In the derivation of (3.54) one recognizes the convergence of the binomial probability $\binom{i-1}{r_n-1-j}(1-p_n)^{r_n-1-j}p_n^{i-1-(r_n-1-j)}$ to the Poisson probability $\frac{1}{(r_n-1-j)!}(s/\hat{\beta}_n)^{r_n-1-j}e^{-s/\hat{\beta}_n}$, cf. Feller [1950, Ch. 6]; a similar phenomenon occurs in the derivation of (3.53)).

The integrals in (3.53) and (3.54) can be evaluated by noting that

$$\int_{s=0}^{x} s^{n} e^{-s/c} ds = n! c^{n+1} (1-e^{-x/c}) - e^{-x/c} \sum_{j=0}^{n-1} \frac{n!}{(n-j)!} x^{n-j} c^{j+1}.$$

Using the resulting expressions and (3.51) we obtain from (3.4):

.

THEOREM 3.2 In the M/G/1 PS queue with service time LST given by (3.39), for Re $\omega_0 \ge 0$,

$$E\{e^{-\omega_{0}S^{rs}(x)}\} = \lim_{k \to \infty} E\{e^{-\omega_{0}S^{ts}}\} =$$

$$(3.55)$$

$$(1-\rho) \left[\sum_{h=1}^{L} A_{h}e^{-xd_{h}} - \lambda \sum_{n=1}^{m} \alpha_{n}\hat{\beta}_{n}e^{-x/\hat{\beta}_{n}} \sum_{j=0}^{r_{n}-1} \sum_{i=0}^{j} \frac{(x/\hat{\beta}_{n})^{i}}{i!} - \lambda \sum_{h=1}^{L} A_{h} \sum_{n=1}^{m} \alpha_{n}\hat{\beta}_{n} (\frac{1}{1-\hat{\beta}_{n}d_{h}})^{r_{n}} \sum_{j=0}^{r_{n}-1} (1-\hat{\beta}_{n}d_{h})^{j} (e^{-xd_{h}} - e^{-x/\hat{\beta}_{n}} \sum_{i=0}^{r_{n}-1-j} \frac{(x(1-\hat{\beta}_{n}d_{h})/\hat{\beta}_{n})^{i}}{i!})\right]^{-1}$$

with d_1, \ldots, d_L the roots of (3.47) and A_1, \ldots, A_L determined by (3.45), cf. Remark 3.4; d_h and A_h are independent of x, h = 1, ..., L.

For hyperexponentially (H_m) distributed service times $(r_j = 1, j = 1, ..., m, cf.$ (3.39)) (3.55) reduces to

$$E\{e^{-\omega_{0}\mathbf{S}^{PS}(x)}\} = (1-\rho) \left[\sum_{h=1}^{L} A_{h}e^{-xd_{h}} - \lambda \sum_{n=1}^{m} \alpha_{n}\hat{\beta}_{n}e^{-x/\hat{\beta}_{n}} - (3.56)\right]^{-1} \\ \lambda \sum_{h=1}^{L} A_{h}e^{-xd_{h}} \sum_{n=1}^{m} \alpha_{n}\hat{\beta}_{n} \frac{1}{1-\hat{\beta}_{n}d_{h}} (1-e^{-x(1-\hat{\beta}_{n}d_{h})/\hat{\beta}_{n}})\right]^{-1}.$$

It is easily verified that for m = 1 (the M/M/1 case) (3.56) coincides with (3.50).

Remark 3.5

Our formulas for the variance ((3.34), (3.35)) and the LST ((3.55)) of the sojourn time are given in terms of the roots of a polynomial and the solution of a set of linear equations. The corresponding formulas presented in Yashkov [1987] are given in terms of multiple integrals. In general both types of formulas can only be evaluated numerically. For obtaining numerical results it seems in our case to be more convenient to use the feedback results (2.22), (2.26) and (3.4), (3.5) and to evaluate a finite number of steps of the limiting procedure described in Subsection 3.3.1.

As we remarked at the beginning of this subsection the analysis above does

,

not apply to the M/D/1 PS queue. In fact, the deterministic service times can be approximated by an Erlang-*n* distribution (for large *n*) but this leads to the problem of finding the roots of an (n + 1)-th degree polynomial and the solution of a set of n + 1 linear equations (cf. (3.44), (3.45)). Below we shall show how explicit formulas for the sojourn time in the M/D/1 PS queue can be easily obtained from the sojourn time in the M/M/1 queue with deterministic feedback analyzed in Chapter 2.

The M/D/1 PS queue

Consider the M/M/1 queue with deterministic feedback in which each customer receives exactly N services, see Subsection 2.5.2. Taking $N = \lceil \hat{\beta} / \beta \rceil$ and $\beta = x/k$ it is clear that the total sojourn time after k services in the feedback queue approaches, for $k \to \infty$, the sojourn time of a (special) customer with service demand x in the M/D/1 PS queue with service time $\hat{\beta}$. Application of this limiting procedure to (2.45) yields immediately the LST of the distribution of $\mathbf{S}^{PS}(x)$:

$$E\{e^{-\omega_0 \mathbf{S}^{PS}(x)}\} = \frac{(1-\lambda\hat{\beta})(\lambda+\omega_0)^2}{\omega_0^2 e^{x(\lambda+\omega_0)} + \lambda(\lambda+\omega_0)[1-\lambda\hat{\beta}+\omega_0(x-\hat{\beta})] + \lambda\omega_0},$$
(3.57)

Re $\omega_0 \ge 0$, $x \le \hat{\beta}$.

This result has been obtained before by Ott [1984].

Remark 3.6

(3.57) holds only for $0 \le x \le \hat{\beta}$. To obtain a similar formula for $x > \hat{\beta}$ we need an explicit expression for $E\{e^{-\omega_0 \mathbf{S}^{(k)}}\}$ for k > N, cf. (2.45); but as noted in Remark 2.4 this seems to be impossible for the case of deterministic feedback. Ott's method also precludes the derivation of an explicit formula for $x > \hat{\beta}$ (cf. Remark 5.2 in Ott [1984]).

We conclude this section with a remark on the state of the PS system just after the departure of a tagged customer.

Remark 3.7

From Corollary 2.2 and application of the limiting procedure, see Subsection 3.3.1, it follows that for the M/G/1 PS queue the state of the system (the number of customers present *and* their residual service requests) just after the departure of a tagged customer who has received an amount $x \ge 0$ of service is described by the stationary distribution of the state of the system at an arbitrary epoch, *independent* of x. This result slightly extends Theorem 2.3 of Ott [1984]. Ott's theorem concerns only the distribution of the number of customers at a departure epoch of a tagged customer with initial service demand x.

3.4 The M/G/1 queue with generalized processor sharing 3.4.1 Introduction

In the previous sections it has been shown that the M/G/1 PS queue can be considered as a limiting case of an M/M/1 queue with feedback. We exploited the well known product form property of the joint queue length distribution to derive some sojourn time characteristics in the feedback queue and we used these results to obtain corresponding performance measures in the PS queue. In this section we shall show how a completely similar method can be applied to analyze an interesting generalization of the PS service discipline denoted as 'generalized processor sharing' (GPS). The GPS service discipline, investigated by Cohen [1979], generalizes the PS discipline as follows: when there are *j* customers present in the system then the service rate for each of these customers is f(j) > 0, i.e. during a small time Δt the attained service of each customer increases with $f(j)\Delta t$, j = 1, 2, ... Note that, when j customers are present, the capacity of the server (the total service rate) is equal to if(i), $j = 1, 2, \dots$. If f(j) = 1/j then the GPS model clearly reduces to the PS model. It will be shown below that results for the M/G/1 GPS queue can be easily obtained from the analysis of the M/M/1 feedback queue with state dependent service rates. Indeed, it is intuitively clear that when we choose the service rate in the feedback queue equal to if(j) when there are j customers present and let the mean service times $\beta \rightarrow 0$ ($p(i) \rightarrow 1$) as described in Subsection 3.3.1 then the behaviour of the feedback queue approaches that of the GPS queue. The formal proof of this convergence is completely analogous to that of the PS case, see Section 3.1. To illustrate this new approach to the GPS service discipline we shall show how the following two basic GPS results can be easily obtained from corresponding results for the feedback queue with state dependent service rates. Let

$$\phi(n) := \left(\prod_{j=1}^{n} f(j)\right)^{-1}, \quad n = 1, 2, ...,$$

$$:= 1, \quad n = 0.$$
(3.58)

For the M/G/1 GPS queue, cf. Cohen [1979],

(i) the distribution of the number of customers \mathbf{X}^{GPS} present in the system is given by

$$Pr\{\mathbf{X}^{GPS}=n\} = \frac{\frac{\rho^n}{n!}\phi(n)}{\sum_{j=0}^{\infty}\frac{\rho^j}{j!}\phi(j)},$$
(3.59)

(ii) the mean conditional sojourn time, $E\{\mathbf{S}^{GPS}(x)\}$ of a customer with service demand x is given by

$$E\{\mathbf{S}^{GPS}(x)\} = x \frac{\sum_{n=0}^{\infty} \frac{\rho^n}{n!} \phi(n+1)}{\sum_{j=0}^{\infty} \frac{\rho^j}{j!} \phi(j)},$$
(3.60)

where ρ denotes, as usual, the offered load to the system per unit of time.

The formulas (3.59) and (3.60) have been derived under the stability condition $\sum_{j=0}^{\infty} \frac{\rho^j}{j!} \phi(j) < \infty$. Note that both the queue length distribution and the mean conditional sojourn time are independent of the service time distribution apart from its first moment; $E\{\mathbf{S}^{GPS}(x)\}$ is linear in x. Apparently, the GPS service discipline generalizes similar properties of the M/G/1 PS queue, cf. (3.19).

In fact, Cohen [1979] studies the GPS discipline in a very general model of open and closed networks with different job classes. This general model contains e.g. the classical Erlang and Engset systems. For the analysis Cohen uses the technique of the supplementary variable. He obtains generalizations of known results for closed and open networks such as the product form and the insensitivity property of the probabilities of the network states, cf. Baskett et al. [1975]; sojourn time results are restricted to means. We have found that most of these results can also be obtained from the M/M/1 feedback queue with state dependent service rates or from networks of these feedback queues. In fact the GPS network can be considered as a limiting model of a network consisting of M/M/1 feedback queues with state dependent service rates and suitably chosen routing probabilities. Each feedback queue corresponds to a node in the GPS network. This network of feedback queues is contained in the well known class of product form networks analyzed by Baskett et al. [1975]. Application of the limiting procedure to their results yields Cohen's GPS results. Here we shall restrict ourself to the derivation of (3.59) and (3.60).

3.4.2 Analysis

Consider the M/M/1 feedback model described in Section 2.2 but with one difference: when there are j customers present in the system then the server works with a rate $\mu(j)$, j = 1, 2, Note that the amount of service that a customer receives during each pass is still exponentially distributed with mean β ; only the speed with which he is served may change during time. The 'old' case is obtained by taking $\mu(j) \equiv 1$. As in Section 2.2 we start the analysis assuming that the feedback probabilities remain constant after a finite number of services, i.e. $p(i) \equiv p$, $i \ge N$ for some $N \ge 1$. (The notation introduced before is extended in an obvious way). Thus, the total number of different customer types is limited to N. It will appear later on that the results are independent of N; so, this assumption is no restriction. We define

$$\rho_{i} := \lambda \beta \prod_{j=1}^{i-1} p(j), \quad i = 1, ..., N-1,$$
$$\rho_{N} := \frac{\lambda \beta}{1-p} \prod_{j=1}^{N-1} p(j).$$

.

Thus, ρ_i is the offered load to the system per unit of time due to type-*i* customers. Obviously, the *total* offered load to the system per unit of time, ρ , is equal to $\sum_{i=1}^{N} \rho_i$.

The (stationary) joint queue length distribution is found from the general network results obtained by Baskett et al. [1975]:

$$Pr\{\mathbf{X}_{1}=x_{1},\ldots,\mathbf{X}_{N}=x_{N}\} =$$

$$C \frac{(x_{1}+\cdots+x_{N})!}{x_{1}!\cdots x_{N}!}\prod_{i=1}^{N}\rho_{i}^{x_{i}}\prod_{j=1}^{x_{1}+\ldots+x_{N}}(\mu(j))^{-1},$$
(3.61)

with

$$C = \frac{1}{\sum_{m=0}^{\infty} \rho^m \prod_{j=1}^{m} (\mu(j))^{-1}},$$
(3.62)

an empty product being one by definition. (Note, for the derivation of (3.61) and (3.62), that in Baskett et al. [1975] the service rate is defined as the mean number of customers that can be served per unit of time; multiplication by β yields our definition of the service rate.)

It follows from (3.61) and (3.62) that the distribution of the *total* number of customers in the system is given by

$$Pr\{\mathbf{X}=n\} = Pr\{\mathbf{X}_{1}+\dots+\mathbf{X}_{N}=n\} = \frac{\rho^{n}\prod_{j=1}^{n}(\mu(j))^{-1}}{\sum_{m=0}^{\infty}\rho^{m}\prod_{j=1}^{m}(\mu(j))^{-1}}, \quad (3.63)$$

The mean number of type-h customers in the system can also be obtained from (3.61) and (3.62):

$$E\{\mathbf{X}_{h}\} =$$

$$C\sum_{x_{1}=0}^{\infty} \cdots \sum_{x_{N}=0}^{\infty} x_{h} \frac{(x_{1} + \cdots + x_{N})!}{x_{1}! \cdots x_{N}!} \prod_{i=1}^{N} \rho_{i}^{x_{i}} \prod_{j=1}^{x_{1} + \cdots + x_{N}} (\mu(j))^{-1} =$$
(3.64)

$$\rho_h C \sum_{x_1=0}^{\infty} \cdots \sum_{x_N=0}^{\infty} \frac{(x_1 + \cdots + x_N + 1)!}{x_1! \cdots x_N!} \prod_{i=1}^N \rho_i^{x_i} \prod_{j=1}^{x_1 + \cdots + x_N + 1} (\mu(j))^{-1} = \rho_h C \sum_{m=0}^{\infty} (m+1) \rho^m \prod_{j=1}^{m+1} (\mu(j))^{-1} , \quad h = 1, \dots, N.$$

Using Little's formula it follows immediately that the mean duration of the *h*-th sojourn time, $E{S_h}$, of a customer is given by

$$E\{\mathbf{S}_h\} = \beta C \sum_{m=0}^{\infty} (m+1) \rho^m \prod_{j=1}^{m+1} (\mu(j))^{-1}, \quad h = 1, ..., N.$$
(3.65)

Note that both (3.63) and (3.65) are independent of N. In addition $E{S_h}$ is independent of h. Hence, the mean total sojourn time after k services is linear in k:

$$E\{\mathbf{S}^{(k)}\} = k \frac{\beta \sum_{m=0}^{\infty} (m+1)\rho^m \prod_{j=1}^{m+1} (\mu(j))^{-1}}{\sum_{m=0}^{\infty} \rho^m \prod_{j=1}^{m} (\mu(j))^{-1}}, \quad k = 1, 2, \dots.$$
(3.66)

Now, choose the feedback probabilities such that the total required service time has the same distribution as the service time in the GPS queue, cf. (3.14)-(3.16). Substitution of $\mu(j)=jf(j)$ in (3.63) and (3.66) and application of the limiting procedure described in Subsection 3.3.1 yield immediately the GPS results (3.59) and (3.60).

Although our approach to the GPS queue yields simple derivations of the results previously obtained by Cohen [1979] it seems to be very hard to derive new results such as the sojourn time distribution. The problem is the untractability of the distribution of the successive sojourn times of a customer in the feedback queue with state dependent service rate. For the derivation of the distribution of the sojourn time in the feedback queue with constant service rate, which led to the sojourn time distribution in the PS queue, we used the property that the joint process of successive departure epochs and queue length vectors at these departure epochs is a Markov renewal process, cf. the derivation of (2.8). However, this property does not hold in the feedback queue with state dependent service rate. Indeed, in the latter model a customer's sojourn time does not only depend on the number of customers of each type present at the beginning of the previous sojourn time and the number of new arrivals during it but it depends also on the epochs at which these arrivals occur. Moreover, the sojourn time is also dependent on the order of the different types of customers in the queue.
3.5 THE M/G/1 PROCESSOR SHARING QUEUE WITH FEEDBACK 3.5.1 Introduction

In this section we consider an M/G/1 PS queue with feedback. The feedback mechanism has the same structure as described in Chapter 2 for the M/M/1 FCFS queue, i.e. the probability that a customer is fed back after completing his service may depend on the number of times he has already been served. We shall study the successive sojourn times of a tagged customer. In particular we are interested in dependencies between these sojourn times.

The PS queue with feedback has been studied before by Klutke et al. [1988]. They consider the special case of Bernoulli feedback and analyze the behaviour of the internal input and output processes. In particular they study the influence of the shape of the service time distribution on the interoutput time distribution. Their main result is that when service time distributions with the same mean are convexly ordered (see Stoyan [1983]), so are interoutput time distributions. The purpose of their study is to gain insight into the properties of traffic processes in general queueing networks with processor sharing nodes.

In Klutke et al. [1988] it is remarked that the study of flow processes is crucial to understanding the behaviour of more complicated processes in the system. As an example the authors mention the sojourn time process and say that "this is still an open problem". In this section we shall show that sojourn time results for the M/G/1 PS queue with feedback can be obtained from the sojourn time results for the M/M/1 FCFS feedback queue derived in Chapter 2.

3.5.2 Model description and notations

We consider an M/G/1 PS queue with feedback (PSFB), see Fig. 3.1. When a customer in the system has completed his *i*-th service, he departs from the system with probability $1-\tilde{p}(i)$ and is fed back with probability $\tilde{p}(i)$, i=1,2,.... Fed back customers return instantaneously, i.e. due to the PS service discipline a returning customer is immediately taken into service again. The successive service requests $\tilde{\tau}_1, \tilde{\tau}_2, \ldots$ of a customer are independent random variables with distribution functions $\tilde{B}_1(\cdot), \tilde{B}_2(\cdot), \ldots$ and means $\tilde{\beta}_1, \tilde{\beta}_2, \ldots$ respectively. New customers arrive according to a Poisson process with intensity λ . Obviously, for stability it is required that the offered load $\rho = \lambda \sum_{i=1}^{\infty} ((1-\tilde{p}(j))) \prod_{i=1}^{j-1} \tilde{p}(i))(\tilde{\beta}_1 + \cdots + \tilde{\beta}_j) < 1.$

^{j=1} ⁱ⁼¹ We are interested in the successive sojourn times $\tilde{\mathbf{S}}_1(T_1), \ldots, \tilde{\mathbf{S}}_N(T_N)$ of a (tagged) customer in the PSFB queue who requires at least $N \ge 1$ services of length $T_1, \ldots, T_N \ge 0$ respectively. In particular we shall derive an expression for the correlation coefficient, $corr(\tilde{\mathbf{S}}_i(T_i), \tilde{\mathbf{S}}_j(T_j))$, of the *i*-th and the *j*-th sojourn time of a tagged customer, i, j = 1, ..., N.

For the analysis of the successive sojourn times in the PSFB queue we shall consider corresponding sojourn times in an associated processor sharing queue without feedback. Let $\hat{B}(\cdot)$ denote the distribution function of the total



Fig. 3.1 The M/G/1 PS queue with feedback.

required service time, i.e.

$$\hat{B}(t) := \sum_{j=1}^{\infty} ((1 - \tilde{p}(j)) \prod_{i=1}^{j-1} \tilde{p}(i)) (\tilde{B}_1(t)^* \cdots * \tilde{B}_j(t)), \quad t \ge 0.$$
(3.67)

It is easily seen that the behaviour of the M/G/1 PS queue with service time distribution $\hat{B}(\cdot)$ is exactly the same as the behaviour of the PSFB queue described above. In the sequel the PS queue with service time distribution $\hat{B}(\cdot)$ will be called "the associated PS queue" (or simply "the PS queue"). For a tagged customer with initial service demand $\tau^{PS} \ge T_1 + \cdots + T_N$, $T_1, \ldots, T_N \ge 0$, in the associated PS queue we define:

- $\mathbf{S}_i^{PS}(T_i)$: time during which the remaining service demand of the tagged customer is in the range $(\boldsymbol{\tau}^{PS} - \sum_{j=1}^{i} T_j, \boldsymbol{\tau}^{PS} - \sum_{j=1}^{i-1} T_j], i = 1, ..., N.$

Obviously, the joint distribution of $\mathbf{S}_1^{PS}(T_1), \ldots, \mathbf{S}_i^{PS}(T_i)$ does not depend on $T_{i+1}, \ldots, T_N, i = 1, \ldots, N-1$; $\mathbf{S}_1^{PS}(T_1)$ is distributed as the conditional sojourn time of a tagged customer with service demand T_1 :

$$E\{e^{-\omega_0 \mathbf{S}_1^{PS}(T_1)}\} = E\{e^{-\omega_0 \mathbf{S}^{PS}(T_1)}\}, \quad T_1 \ge 0, \quad \text{Re } \omega_0 \ge 0.$$
(3.68)

It is clear that the quantities $\mathbf{S}_i^{PS}(T_i)$, i = 1, ..., N in the associated PS queue correspond to the successive sojourn times $\tilde{\mathbf{S}}_1(T_1), \ldots, \tilde{\mathbf{S}}_N(T_N)$ in the PSFB queue, i.e., for $t_1, \ldots, t_N \ge 0$,

$$Pr\{\tilde{\mathbf{S}}_1(T_1) < t_1, \dots, \tilde{\mathbf{S}}_N(T_N) < t_N\} =$$
(3.69)

$$Pr\{\mathbf{S}_1^{PS}(T_1) \leq t_1, \ldots, \mathbf{S}_N^{PS}(T_N) \leq t_N\}, \quad T_1, \ldots, T_N \geq 0.$$

Specifically,

$$corr(\tilde{\mathbf{S}}_{i}(T_{i}),\tilde{\mathbf{S}}_{j}(T_{j})) = corr(\mathbf{S}_{i}^{PS}(T_{i}),\mathbf{S}_{j}^{PS}(T_{j})), \quad i,j = 1,...,N.$$
(3.70)

So below we shall focus on the sojourn times $S_i^{PS}(T_i)$, $T_i \ge 0$, i = 1, ..., N, in the PS queue.

3.5.3 Analysis

Consider the M/G/1 PS queue with service time distribution $B(\cdot)$. We assume that $\hat{B}(\cdot)$ belongs to the class of phase-type distributions given by (3.12). The first moment of $\hat{B}(\cdot)$ is denoted by $\hat{\beta}$. From Remark 3.7 it follows immediately that for $2 \le i \le N$ the joint distribution of $\mathbf{S}_i^{PS}(T_i), \ldots, \mathbf{S}_N^{PS}(T_N)$ does not depend on $T_i, j = 1, ..., i - 1$; in fact Remark 3.7 implies that, cf. (3.68),

$$E\{e^{-\omega_0 \mathbf{S}_i^{FS}(T_i)}\} = E\{e^{-\omega_0 \mathbf{S}^{FS}(T_i)}\}, \quad T_i \ge 0, \ i = 1, ..., N, \quad \text{Re } \omega_0 \ge 0.$$
(3.71)

Hence means are simply given by, see (3.19),

$$E\{\mathbf{S}_{i}^{PS}(T_{i})\} = \frac{T_{i}}{1-\rho}, \quad i = 1, ..., N,$$
(3.72)

with offered load $\rho = \lambda \hat{\beta}$. It also follows that $corr(\mathbf{S}_i^{PS}(T_i), \mathbf{S}_j^{PS}(T_j))$, $T_1, \ldots, T_N \ge 0$ depends only on T_i, T_j and $\sum_{n=i+1}^{j-1} T_n, 1 \le i < j \le N$. Hence, for the analysis of $corr(\mathbf{S}_i^{PS}(T_i), \mathbf{S}_j^{PS}(T_j)), T_1, \ldots, T_N \ge 0, i, j = 1, ..., N$ we can restrict ourself to the determination of $corr(\mathbf{S}_1^{PS}(T_1), \mathbf{S}_3^{PS}(T_3)), T_1, T_2, T_3 \ge 0$ without loss of generality. Below we shall derive an expression for $corr(\mathbf{S}_1^{PS}(T_1), \mathbf{S}_3^{PS}(T_3)), T_1, T_2, T_3 \ge 0$. We shall consider corresponding sojourn times in the M/M/1 FCFS feedback queue and apply the limiting procedure described in Subsection 3.3.1. The analysis is largely analogous to the derivation of the sojourn time variance in the M/G/1 PS queue, see Subsection 3.3.3.

Consider the M/M/1 FCFS feedback queue with mean service time β and feedback probabilities p(i), i=1,2,... related with β such that the total required service time has distribution function $\hat{B}(\cdot)$, see (3.14)-(3.16). We follow a tagged customer during his first $k=k_1+k_2+k_3$ successive sojourn times S_1, \ldots, S_k . Define

$$S_{1}(k_{1}):=S_{1}+\cdots+S_{k_{1}},$$

$$S_{2}(k_{2}):=S_{k_{1}+1}+\cdots+S_{k_{1}+k_{2}},$$

$$S_{3}(k_{3}):=S_{k_{1}+k_{2}+1}+\cdots+S_{k_{1}+k_{2}+k_{3}},$$

Clearly, when we take $k_2 = [T_2/\beta]$, $k_3 = [T_3/\beta]$, $\beta = T_1/k_1$ and let $k_1 \rightarrow \infty$ then $S_1(k_1)$, $S_2(k_2)$ and $S_3(k_3)$ correspond to the PS quantities $S_1^{PS}(T_1)$, $S_2^{PS}(T_2)$ and $S_3^{PS}(T_3)$ respectively (cf. Subsection 3.3.1; note that, for $k_1 \rightarrow \infty$, $k_i \beta \rightarrow T_i$, i = 1, 2, 3). We shall first derive corr $(S_1(k_1), S_3(k_3))$ for general $k_1, k_2, k_3 \ge 0$. Next, taking k_2, k_3 and β as indicated above we use

$$corr(\mathbf{S}_{1}^{PS}(T_{1}), \mathbf{S}_{3}^{PS}(T_{3})) = \lim_{k_{1} \to \infty} corr(\mathbf{S}_{1}(k_{1}), \mathbf{S}_{3}(k_{3})).$$
 (3.73)

From the definition of $S_i(k_i)$, i = 1,2,3, it follows that the covariance of $S_1(k_1)$ and $S_3(k_3)$ can be written as

$$cov(\mathbf{S}_1(k_1), \mathbf{S}_3(k_3)) = \sum_{i=1}^{k_1} \sum_{j=1}^{k_3} cov(\mathbf{S}_i, \mathbf{S}_{k_1+k_2+j}), \quad k_1, k_2, k_3 \ge 0.$$
 (3.74)

The covariance of S_i and S_j is found from (2.20) and (2.24):

$$cov(\mathbf{S}_i, \mathbf{S}_j) = \left(\frac{\beta}{1-\rho}\right)^2 (1-(1-\rho)C_{j-i}), \quad 1 \le i < j \le k,$$
(3.75)

with C_n , n = 1, ..., k as in Subsection 3.3.3, see (3.27)-(3.30).

Substituting (3.75) into (3.74) and writing C_n as in (3.27) it follows that, for $k_1, k_2, k_3 \ge 0$,

$$cov(\mathbf{S}_{1}(k_{1}), \mathbf{S}_{3}(k_{3})) = \left(\frac{\beta}{1-\rho}\right)^{2} \sum_{i=1}^{k_{1}} \sum_{j=1}^{k_{3}} (1-(1-\rho)C_{k_{1}+k_{2}+j-i}) \quad (3.76)$$

$$= \left(\frac{\beta}{1-\rho}\right)^{2} \left[k_{1}k_{3}-(1-\rho)\sum_{i=1}^{k_{1}} \sum_{j=1}^{k_{3}} \sum_{l=1}^{L} U_{l}x_{l}^{k_{1}+k_{2}+j-i-1}\right]$$

$$= -\frac{\beta^{2}}{1-\rho} \sum_{l=2}^{L} U_{l}x_{l}^{k_{1}+k_{2}-1} \sum_{i=1}^{k_{1}} x_{l}^{-i} \sum_{j=1}^{k_{3}} x_{l}^{j}$$

$$= -\frac{\beta^{2}}{1-\rho} \sum_{l=2}^{L} U_{l}x_{l}^{k_{2}} (\frac{x_{l}^{k_{1}}-x_{l}^{-1}}{x_{l}-1}-x_{l}^{k_{1}-1}) (\frac{1-x_{l}^{k_{3}+1}}{1-x_{l}}-1)$$

The third equality of (3.76) follows from (iii) of Lemma 3.1. Replacing in (3.76) x_l by $1-\beta a_l$, l=2,...,L, see (3.30), we obtain

$$cov(\mathbf{S}_1(k_1),\mathbf{S}_3(k_3)) = -\frac{\beta^2}{1-\rho} \times$$
(3.77)

$$\sum_{l=2}^{L} U_{l} (1-\beta a_{l})^{k_{2}} (\frac{(1-\beta a_{l})^{k_{1}} - (1-\beta a_{l})^{-1}}{-\beta a_{l}} - (1-\beta a_{l})^{k_{1}-1}) (\frac{1-(1-\beta a_{l})^{k_{3}+1}}{\beta a_{l}} - 1),$$

$$k_{1}, k_{2}, k_{3} \ge 0.$$

Now, taking in (3.77) appropriate limits, i.e. $k_2 = \lceil T_2/\beta \rceil$, $k_3 = \lceil T_3/\beta \rceil$, $\beta = T_1/k_1$ and $k_1 \rightarrow \infty$, we find, cf. (3.73):

$$cov(\mathbf{S}_{1}^{PS}(T_{1}), \mathbf{S}_{3}^{PS}(T_{3})) =$$

$$-\frac{1}{1-\rho} \sum_{l=2}^{L} U_{l}(1/a_{l})^{2} e^{-T_{2}a_{l}} (1-e^{-T_{1}a_{l}})(1-e^{-T_{3}a_{l}}), \quad T_{1}, T_{2}, T_{3} \ge 0.$$
(3.78)

Hence, from (3.71) and (3.34),

. •

$$corr(\mathbf{S}_{1}^{PS}(T_{1}), \mathbf{S}_{3}^{PS}(T_{3})) =$$

$$\frac{-\sum_{l=2}^{L} U_{l}(1/a_{l})^{2} e^{-T_{2}a_{l}}(1-e^{-T_{1}a_{l}})(1-e^{-T_{3}a_{l}})}{2\left[\sum_{l=2}^{L} (1/a_{l})^{2} U_{l}(1-T_{1}a_{l}-e^{-T_{1}a_{l}})\right]^{\frac{1}{2}} \left[\sum_{l=2}^{L} (1/a_{l})^{2} U_{l}(1-T_{3}a_{l}-e^{-T_{3}a_{l}})\right]^{\frac{1}{2}}},$$

$$T_{1}, T_{2}, T_{3} \ge 0.$$
(3.79)

Returning to the PS queue with feedback we have from (3.79) (cf. (3.70), the discussion below (3.72) and Lemma 3.1):

THEOREM 3.3

For the successive sojourn times $\tilde{\mathbf{S}}_1(T_1), \ldots, \tilde{\mathbf{S}}_N(T_N), T_1, \ldots, T_N \ge 0$, of a tagged customer in the M/G/1 PSFB queue with total service request LST given by (3.12),

$$corr(\tilde{\mathbf{S}}_{i}(T_{i}), \tilde{\mathbf{S}}_{j}(T_{j})) =$$
 (3.80)

$$\frac{-\sum_{l=2}^{L} U_{l}(1/a_{l})^{2} e^{-T_{i,j}a_{l}}(1-e^{-T_{i}a_{l}})(1-e^{-T_{j}a_{l}})}{2\left[\sum_{l=2}^{L} (1/a_{l})^{2} U_{l}(1-T_{i}a_{l}-e^{-T_{i}a_{l}})\right]^{\frac{1}{2}} \left[\sum_{l=2}^{L} (1/a_{l})^{2} U_{l}(1-T_{j}a_{l}-e^{-T_{j}a_{l}})\right]^{\frac{1}{2}}},$$

 $1 \le i \le j \le N$,

with $T_{i,j}:=\sum_{\substack{n=i+1\\n=i+1}}^{j-1}T_n$. a_2, \ldots, a_L are the roots of (3.31) and U_2, \ldots, U_L are determined by (3.29), cf. Remark 3.2; a_l and U_l are independent of T_n , n=1,...,N, l=2,...,L.

It is interesting to consider some asymptotic properties of $corr(\tilde{\mathbf{S}}_{i}(T_{i}), \tilde{\mathbf{S}}_{j}(T_{j}))$. First, noting that in (3.80) Re $a_{l} > 0$, l = 2, ..., L, see Lemma 3.1, we obtain

$$\operatorname{corr}(\tilde{\mathbf{S}}_{i}(T_{i}), \tilde{\mathbf{S}}_{j}(T_{j})) \to 0, \quad T_{i}, T_{j} \ge 0, \ T_{i,j} \to \infty, \ 1 \le i < j \le N,$$
(3.81)

which is intuitively clear. Another asymptotic result applies to the case that T_i , T_j and $T_{i,j}$ become very small. Using $\sum_{l=2}^{L} U_l = 1 - 1/(1-\rho)$ and (3.34), (3.38) it follows from (3.80) that

$$corr(\tilde{\mathbf{S}}_{i}(T_{i}), \tilde{\mathbf{S}}_{j}(T_{j})) \to 1, \quad T_{i}, T_{j}, T_{i,j} \to 0, \quad 1 \le i < j \le N.$$
(3.82)

This result can be explained as follows. Suppose a tagged customer starts his *i*-th service at time *t*. For T_i , T_j and $T_{i,j}$ close to zero it may be expected that the successive sojourn times $\tilde{\mathbf{S}}_i(T_i), \ldots, \tilde{\mathbf{S}}_j(T_j)$ of the tagged customer are small (cf. (3.72)) and no new arrivals or departures occur during the time interval $[t, t + \tilde{\mathbf{S}}_i(T_i) + \ldots + \tilde{\mathbf{S}}_j(T_j)]$. Hence, due to the PS service discipline $\tilde{\mathbf{S}}_j(T_j) = T_j \tilde{\mathbf{S}}_i(T_i) / T_i$, i.e. $\tilde{\mathbf{S}}_j(T_j)$ is completely determined by $\tilde{\mathbf{S}}_i(T_i)$.

We conclude this section with an example.

The M/M/1 PS queue with Bernoulli feedback

Consider the M/M/1 PS queue with Bernoulli feedback, i.e. $B(t) = 1 - e^{-t/\beta}$, $\tilde{p}(i) \equiv p, 0 \leq p < 1$, see Subsection 3.5.2. For this case the total required service time is exponentially distributed with mean $\hat{\beta} = \tilde{\beta}/(1-p)$, cf. (3.67). From the calculations for the determination of the sojourn time variance in the M/M/1 PS queue, see Subsection 3.3.3, we have in (3.80) L = 2, $U_2 = -1/(1-\rho)$, $a_2 = (1-x_2)/\beta = (1-\rho)/\hat{\beta}$. Hence, for the M/M/1 queue with Bernoulli feedback (3.80) reduces to

$$corr(\tilde{\mathbf{S}}_{i}(T_{i}), \tilde{\mathbf{S}}_{j}(T_{j})) =$$
(3.83)

$$\frac{e^{-T_{i,j}(1-\rho)/\hat{\beta}}(1-e^{-T_{i}(1-\rho)/\hat{\beta}})(1-e^{-T_{j}(1-\rho)/\hat{\beta}})}{2\left[e^{-T_{i}(1-\rho)/\hat{\beta}}-1+T_{i}(1-\rho)/\hat{\beta}\right]^{\frac{1}{2}}\left[e^{-T_{j}(1-\rho)/\hat{\beta}}-1+T_{j}(1-\rho)/\hat{\beta}\right]^{\frac{1}{2}}}, \quad 1 \le i < j \le N.$$

Chapter 4

SIMPLE APPROXIMATIONS FOR SECOND MOMENT CHARACTERISTICS OF THE SOJOURN TIME IN THE M/G/1 PROCESSOR SHARING QUEUE

4.1 INTRODUCTION

Although in literature considerable attention has been paid to the exact analysis of the sojourn time in the M/G/1 PS queue, little work has been done on the investigation of the practical implications of the results. The expressions for the second moment and the LST of the sojourn time distribution obtained by Yashkov [1983], Ott [1984] and Schassberger [1984], see also (3.34), (3.35) and (3.55), are complex and not very attractive for practical applications. Only for the mean sojourn time a simple explicit expression exists; this expression is insensitive to the service time distribution apart from its first moment, see (3.19). The formulas for the second moment of the sojourn time require perfect information about the service time distribution, which is almost never available in practice. Moreover, in general these formulas can only be evaluated numerically. As far as we know no attention has been paid to the derivation of approximations or asymptotic formulas which are useful for practical evaluation, apart from a paper by Yashkov [1986]. Yashkov derives some asymptotic estimates for the conditional sojourn time variance for customers with small or large service times. We have obtained similar results in Section 3.3, see (3.37) and (3.38). In particular the asymptotic formula (3.38), for customers with small service times, slightly generalizes Yashkov's result, cf. Remark 3.3.

Actually, our interest in approximations for second moment characteristics of the sojourn time in the M/G/1 PS queue started with the derivation of (3.37) and (3.38). We found that these asymptotic formulas yield reasonable estimates for a wide range of the parameter values. The discovery of simple bounds for the second moment of the sojourn time also stimulated the investigation of approximations. Indeed, noting that in (3.7) $C_n \ge 1$, n = 1, 2, ..., it is easily seen that, for the total sojourn time after k services in the feedback queue,

$$var(\mathbf{S}^{(k)}) \leq \frac{\rho}{(1-\rho)^2} (k\beta)^2, \quad k=1,2,...$$

Hence, applying the limiting procedure described in Section 3.3,

$$\operatorname{var}(\mathbf{S}^{PS}(x)) \leq \frac{\rho}{\left(1-\rho\right)^2} x^2 \,. \tag{4.1}$$

Note that this bound depends on the service time distribution only through its first moment. Obviously, (4.1) implies an upper bound for the unconditional sojourn time variance which depends on the first two moments of the service time distribution. After having noticed the existence of these bounds we found out that they can also be easily obtained from the results in Yashkov [1983] and Ott [1984]. However, neither Yashkov nor Ott points at this interesting property of the sojourn time variance in the PS queue.

The aim of the present study is to derive approximations for the second moment of the sojourn time distribution, which are quite simple and yet accurate enough for most practical purposes. Some very simple approximation formulas based on the first and second moment of the service time are presented. The accuracy of the approximations is tested for a large number of different service time distributions and a wide range of traffic intensities. A refinement of the approximation is obtained by taking the third moment of the service time into account. This refinement yields remarkably accurate results with relative errors less than 1.5 percent in most cases.

The organization of the rest of this chapter is as follows. In Section 4.2 we introduce the notations and give a summary of those known sojourn time results which are relevant for our study. We also present some extensions and new results. In particular the heavy traffic behaviour of the second moment of the sojourn time is derived. Section 4.3 is concerned with the second moment of the *conditional* sojourn time of a customer with service demand x. We show that the asymptotic result (3.38) yields reasonable approximations for a wide range of x values. In Section 4.4 approximations are developed for the second moment of the *unconditional* sojourn time. We first propose an approximation which uses only information about the first and second moment of the service time distribution (Subsection 4.4.1). In Subsection 4.4.2 we construct a more detailed (and more accurate) approximation formula, which is based on the first three moments of the service time distribution.

4.2 NOTATIONS AND PRELIMINARY RESULTS

We consider an M/G/1 PS queue with arrival rate λ and service time distribution $\hat{B}(\cdot)$ with first and second moment $\hat{\beta}$ and $\hat{\beta}_2$. It is assumed that $\rho:=\lambda\hat{\beta}<1$ and that the system is in steady state. Since confusion with the sojourn time in the feedback queue is not possible we use in the rest of this

chapter S(x) and S instead of $S^{PS}(x)$ and S^{PS} to denote the conditional and unconditional sojourn time in the PS queue.

.

The sojourn time formulas obtained by Yashkov [1983] and Ott [1984] are more suitable for heavy traffic analysis than our expressions derived in Chapter 3. They have for the second moment of the conditional sojourn time,

$$E\{\mathbf{S}^{2}(x)\} = \frac{x^{2}}{(1-\rho)^{2}} + \frac{2}{(1-\rho)^{2}} \int_{t=0}^{x} (x-t)(1-R(t)) dt , \qquad (4.2)$$

where R(t) represents the waiting time distribution for the M/G/1 first come first served (FCFS) queue with service time distribution $\hat{B}(\cdot)$,

$$R(t) = (1-\rho) \sum_{n=0}^{\infty} \rho^n F^{n^*}(t) , \qquad (4.3)$$
$$F(t) = \frac{1}{\hat{\beta}} \int_{u=0}^{t} (1-\hat{B}(u)) \, du .$$

Note, for the waiting time distribution R(t) in (4.2), that $1-R(t) \le 1-R(0) = \rho$, $t \ge 0$. Hence, cf. (4.1),

$$E\{\mathbf{S}^{2}(x)\} \leq \frac{1+\rho}{(1-\rho)^{2}}x^{2}.$$
(4.4)

A lower bound for $E{S^2(x)}$ follows immediately from the mean sojourn time, given by (3.19), and Schwartz' inequality (or alternatively from (4.2)),

$$E\{\mathbf{S}^{2}(x)\} \ge \frac{x^{2}}{(1-\rho)^{2}}$$
 (4.5)

So,

$$\frac{1}{(1-\rho)^2} x^2 \le E\{\mathbf{S}^2(x)\} \le \frac{1+\rho}{(1-\rho)^2} x^2 \,. \tag{4.6}$$

Note that the upper bound is $100\rho\%$ higher than the lower bound and that these bounds depend only on the mean service time; this supports a certain robustness of $E\{\mathbf{S}^2(x)\}$ for the service time distribution.

The heavy traffic behaviour of $E{S^2(x)}$ can be derived from (4.2) by noting that the heavy traffic behaviour of the waiting time distribution for the M/G/1 FCFS queue is, for $\rho \rightarrow 1$, negative exponential, i.e. (see Cohen [1982, p. 596])

$$R(\frac{t}{1-\rho}) \sim 1 - e^{-t/d}, \quad \text{for } \rho \to 1,$$
(4.7)

where $d = \frac{1}{2}\rho\hat{\beta}_2/\hat{\beta}$.

Substituting (4.7) into (4.2) yields

$$E\{\mathbf{S}^{2}(x)\} \sim \frac{1+\rho}{(1-\rho)^{2}}x^{2}$$
, for $\rho \to 1$. (4.8)

The asymptotic behaviour of $E{\{S^2(x)\}}$ for $x\to 0$ is given by (3.38) and (3.19):

$$E\{\mathbf{S}^{2}(x)\} \sim \frac{1+\rho}{(1-\rho)^{2}} x^{2} - \frac{\rho}{3\hat{\beta}(1-\rho)} x^{3}, \quad \text{for } x \to 0.$$
(4.9)

For exponential and deterministic service times, simple explicit expressions for $E S^{2}(x)$ exist. For future use we state these expressions. From (3.23) and (3.57) it follows that

$$E\{\mathbf{S}^{2}(x)\}_{EXP} = \frac{2\rho\hat{\beta}}{(1-\rho)^{3}}x - \frac{2\rho\hat{\beta}^{2}}{(1-\rho)^{4}}(1-e^{-x(1-\rho)/\hat{\beta}}), \quad x \ge 0, \quad (4.10)$$

$$E\{\mathbf{S}^{2}(x)\}_{DET} = \frac{2}{(1-\rho)^{2}}x^{2} - \frac{2\hat{\beta}^{2}}{\rho^{2}(1-\rho)}(e^{\rho x/\hat{\beta}} - 1 - \rho x/\hat{\beta}), \quad 0 \le x \le \hat{\beta} . \quad (4.11)$$

From (3.19) and the above results for S(x) it follows immediately that for the unconditional sojourn time S, cf. (3.18),

$$E\{\mathbf{S}\} = \frac{\hat{\boldsymbol{\beta}}}{1-\rho} , \qquad (4.12)$$

$$\frac{\hat{\beta}_2}{(1-\rho)^2} \le E\{\mathbf{S}^2\} \le \frac{1+\rho}{(1-\rho)^2}\hat{\beta}_2 , \qquad (4.13)$$

$$E\{\mathbf{S}^2\} \sim \frac{1+\rho}{(1-\rho)^2} \hat{\beta}_2 , \quad \text{for } \rho \to 1 .$$
 (4.14)

For exponential and deterministic service times,

$$E\{\mathbf{S}^2\}_{EXP} = (1 + \frac{2+\rho}{2-\rho})\frac{\hat{\beta}^2}{(1-\rho)^2}, \qquad (4.15)$$

 $E\{\mathbf{S}^2\}_{DET} = \frac{2\hat{\beta}^2}{(1-\rho)^2} - \frac{2\hat{\beta}^2}{\rho^2(1-\rho)}(e^{\rho}-1-\rho).$ (4.16)

Remark 4.1

(4.13) implies that, for the M/G/1 PS queue, the dependence of $E\{S^2\}$ on the third moment of the service time distribution is limited. This should be contrasted with the behaviour of the second moment of the sojourn time distribution for the M/G/1 FCFS queue. For the FCFS discipline it depends linearly on the third moment of the service time distribution, see e.g. Cohen [1982].

In the following sections the above results are exploited to develop simple approximations for $E\{S^2(x)\}$ and $E\{S^2\}$. We present extensive tables comparing the approximations with exact values. The service time distributions which we have chosen to test the approximations are:

- exponential distribution
- deterministic distribution
- k-stage Erlang distribution (E_k)
- two-stage hyperexponential distribution (H_2) , in particular H_2 with balanced means (H_2^{BM}) , and
- H_2 with gamma normalization (H_2^{GN})
- two-stage Coxian distribution (C_2)
- three-stage hyperexponential distribution (H_3)

These types of service time distributions are often used for practical applications in queueing theory, see Tijms [1986] and Whitt [1982, 1984].

In practice service times are often characterized by the mean, β , and the squared coefficient of variation, c^2 , defined by

$$c^2=rac{\sigma^2}{\hat{oldsymbol{eta}}^2}\,,$$

where σ^2 denotes the service time variance, see Tijms [1986]. Here we shall use c^2 rather than σ^2 to characterize the variability of the service times. The H_2^{BM} and H_2^{GN} distributions have been introduced to reduce the number

The H_2^{BM} and H_2^{GN} distributions have been introduced to reduce the number of parameters of the H_2 distribution, see Tijms [1986]; they are uniquely determined by their first two moments. In particular, the H_2^{GN} distribution with mean $\hat{\beta}$ and $c^2 \ge 1$ has the same third moment as the gamma distribution with mean $\hat{\beta}$ and squared coefficient of variation c^2 . In Section 4.4 the class of H_2 distributions will be considered in more detail.

The tables presented at the end of this chapter contain relative errors of the approximations for various service time distributions. The relative error is defined as

The exact values of $E{S^2(x)}$ and $E{S^2}$ have been obtained from (3.34) and

(3.35). For H_k and C_k (and E_k) service time distributions these formulas require the roots of a polynomial of degree k and the solution of a set of k linear equations. Even for the case k=2, the resulting expressions are very large and complicated and do not give much insight into the influence of the parameters.

4.3 Approximation of $E{\mathbf{S}^2(x)}$

In this section we show that the asymptotic result (4.9) yields a good approximation for $E\{S^2(x)\}$ for an important range of x-values.

We define (cf. (4.9)),

$$E\{\mathbf{S}^{2}(x)\}_{APPX} := \frac{1+\rho}{(1-\rho)^{2}}x^{2} - \frac{\rho}{3\hat{\beta}(1-\rho)}x^{3}.$$
(4.17)

Note that $E\{\mathbf{S}^2(x)\}_{APPX}$ satisfies the heavy traffic behaviour of $E\{\mathbf{S}^2(x)\}$ (see (4.8)) and that $E\{\mathbf{S}^2(x)\}_{APPX}$ is smaller than the upper bound of $E\{\mathbf{S}^2(x)\}$ given by (4.4). Approximation $E\{\mathbf{S}^2(x)\}_{APPX}$ is independent of the service time distribution apart from its first moment. Obviously it can not be applied for too large values of x because it becomes negative for $x > 3\beta(1+\rho)/(\rho(1-\rho))$. Moreover, assuming that the variance of $\mathbf{S}(x)$ is a convex function of x (cf. (4.9) and (3.37)), we may not expect that $E\{\mathbf{S}^2(x)\}_{APPX}$ is a good approximation for $x > x_1$, where $x_1 = \beta/(1-\rho) = E\{\mathbf{S}\}$ is the point of inflection of (cf. (3.38))

$$f(x) = \frac{\rho}{(1-\rho)^2} x^2 - \frac{\rho}{3\hat{\beta}(1-\rho)} x^3 .$$
(4.18)

For $x < x_1$, $E\{\mathbf{S}^2(x)\}_{APPX}$ is within the bounds of $E\{\mathbf{S}^2(x)\}$ given by (4.6).

In Table 4.1 approximation results are compared with exact results for a number of different values of x $(x = \frac{1}{2}\hat{\beta}, \hat{\beta}, \frac{3}{2}\hat{\beta}, 2\hat{\beta}, \frac{\hat{\beta}/2}{1-\rho}, \frac{\hat{\beta}}{1-\rho})$ and different service time distributions. For each of these cases ρ varies from 0.1 to 0.9. For the sake of clarity only the relative approximation errors are given. It appears that for most cases the relative approximation errors are negative. As expected, the approximation becomes less accurate when x grows. For $0 \le x \le \hat{\beta}$ the relative errors are less than 2.34% in absolute value. For $0 \le x \le \hat{\beta}$ the maximum relative error is 6.56%. When x remains constant the maximum errors occur for $\rho \approx 0.3$. For $x = \hat{\beta}/(2(1-\rho))$ and $x = \hat{\beta}/(1-\rho)$ the relative error is 11.29%.

It is seen from the results for different service time distributions that the accuracy of the approximation tends to decrease when c^2 becomes larger.

Remark 4.2

In Van den Berg [1988] we have derived an approximation for $E\{S^2(x)\}$ for the whole range of possible x-values $(x \ge 0)$ by appropriately combining the two asymptotic formulas (4.9) and (3.37). The idea is as follows. Two values \tilde{x}_1 and \tilde{x}_2 are determined, such that for $x \le \tilde{x}_1$ (4.9) yields a good approximation and for $x \ge \tilde{x}_2$ (3.37) yields a good approximation. For $\tilde{x}_1 \le x \le \tilde{x}_2$, $E\{S^2(x)\}$ is approximated by the term $x^2/(1-\rho)^2$ plus a linear function of x, cf. (3.37). We took $\tilde{x}_1 = \hat{\beta}/(1-\rho)$. Details about the determination of \tilde{x}_2 are given in Van den Berg [1988]. The approximation yields reasonably good results for service time distributions with c^2 not too large $(0 \le c^2 \le 2)$. For these cases we found relative errors which are typically less than 10%. A minor drawback of this approximation is that it needs the first *three* moments of the service time distribution (cf. (3.37)).

4.4 Approximation of $E\{\mathbf{S}^2\}$

In this section we propose two different approximations for the second moment of the unconditional sojourn time S. First we derive a simple approximation which is based on the exact formula of $E\{S^2\}$ for the case of deterministic service times and for exponentially distributed service times. This approximation uses only the first two moments of the service time distribution. Next it is shown how this simple approximation can be improved. We derive a (second) approximation based on exact expressions of $E\{S^2\}$ for two classes of H_2 distributions. This latter approximation also takes the *third* moment of the service time distribution into account.

4.4.1 Simple approximation

It follows from (4.13) that an approximation $E\{\mathbf{S}^2\}_{APP}$ of $E\{\mathbf{S}^2\}$, which satisfies

$$\frac{\hat{\beta}_2}{(1-\rho)^2} \le E\{\mathbf{S}^2\}_{APP} \le \frac{1+\rho}{(1-\rho)^2}\hat{\beta}_2 , \qquad (4.19)$$

yields relative errors which are bounded by $100\rho\%$ in absolute value. This observation and the relations for $E\{S^2\}$ given in Section 4.2 support the idea to derive an approximation for $E\{S^2\}$ which is based only on the first two moments of the service time distribution. We propose an approximation which is a linear interpolation on the service time squared coefficient of variation such that it yields exact results for the case of exponential and deterministic service times. This type of two-moment approximations is often used to estimate performance measures (e.g. mean sojourn times) in (complex) queueing systems, see Tijms [1986, Ch. 4]. The rationale behind it is that the Pollaczek-Khinchine formula for the mean sojourn time $E\{S^{FCFS}\}$ in the M/G/1 FCFS queue allows the representation

$$E\{\mathbf{S}^{FCFS}\} = c^2 E\{\mathbf{S}^{FCFS}\}_{EXP} + (1-c^2) E\{\mathbf{S}^{FCFS}\}_{DET}, \qquad (4.20)$$

where $E\{\mathbf{S}^{FCFS}\}_{EXP}$ and $E\{\mathbf{S}^{FCFS}\}_{DET}$ denote the mean sojourn time for the special cases of exponential and deterministic service times (with the same means). Note that this kind of representation is also allowed for the mean sojourn time in the M/G/1 PS queue, cf. (4.12). For the present case of the second moment of the sojourn time distribution in the M/G/1 PS queue this idea leads to an approximation $E\{\mathbf{S}^2\}_{APP1}$ for $E\{\mathbf{S}^2\}$ which reads as follows:

$$E\{\mathbf{S}^2\}_{APP1} = c^2 E\{S^2\}_{EXP} + (1-c^2) E\{\mathbf{S}^2\}_{DET} =$$
(4.21)

$$c^{2}(1+\frac{2+\rho}{2-\rho})\frac{\hat{\beta}^{2}}{(1-\rho)^{2}}+(1-c^{2})(\frac{2\hat{\beta}^{2}}{(1-\rho)^{2}}-\frac{2\hat{\beta}^{2}}{\rho^{2}(1-\rho)}(e^{\rho}-1-\rho)).$$

Note that this approximation has the following appealing properties:

APPROXIMATION PROPERTIES

- (1) The approximation is exact for deterministic service times.
- (2) The approximation is exact for exponentially distributed service times.
- (3) The approximation yields values between the lower and upper bound of $E\{\mathbf{S}^2\}$ given by (4.13).
- (4) The approximation satisfies the heavy traffic behaviour of $E\{S^2\}$ (see (4.14)).
- (5) The approximation yields the exact value of $E\{\mathbf{S}^2\}$ for $\rho=0$: $E\{\mathbf{S}^2\}_{APP} = E\{\mathbf{S}^2\} = \hat{\boldsymbol{\beta}}_2$, for $\rho=0$.

The approximation results for the test set of service time distributions and traffic intensities are presented in Table 4.2. It appears that the approximation yields reasonably good results. In all tested cases the relative approximation error is smaller than 5%. In particular for service time distributions with c^2 close to one ($0 \le c^2 \le 2$) the relative errors are less than 1.89%. Obviously this small error is due to the fact that the approximation is exact for exponential and deterministic service times. For larger values of c^2 ($c^2 > 4$) the approximation becomes worse. It is noticeable that the approximation is significantly better for the H_2 distribution with gamma normalization (H_2^{GN}) than for the H_2 distribution with balanced means (H_2^{BM}). In the next subsection we shall show that this is due to the influence of the third moment of the service time distribution on $E\{\mathbf{S}^2\}$.

4.4.2 Detailed approximation

The simple approximation (4.21) tends to be less accurate if the squared coefficient of variation of the service time distribution becomes larger.

Therefore, for service time distributions with $c^2 \ge 1$ we shall develop a new approximation, *APP2*, for $E\{S^2\}$. This approximation is based on simple exact formulas for two classes of extreme H_2 distributions. It contains the first three moments of the service time distribution.

We start with recalling some characteristics of the class of H_2 distributions. The H_2 distribution function is given by

$$B_{H_2}(t) = \alpha (1 - e^{-t/\hat{\beta}^{(1)}}) + (1 - \alpha) (1 - e^{-t/\hat{\beta}^{(2)}}), \qquad (4.22)$$

where $0 \leq \alpha \leq 1$, $0 \leq \hat{\beta}^{(1)} \leq \hat{\beta}^{(2)}$.

So, there are three parameters. Given the mean $\hat{\beta} = \alpha \hat{\beta}^{(1)} + (1-\alpha)\hat{\beta}^{(2)}$ and $c^2 \ge 1$ there is thus one remaining degree of freedom, *r*, defined by

$$r = \frac{\alpha \hat{\beta}^{(1)}}{\alpha \hat{\beta}^{(1)} + (1-\alpha) \hat{\beta}^{(2)}}$$

r = 1/2 yields the class of H_2 distributions with balanced means (H_2^{BM}) .

Obviously, if $\hat{\beta}$ and c^2 are given, r determines the third moment, $\hat{\beta}_3$, of the H_2 distribution. For fixed $\hat{\beta}$ and $\hat{\beta}_2$ (c^2), the smallest possible value of $\hat{\beta}_3$ is obtained for r=0. In that case $\hat{\beta}_3 = \frac{3}{2} \hat{\beta}_2^2 / \hat{\beta}$. For $r \to 1$, $\hat{\beta}_3 \to \infty$ (see Whitt [1982,1984]).

Our numerical experience with respect to H_2 distributions indicates that $E\{\mathbf{S}^2\}$ becomes smaller when $\hat{\beta}_3$ grows ($\hat{\beta}$ and $\hat{\beta}_2$ constant). So (cf. (4.13)), we expect that $E\{\mathbf{S}^2\}_{H_2}$ has a limit for $\hat{\beta}_3 \rightarrow \infty$, $\hat{\beta}$ and $\hat{\beta}_2$ fixed. From (3.35) it is found that, for $\hat{\beta}$ and $\hat{\beta}_2$ fixed,

$$E\{\mathbf{S}^2\}_{H_2^{r=1}} = \lim_{\hat{\beta}_3 \to \infty(r \to 1)} E\{\mathbf{S}^2\}_{H_2} = \frac{\hat{\beta}_2}{(1-\rho)^2} + \frac{2\rho}{2-\rho} \frac{\hat{\beta}^2}{(1-\rho)^2} , \qquad (4.23)$$

and, for $\hat{\beta}_3 = \frac{3}{2} \hat{\beta}_2^2 / \hat{\beta}$ (r=0),

$$E\{\mathbf{S}^2\}_{H_2'^{=0}} = (1 + \frac{\rho}{2-\rho})\frac{\hat{\beta}_2}{(1-\rho)^2} .$$
(4.24)

It is easily seen that, for $c^2 \ge 1$,

$$E\{\mathbf{S}^2\}_{H_2^{\prime=1}} \leq E\{\mathbf{S}^2\}_{H_2^{\prime=0}}, \quad 0 \leq \rho \leq 1.$$
(4.25)

In (4.25), equality holds if $c^2 = 1$ ($\hat{\beta}_2 = 2\hat{\beta}^2$), hence if the service times are exponentially distributed.

Now we introduce two approximation assumptions to extend the above results with respect to H_2 distributions to general service time distributions.

Assumption 1: $E\{\mathbf{S}^2\}$ depends only on the first three moments $(\hat{\beta}, \hat{\beta}_2, \hat{\beta}_3)$ of the service time distribution.

Assumption 2: $E\{\mathbf{S}^2\}$ decreases if $\hat{\beta}_3$ grows ($\hat{\beta}$ and $\hat{\beta}_2$ fixed).

Under these assumptions it follows from (4.23) and (4.24) that, for $c^2 \ge 1$, $\hat{\beta}_3 \ge \frac{3}{2}\hat{\beta}_2^2/\hat{\beta}$,

$$\frac{\hat{\beta}_2}{(1-\rho)^2} + \frac{2\rho}{2-\rho}\frac{\hat{\beta}^2}{(1-\rho)^2} \le E\{\mathbf{S}^2\} \le (1+\frac{\rho}{2-\rho})\frac{\hat{\beta}_2}{(1-\rho)^2}.$$
 (4.26)

(4.23), (4.24) and (4.26) suggest an approximation, APP2, for $E{S^2}$ which reads as follows:

$$E\{\mathbf{S}^2\}_{APP2} = \gamma(1+\frac{\rho}{2-\rho})\frac{\hat{\beta}_2}{(1-\rho)^2} + (1-\gamma)(\frac{\hat{\beta}_2}{(1-\rho)^2} + \frac{2\rho}{2-\rho}\frac{\hat{\beta}^2}{(1-\rho)^2}), \quad (4.27)$$

where $\gamma := \gamma(\rho, \hat{\beta}, \hat{\beta}_2, \hat{\beta}_3), \ 0 \le \gamma \le 1.$

The choice of the weight factor γ will be partially determined by the approximation properties listed below (4.21). Besides the properties (2)-(5) we require that

- (6) for $\hat{\beta}$, $\hat{\beta}_2$ fixed, $\lim_{\hat{\beta}_3 \to \infty} E S^2_{APP2} = \frac{\hat{\beta}_2}{(1-\rho)^2} + \frac{2\rho}{2-\rho} \frac{\hat{\beta}^2}{(1-\rho)^2} ,$
- (7) for $\hat{\beta}_3 = \frac{3}{2} \hat{\beta}_2^2 / \hat{\beta}$, $E\{\mathbf{S}^2\}_{APP2} = (1 + \frac{\rho}{2-\rho}) \frac{\hat{\beta}_2}{(1-\rho)^2}$.

Note, that, without any further specification of γ , *APP* 2 satisfies the approximation properties (2) and (5). Considering the other required properties ((3), (4), (6) and (7)) it is natural to choose γ as follows,

$$\gamma = \frac{1}{1 + \gamma_1(\hat{\beta}_3 - \frac{3}{2}\hat{\beta}_2^2 / \hat{\beta})(1 - \rho)},$$
(4.28)

where γ_1 represents the relative influence of $\hat{\beta}_3$ on $E\{\mathbf{S}^2\}$. γ_1 remains to be

specified. We assume that γ_1 depends only on $\hat{\beta}$ and $\hat{\beta}_2$. Note that γ_1 has to be chosen such that γ is dimensionless. The most obvious choices are $\gamma_1 = 1/\hat{\beta}^3$ or $\gamma_1 = 1/(\hat{\beta}\hat{\beta}_2)$. For both cases we compared approximation results with exact results. Our test set consisted of H_2 service time distributions with c^2 ranging from 1 to 20. For each value of c^2 a large number of $\hat{\beta}_3$ values was considered. It appeared that the choice $\gamma_1 = 1/(\hat{\beta}\hat{\beta}_2)$ yields much better results than $\gamma_1 = 1/\hat{\beta}$. However, in most cases the choice $\gamma_1 = 1/(\hat{\beta}\hat{\beta}_2)$ underestimated $E\{S^2\}$. In particular for larger values of c^2 the approximation results became worse. Extensive tests of the approximation for some variants of $\gamma_1 = 1/(\hat{\beta}\hat{\beta}_2)$ led to a modification which yields remarkably accurate results:

$$\gamma_1 = \frac{1}{(c^2-1)} \frac{1}{\hat{\beta}\hat{\beta}_2}$$

So, the ultimate approximation formula is given by (4.27), with

$$\gamma = \frac{1}{1 + (1 - \rho)(\frac{\hat{\beta}_3}{\hat{\beta}\hat{\beta}_2} - \frac{3}{2}\frac{\hat{\beta}_2}{\hat{\beta}^2})/(\frac{\hat{\beta}_2}{\hat{\beta}^2} - 2)}.$$
(4.29)

It is seen from Table 4.3 that for H_2^{BM} and H_2^{GN} service time distributions (with $c^2=2$, 4, 6) APP2 yields very accurate results with relative errors less than 1%.

Table 4.4 illustrates the influence of $\hat{\beta}_3$ on $E\{\mathbf{S}^2\}$. This table shows exact values of $E\{\mathbf{S}^2\}$ for a number of H_2 distributions with the same first and second moment but with a different third moment. The traffic intensity varies from 0.1 to 0.95. The relative approximation errors of *APP2* are indicated below the exact values of $E\{\mathbf{S}^2\}$. As we stated before $E\{\mathbf{S}^2\}$ decreases when $\hat{\beta}_3$ grows. Note that even for large c^2 ($c^2=10$) the relative approximation errors are less than 1.5%. It may be concluded from Table 4.4 that the influence of the third moment of the service time distribution on $E\{\mathbf{S}^2\}$ increases when c^2 grows, cf. (4.13).

In Table 4.5 APP2 is tested for some arbitrarily chosen H_3 and C_2 service time distributions. The relative errors are in all cases less than 1.5%.

Originally, APP2 has been developed for service time distributions with $c^2 \ge 1$, $\hat{\beta}_3 \ge \frac{3}{2} \hat{\beta}_2^2 / \hat{\beta}$. For these cases $E\{\mathbf{S}^2\}_{APP2}$ can be interpreted as an interpolation formula, see (4.27). Nevertheless, approximation formula (4.27) (together with (4.29)) can be applied to service time distributions with $c^2 < 1$ or $\hat{\beta}_3 < \frac{3}{2} \hat{\beta}_2^2 / \hat{\beta}$ as well. In Table 4.6 some results are shown for deterministic, C_2 and E_k service time distributions with $c^2 < 1$. It appears that the accuracy of APP2 for these cases is about the same as the accuracy of APP1.

4.5 CONCLUSIONS

In this section we briefly review the results and sum up the main characteristics of the different approximation formulas proposed in Section 4.3 and Section 4.4.

We have studied the second moment of the conditional and unconditional sojourn time, $E{S^2(x)}$ and $E{S^2}$, for the M/G/1 processor sharing queue. An upper bound and some asymptotic properties (like the heavy traffic behaviour) have been derived. Based on these properties and on exact expressions for specific service time distributions we developed some simple approximations. The approximations have been compared with exact results for a large number of different service time distributions and a wide range of traffic intensities. We conclude as follows.

- The influence of the third and higher moments of the service time distribution on $E\{\mathbf{S}^2(x)\}$ and $E\{\mathbf{S}^2\}$ is limited. An upper and a lower bound for $E\{\mathbf{S}^2(x)\}$ can be expressed in terms of x (the service demand of a tagged customer) and the traffic intensity ρ , see (4.6). The corresponding upper and lower bound for $E\{\mathbf{S}^2\}$ contain only the second moment of the sojourn time distribution and ρ , see (4.13).

- Approximation APPX for $E\{S^2(x)\}$, given by (4.17), is based on the asymptotic result (3.38) for $x \rightarrow 0$. It depends on the service time distribution only through its first moment. APPX yields reasonably good results for not too large values of x, see Table 4.1. For $0 \le x \le \hat{\beta}$ the relative error of the approximation is a few percent. The approximation becomes less accurate when x increases. For $x = 2\hat{\beta}$ the relative errors are typically less than 7%. APPX satisfies the heavy traffic behaviour of $E S^2(x)$, see (4.8).

- The approximations for $E\{S^2\}$, APP 1 and APP 2, given by (4.21) and (4.27), have been constructed in such a way that they have the following appealing properties:

* they are exact for exponential service times

* they yield values between the lower and upper bound of $E{\{S^2\}}$

* they satisfy the heavy traffic behaviour of $E{\{S^2\}}$

* they yield the exact value of $E{S^2}$ for $\rho=0$.

In addition, APP1 yields exact results for deterministic service times; APP2 is exact for two classes of extreme H_2 distributions.

- Approximation APP 1 is the most simple approximation. It depends on the first two moments of the service time distribution. For not too large values of c^2 ($c^2 \le 6$) it yields fairly accurate results, see Table 4.2. In practical situations APP 1 may be applied as a first order approximation for $E\{S^2\}$.

- APP2 depends on the first three moments of the service time distribution. It is based on exact formulas of $E\{S^2\}$ for two classes of extreme H_2 distributions. The details of the construction of APP2 are rather heuristic. Nevertheless, it yields remarkably accurate results. APP2 has been tested for a large number of different service time distributions with c^2 ranging from 0 to 10, see Tables 4.3 through 4.6. In all of these cases the relative error is less than 1.5%.

TABLE 4.1. Approximation of $E\{S^2(x)\}$. The table contains relative errors (%) of approximation APPX (given by (4.17)) for various service time distributions.

Service time distribution: H_2^{BM} , $c^2 = 2$.

•

ρ	$x = \frac{1}{2}\hat{\beta}$	$x = \hat{\beta}$	$x = \frac{3}{2}\hat{\beta}$	$x=2\hat{\beta}$	$x = \frac{1}{2} \frac{\hat{\beta}}{1 - \rho}$	$x = \frac{\hat{\beta}}{1 - \rho}$
0.1	-0.19	-0.66	-1.35	-2.18	-0.23	-0.80
0.3	-0.32	-1.16	-2.40	-3.94	-0.62	-2.20
0.5	-0.27	-0.99	-2.08	-3.46	-0.99	-3.46
0.7	-0.15	-0.56	-1.19	-1.99	-1.44	-4.80
0.9	-0.04	-0.15	-0.30	-0.51	-2.31	-6.61

Service time distribution: H_2^{BM} , $c^2 = 4$.

ρ	$x = \frac{1}{2}\hat{\beta}$	$x = \hat{\beta}$	$x = \frac{3}{2}\hat{\beta}$	$x=2\hat{\beta}$	$x = \frac{1}{2} \frac{\hat{\beta}}{1 - \rho}$	$x = \frac{\hat{\beta}}{1 - \rho}$
0.1	-0.22	-0.77	-1.55	-2.46	-0.27	-0.93
0.3	-0.39	-1.39	-2.83	-4.58	-0.76	-2.61
0.5	-0.35	-1.26	-2.58	-4.23	-1.26	-4.23
0.7	-0.21	-0.77	-1.60	-2.64	-1.93	-6.11
0.9	-0.06	-0.23	-0.48	-0.79	-3.39	-8.98

Service time distribution: H_2^{BM} , $c^2 = 6$.

ρ	$x = \frac{1}{2}\hat{\beta}$	$x = \hat{\beta}$	$x = \frac{3}{2}\hat{\beta}$	$x=2\hat{\beta}$	$x = \frac{1}{2} \frac{\hat{\beta}}{1 - \rho}$	$x = \frac{\hat{\beta}}{1 - \rho}$
0.1	-0.24	-0.82	-1.63	-2.59	-0.29	-0.99
0.3	-0.42	-1.50	-3.02	-4.86	-0.81	-2.78
0.5	-0.38	-1.37	-2.80	-4.56	-1.37	-4.56
0.7	-0.24	-0.86	-1.78	-2.92	-2.14	-6.67
0.9	-0.07	-0.27	-0.55	-0.91	-3.85	-10.01

Service time distribution: exponential.

ρ	$x = \frac{1}{2}\hat{\beta}$	$x = \hat{\beta}$	$x = \frac{3}{2}\hat{\beta}$	$x=2\hat{\beta}$	$x = \frac{1}{2} \frac{\hat{\beta}}{1 - \rho}$	$x = \frac{\hat{\beta}}{1 - \rho}$
0.1	-0.14	-0.53	-1.11	-1.85	-0.17	-0.64
0.3	-0.23	-0.86	-1.86	-3.17	-0.45	-1.70
0.5	-0.17	-0.66	-1.46	-2.53	-0.66	-2.53
0.7	-0.08	-0.30	-0.67	-1.19	-0.83	-3.19
0.9	-0.01	-0.04	-0.09	-0.16	-0.96	-3.74

TABLE 4.1 (Cont'd)

Service time distribution: H_2^{GN} , $c^2 = 2$.

ρ	$x = \frac{1}{2}\hat{\beta}$	$x = \hat{\beta}$	$x = \frac{3}{2}\hat{\beta}$	$x=2\hat{\beta}$	$x = \frac{1}{2} \frac{\hat{\beta}}{1 - \rho}$	$x = \frac{\hat{\beta}}{1 - \rho}$
0.1	-0.25	-0.83	-1.58	-2.45	-0.30	-0.98
0.3	-0.45	-1.51	-2.91	-4.56	-0.85	-2.69
0.5	-0.41	-1.38	-2.68	-4.21	-1.38	-4.21
0.7	-0.26	-0.87	-1.69	-2.64	-1.99	-5.66
0.9	-0.08	-0.27	-0.52	-0.79	-2.82	-7.15

Service time distribution: H_2^{GN} , $c^2 = 4$.

ρ	$x = \frac{1}{2}\hat{\beta}$	$x = \hat{\beta}$	$x = \frac{3}{2}\hat{\beta}$	$x=2\hat{\beta}$	$x = \frac{1}{2} \frac{\hat{\beta}}{1 - \rho}$	$x = \frac{\hat{\beta}}{1 - \rho}$
0.1	-0.34	-1.07	-1.98	-2.98	-0.41	-1.26
0.3	-0.64	-2.03	-3.79	-5.75	-1.17	-3.52
0.5	-0.61	-1.96	-3.68	-5.61	-1.96	-5.61
0.7	-0.41	-1.33	-2.50	-3.81	-2.93	-7.72
0.9	-0.14	-0.46	-0.86	-1.30	-4.33	-10.02

Service time distribution: H_2^{GN} , $c^2 = 6$.

ρ	$x = \frac{1}{2}\hat{\beta}$	$x = \hat{\beta}$	$x = \frac{3}{2}\hat{\beta}$	$x=2\hat{\beta}$	$x = \frac{1}{2} \frac{\hat{\beta}}{1 - \rho}$	$x = \frac{\hat{\beta}}{1 - \rho}$
0.1	-0.38	-1.18	-2.16	-3.23	-0.45	-1.39
0.3	-0.72	-2.26	-4.17	-6.29	-1.31	-3.88
0.5	-0.69	-2.21	-4.11	-6.22	-2.21	-6.21
0.7	-0.48	-1.53	-2.85	-4.32	-3.33	-8.63
0.9	-0.17	-0.54	-1.00	-1.52	-4.98	-11.29

Service time distribution: E_2 .

ρ	$x = \frac{1}{2}\hat{\beta}$	$x = \hat{\beta}$	$x = \frac{3}{2}\hat{\beta}$	$x=2\hat{\beta}$	$x = \frac{1}{2} \frac{\hat{\beta}}{1 - \rho}$	$x = \frac{\hat{\beta}}{1 - \rho}$
0.1	-0.03	-0.24	-0.68	-1.32	-0.05	-0.32
0.3	-0.00	-0.24	-0.86	-1.88	-0.06	-0.75
0.5	0.07	0.05	-0.26	-0.91	0.05	-0.91
0.7	0.11	0.27	0.33	0.24	0.32	-0.84
0.9	0.06	0.19	0.34	0.48	0.85	-0.53

TABLE 4.2. Approximation of $E\{S^2\}$. The table contains relative errors (%) of approximation APP1 (given by (4.21)) for various service time distributions.

Service time distribution: H_2^{BM} .

.

ρ	$c^2 = 2$	$c^2 = 4$	$c^2 = 6$
0.10	0.29	0.52	0.62
0.30	0.91	1.67	2.01
0.50	1.52	2.88	3.49
0.70	1.89	3.65	4.48
0.90	1.26	2.50	3.11
0.95	0.75	1.50	1.87

Service time distribution: H_2^{GN} .

ρ	$c^2=2$	$c^2 = 4$	$c^2 = 6$
0.10	-0.13	-0.22	-0.28
0.30	-0.29	-0.52	-0.61
0.50	-0.30	-0.54	-0.63
0.70	-0.17	-0.27	-0.31
0.90	0.00	0.05	0.07
0.95	0.01	0.05	0.08

Service time distributions with $c^2 < 1$.

ρ	E ₄	E_3	E_2	$C_{2}^{(1)}$	$C_{2}^{(2)}$
0.10	0.18	0.14	0.13	0.00	0.00
0.30	0.35	0.35	0.28	0.00	-0.02
0.50	0.38	0.38	0.32	-0.07	-0.09
0.70	0.25	0.24	0.21	-0.13	-0.11
0.90	0.05	-0.01	0.04	-0.11	-0.07
0.95	0.02	-0.01	0.02	-0.08	-0.04

 $C_2^{(1)}: c^2 = 0.75, C_2^{(2)}: c^2 = 0.92.$

TABLE 4.3. Approximation of $E\{\mathbf{S}^2\}$. The table contains relative errors (%) of approximation APP2 (given by (4.27) and (4.29)) for various service time distributions with $c^2 \ge 1$, $\hat{\beta}_3 \ge \frac{3}{2}\hat{\beta}_2^2/\hat{\beta}$.

Service time distribution: H_2^{BM} .

ρ	$c^2=2$	$c^2 = 4$	$c^2 = 6$
0.10	-0.15	-0.27	-0.32
0.30	-0.32	-0.58	-0.69
0.50	-0.31	-0.53	-0.61
0.70	-0.12	-0.09	-0.04
0.90	0.08	0.34	0.51
0.95	0.07	0.26	0.39

Service time distribution: H_2^{GN} .

ρ	$c^2 = 2$	$c^2 = 4$	$c^2 = 6$
0.10	-0.12	-0.21	-0.25
0.30	-0.23	-0.41	-0.48
0.50	-0.17	-0.30	-0.35
0.70	-0.01	0.01	0.03
0.90	0.09	0.20	0.26
0.95	0.07	0.14	0.18

TABLE 4.4. The influence of the third moment of the service time distribution $(\hat{\beta}_3)$ on $E\{\mathbf{S}^2\}$. In the table the exact values of $E\{\mathbf{S}^2\}$ are given. The relative approximation errors (%) of *APP*2 are indicated in parentheses below the exact values of $E\{\mathbf{S}^2\}$.

$\hat{oldsymbol{eta}}_3$	ρ=0.10	$\rho \!=\! 0.30$	ρ=0.50	ρ=0.70	$\rho \!=\! 0.90$	ρ=0.95
37.500*	6.498	12.00	26.67	85.47	909.1	3810
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
49.084	6.434	11.68	25.65	82.10	889.2	3762
	(-0.25)	(-0.50)	(-0.40)	(-0.01)	(0.26)	(0.19)
68.329	6.388	11.43	24.78	78.74	864.3	3698
	(-0.26)	(-0.58)	(-0.56)	(-0.15)	(0.34)	(0.29)
105.42	6.353	11.24	24.02	75.35	830.6	3609
	(-0.19)	(-0.47)	(-0.54)	(-0.28)	(0.24)	(0.26)
190.02	6.329	11.09	23.41	72.17	785.3	3441
	(-0.11)	(-0.29)	(-0.38)	(-0.31)	(0.01)	(0.08)
310.86	6.318	11.02	23.12	70.47	751.6	3295
	(-0.07)	(-0.19)	(-0.26)	(-0.25)	(-0.13)	(-0.09)
716.53	6.309	10.97	22.86	68.85	709.3	3063
	(-0.03)	(-0.08)	(-0.12)	(-0.14)	(-0.17)	(-0.22)
1391.8	6.306	10.95	22.77	68.21	689.0	2927
	(-0.02)	(-0.04)	(-0.06)	(-0.08)	(-0.13)	(-0.20)
2291.9	6.305	10.94	22.73	67.94	679.6	2856
	(-0.01)	(-0.03)	(-0.04)	(-0.05)	(-0.09)	(-0.16)
4541.9	6.304	10.93	22.70	67.74	671.9	2794
	(-0.00)	(-0.01)	(-0.02)	(-0.03)	(-0.05)	(-0.10)
∞	6.303	10.92	22.67	67.52	663.6	2724
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)

 H_2 service time distributions with $\hat{\beta}=1, c^2=4$.

*
$$\hat{\boldsymbol{\beta}}_3 = \frac{3}{2} \hat{\boldsymbol{\beta}}_2^2 / \hat{\boldsymbol{\beta}}$$

TABLE 4.4 (Cont'd)

.

 H_2 service time distributions with $\hat{\beta} = 1$, $c^2 = 10$.

$\hat{oldsymbol{eta}}_3$	$\rho = 0.10$	ρ=0.30	ρ=0.50	ρ=0.70	ρ=0.90	ρ=0.95	
181.50*	14.29	26.41	58.67	188.0	2000	8381	
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	
223.21	14.17	25.80	56.78	181.9	1965	8298	
	(-0.29)	(-0.50)	(-0.35)	(0.05)	(0.28)	(0.19)	
330.00	14.01	24.95	53.89	171.3	1891	8110	
	(-0.40)	(-0.80)	(-0.68)	(0.03)	(0.70)	(0.53)	
502.34	13.90	24.34	51.62	161.6	1802	7859	
	(-0.31)	(-0.73)	(-0.72)	(-0.04)	(1.00)	(0.86)	
710.16	13.84	24.00	50.26	155.0	1724	7612	
	(-0.24)	(-0.60)	(-0.65)	(-0.09)	(1.13)	(1.09)	
1323.2	13.78	23.61	48.65	146.4	1588	7093	
	(-0.14)	(-0.37)	(-0.44)	(-0.11)	(1.13)	(1.37)	
1845.6	13.76	23.49	48.11	143.2	1523	6797	
	(-0.10)	(-0.27)	(-0.34)	(-0.10)	(1.04)	(1.42)	
4162.2	13.73	23.31	47.31	138.4	1401	6129	
	(-0.05)	(-0.13)	(-0.17)	(-0.06)	(0.69)	(1.26)	
12263	13.72	23.22	46.89	135.7	1315	5543	
	(-0.02)	(-0.04)	(-0.06)	(-0.03)	(0.30)	(0.71)	
30488	13.71	23.19	46.76	134.8	1285	5305	
	(-0.01)	(-0.02)	(-0.02)	(-0.01)	(0.13)	(0.34)	
∞	13.71	23.17	46.67	134.2	1264	5124	
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	

$$* \hat{\beta}_3 = \frac{3}{2} \hat{\beta}_2^2 / \hat{\beta}$$

ρ	$H_{3}^{(1)}$	$H_{3}^{(2)}$	$C_2^{(1)}$	$C_2^{(2)}$	$C_{2}^{(3)}$		
0.10	-0.20	-0.29	-0.15	-0.30	-0.32		
0.30	-0.54	-0.71	-0.31	-0.63	-0.63		
0.50	-0.65	-0.82	-0.25	-0.54	-0.48		
0.70	-0.44	-0.48	-0.04	-0.04	0.05		
0.90	-0.00	0.20	0.12	0.40	0.39		
0.95	0.00	0.10	0.09	0.31	0.28		
$H_{3}^{(1)}: c^{2} = 2.778, \ \hat{\beta}_{3} = 40.963$ $H_{3}^{(2)}: c^{2} = 4.130, \ \hat{\beta}_{3} = 85.622$ $C_{2}^{(1)}: c^{2} = 2.200, \ \hat{\beta}_{3} = 18.240$ $C_{2}^{(2)}: c^{2} = 5.000, \ \hat{\beta}_{3} = 84.000$ $C_{3}^{(3)}: c^{2} = 8.556, \ \hat{\beta}_{2} = 187.33$							

TABLE 4.5. Relative approximation errors (%) of APP2 for H_3 and C_2 service time distributions.

. •

In all cases: $\hat{\beta} = 1$

TABLE 4.6. Relative approximation errors (%) of APP2 for various service time distributions with $c^2 \le 1$ and $\hat{\beta}_3 \le \frac{3}{2} \hat{\beta}_2^2 / \hat{\beta}$.

ρ	DET	E_4	E_2	C_2^*
0.10	-0.02	0.16	0.12	0.06
 0.30	-0.18	0.23	0.22	0.11
0.50	-0.36	0.16	0.18	0.10
0.70	-0.44	-0.02	0.05	0.04
0.90	-0.25	-0.10	-0.04	-0.02
0.95	-0.13	-0.08	-0.03	-0.02

* $c^2 = 0.75$

Chapter 5

QUEUEING MODELS WITH ADDITIONAL PERMANENT CUSTOMERS

5.1 INTRODUCTION

In the previous chapters we considered models with a single Poissonian external arrival stream. However, in many practical situations a service facility is shared by two or more classes of customers originating from different sources. An interesting aspect is the influence of the interference of the different customer streams on their queueing behaviour. A simple case occurs when it is assumed that the arrival processes are independent Poisson processes and the service discipline does not depend on the origin of the customers. Indeed, in that case the resulting 'overall' arrival process is also Poisson and hence we can use known results for the corresponding single arrival stream model.

In this chapter we consider single server queueing models with two classes of customers, viz. (i) ordinary customers who arrive according to a Poisson process, and (ii) permanent customers who immediately return to the end of the queue after having received a service. In Fig. 5.1 we have depicted the basic model: an M/G/1 FCFS queue with K permanent customers all having their own service time distribution $B_i(\cdot)$, i = 1, ..., K; the ordinary customers, in the sequel called 'Poisson customers', have service time distribution $B_0(\cdot)$.



Fig. 5.1 The M/G/1 queue with K additional permanent customers.

Note that the presence of permanent customers may represent the existence of an 'infinite customer pool' (or several infinite customer pools) of which only a fixed number (K) is allowed to be in the system at the same time; when one of

these customers departs upon service completion then a new customer from the pool immediately enters the queue. The main goal of this chapter is to study the influence of the K permanent customers on queue length and sojourn time of the Poisson customers.

Besides the basic model described above we shall also analyze some variants related with the models considered in the previous chapters, viz. the M/M/1 queue with general feedback and its limiting case, the M/G/1 processor sharing queue. The analysis of these models with additional permanent customers yields very interesting new results for the sojourn time distribution of the Poisson customers.

The M/G/1 queue with additional permanent customers is related with a class of models referred to as *vacation queues*. These are queueing models where a server now and then interrupts the service to a customer stream to take a vacation, see e.g. Fuhrmann and Cooper [1985] and the survey of Doshi [1986]. For the special case K = 1 our model behaves exactly like an M/G/1 queue with vacations and so called 'gated service'. This is an M/G/1 vacation queue in which, after a server vacation, the server handles exactly those customers who are present at the end of the vacation, etc.. Clearly, a service of the permanent customer in our model (with K = 1) corresponds to a server vacation in the gated vacation model.

The M/G/1 queue with vacations is a special case of a cyclic service model, a single server multi queue model in which the server attends to the queues in cyclic order, see e.g. Takagi [1986]. In Section 5.5 it will be pointed out that for general $K \ge 1$ the M/G/1 queue with K permanent customers can be viewed as a cyclic service model of a type not yet studied before.

Other related queueing models are the finite and infinite source interaction model studied by Kaufman [1985] (see also Boxma [1986A, 1986B] and Doshi and Wong [1987]), and a model with two stages of waiting introduced and analyzed in Ali and Neuts [1984].

The M/G/1 queue with K permanent customers is also studied by Boxma and Cohen [1989]. In particular, they present a fundamental analysis of the Markov chain of queue lengths at service completion epochs. In this chapter we present a different approach which is based on the observation that the queue length at an arbitrary epoch is the sum of K independent random variables which are related with queue lengths in the model with *one* permanent customer; the distributions of these random variables are obtained from known results for the M/G/1 queue with vacations.

As mentioned above we shall also consider the M/M/1 queue with general feedback and additional permanent customers. This is a very interesting model. Under the assumption that the service times of the permanent customers and the service times of the Poisson customers have the same exponential distribution, the addition of permanent customers to the M/M/1 feedback queue preserves the product form property of the joint queue length

distribution, cf. (2.3). Using this property we can derive the joint distribution of the successive sojourn times of a tagged Poisson customer. The analysis is largely analogous to the case without permanent customers, see Chapter 2. It appears that the queue length and sojourn time distributions become the (K+1)-fold convolution of the queue length and sojourn time distribution in the original system. Application of the limiting procedure described in Section 3.3 leads to similar results for the M/G/1 PS queue with K additional permanent customers. Note that, actually, the service discipline in the latter model is a special case of generalized processor sharing, i.e. an M/G/1 GPS queue with service rate function f(j) = 1/(j+K) (cf. Section 3.4).

The organization of this chapter is as follows. First, in Section 5.2, we give a detailed description of the basic model (the M/G/1 queue with K additional permanent customers) and introduce some notations. Section 5.3 is concerned with the derivation of mean queue lengths and sojourn times. We show that these quantities can be obtained from simple balance arguments. In Section 5.4 distributions of queue lengths and sojourn times are obtained. We start with the case of only one permanent customer (Subsection 5.4.1). The results for this case are obtained from known results for the M/G/1 queue with vacations. Next, in Subsection 5.4.2, the general model with $K \ge 1$ permanent customers is studied. In the remaining part of Section 5.4 we consider the results for some special choices of the service time distributions (Subsection 5.4.3), and we analyze a generalization, viz. the M/G/1 queue with Bernoulli feedback and additional permanent customers (Subsection 5.4.4). Section 5.5 is concerned with the relation with cyclic service models. Finally, in Section 5.6 we study the M/M/1 queue with general feedback and additional permanent customers together with the corresponding M/G/1 (G)PS queue.

5.2 MODEL DESCRIPTION AND NOTATIONS

We consider the single server queueing system with infinite waiting room pictured in Fig. 5.1. There are two classes of customers, viz. (i) ordinary customers who arrive according to a Poisson process with intensity λ , and (ii) a class of K permanent customers who immediately return to the end of the queue after having received a service. The service discipline is first come first served (FCFS). The order of the permanent customers in the system is fixed; they are numbered from 1 to K. The service times of the Poisson customers and of the permanent customers are assumed to be independent random variables; those of the Poisson customers all have the same distribution $B_0(\cdot)$ and the *i*-th permanent customer has service time distribution $B_i(\cdot)$, i = 1, ..., K. The first two moments of $B_i(\cdot)$ are denoted by β_i and $\beta_i^{(2)}$ respectively, i = 0, ..., K. $\beta_i(\cdot)$ denotes the Laplace-Stieltjes transform of $B_i(\cdot)$, i = 0, ..., K. Obviously, the total offered load to the queue per unit of time due to the Poisson customers, ρ_0 , is given by

$$\rho_0 = \lambda \beta_0. \tag{5.1}$$

93

For stability it is required that $\rho_0 < 1$. (For a formal derivation of this stability condition see Boxma and Cohen [1989]).

Observe that ρ_0 can also be viewed as the long range fraction of time spent on serving Poisson customers (as can be formally proved using the theory of regenerative processes, see Cohen [1976]). Similarly, we define for the permanent customers:

- ρ_i : fraction of time that (permanent) customer *i* is in service, i = 1, ..., K.

Noting that the total fraction of time spent on serving permanent customers is equal to $1-\rho_0$ it is easily seen that

$$\rho_i = \frac{\beta_i}{\overline{\beta}} (1 - \rho_0), \quad i = 1, ..., K,$$
(5.2)

with

$$\overline{\beta} := \sum_{i=1}^{K} \beta_{j}.$$

We shall use (5.2) in the next sections.

We are interested in the following steady-state quantities:

- X_0 : number of Poisson customers in the system at an arbitrary epoch;
- S_0 : sojourn time of a Poisson customer;
- C_i : sojourn (cycle) time of permanent customer *i*, *i* = 1,...,*K* (i.e. time between two successive service completions of customer *i*, *i* = 1,...,*K*).

The generating function of the distribution of X_0 is denoted by $\chi_0(\cdot)$. In the next section we shall derive the mean values of these performance measures; in Section 5.4 distributions are obtained.

5.3 Mean queue lengths and sojourn times

In this section we derive expressions for the *mean* sojourn times and queue lengths. It is shown how simple (balance) arguments can be used to derive these quantities. We start with the analysis of the mean sojourn time of the Poisson customers. Next, the mean cycle times of the permanent customers are derived.

Poisson customers

The PASTA property (see Wolff [1982]) implies that a newly arriving (tagged)

Poisson customer 'sees' the system as at an arbitrary epoch. Hence, the probability that the tagged customer arrives during the service of a Poisson customer is equal to ρ_0 ; the probability that permanent customer *i* is in service upon arrival of the tagged customer is equal to ρ_i , i = 1,...,K. Now, considering the mean amount of work in the system that has to be handled before the tagged customer receives his service it follows by a similar argument as used for the derivation of (2.46), that

$$ES_{0} = (EX_{0} - \rho_{0})\beta_{0} + \sum_{i=1}^{K} (1 - \rho_{i})\beta_{i} + \sum_{i=0}^{K} \rho_{i} \frac{\beta_{i}^{(2)}}{2\beta_{i}} + \beta_{0}.$$
 (5.3)

Using Little's formula,

$$E\mathbf{X}_0 = \lambda E\mathbf{S}_0,$$

it follows from (5.3) that

$$E\mathbf{S}_{0} = \frac{1}{1-\rho_{0}} \left[\sum_{i=0}^{K} (1-\rho_{i})\beta_{i} + \sum_{i=0}^{K} \rho_{i} \frac{\beta_{i}^{(2)}}{2\beta_{i}} \right].$$
(5.4)

Substituting (5.2) into (5.4) we find

$$E\mathbf{S}_{0} = \frac{\rho_{0}}{1-\rho_{0}}\frac{\beta_{0}^{(2)}}{2\beta_{0}} + \beta_{0} + \frac{\overline{\beta}}{1-\rho_{0}} + \sum_{i=1}^{K}\frac{\beta_{i}}{\overline{\beta}}(\frac{\beta_{i}^{(2)}}{2\beta_{i}} - \beta_{i}).$$
(5.5)

Again applying Little's formula yields the mean queue length EX_0 .

Note that the first two terms in the right-hand side of (5.5) represent the mean sojourn time in a standard M/G/1 queue (i.e. without permanent customers). The expression becomes very simple when all service times are exponentially distributed (i.e. $\beta_i^{(2)} = 2\beta_i^2$, i = 0, ..., K). In that case,

$$E\mathbf{S}_0 = \frac{\beta_0 + \overline{\beta}}{1 - \rho_0}.$$
(5.6)

Remark 5.1

It follows from (5.5) that the mean sojourn time ES_0 of the Poisson customers in the M/G/1 queue with K permanent customers is larger than the mean sojourn time ES_0^{super} of the Poisson customers in the same M/G/1 queue but with only *one* ('super') permanent customer who has a service time which is equal to the sum of the service times of the K permanent customers in the original model. Noting that the second moment of the service time of the super permanent customer is given by $\sum_{i=1}^{K} (\beta_i^{(2)} + 2\beta_i \sum_{j=i+1}^{K} \beta_j)$ it is easily found

that

$$E\mathbf{S}_0 - E\mathbf{S}_0^{super} = \sum_{i=1}^K rac{eta_i}{\overline{eta}} \sum_{j=1}^{i-1} eta_j \, .$$

We shall now consider the mean cycle times of the permanent customers.

Permanent customers

From the fact that the order of the permanent customers in the system is fixed it follows immediately that their mean cycle times are all equal:

$$E\mathbf{C}_i = E\mathbf{C}_j, \quad 1 \le i, j \le K. \tag{5.7}$$

It is easily seen that, in steady-state, the mean amount of work that *arrives* during a cycle of customer *i* is equal to the mean amount of work that is *served* during a cycle. This balance argument leads to the following equation for EC_i :

$$E\mathbf{C}_i = \boldsymbol{\beta} + (\lambda E \mathbf{C}_i)\boldsymbol{\beta}_0, \quad i = 1, \dots, K,$$
(5.8)

yielding

$$E\mathbf{C}_{i} = \frac{\overline{\beta}}{1-\rho_{0}}, \quad i = 1, \dots, K.$$
(5.9)

Note that in the right-hand side only the first moments of the various service time distributions do occur. Apparently, this is due to the fact that the successive cycle times of a permanent customer consist of a random sum of *complete* service times.

There are some other simple and interesting methods to derive formula (5.9). Here, we mention two of these methods.

- (i) From the definition of ρ_i it follows that ρ_i/β_i equals the throughput of customer i. Hence, from Little's formula: EC_i=1/(ρ_i/β_i), i=1,...,K. Substituting (5.2) into this expression yields (5.9).
- (ii) The service of a permanent customer induces an amount of work to be handled by the server which consists of the permanent customer's own service time plus the sum of the lengths of a (random) number of standard M/G/1 (λ, B₀(·)) busy periods. This number is equal to the number of Poisson customers who arrive during the service time of the permanent customer. So, the mean amount of work induced by a service of customer j is given by β_j +λβ_j(β₀/(1-ρ₀)), j = 1,...,K. Noting that during a cycle of customer i all permanent customers receive exactly one service it follows from a balance argument that, cf. (5.9),

$$E\mathbf{C}_i = \sum_{j=1}^{K} [\beta_j + \lambda \beta_j (\beta_0 / (1 - \rho_0))] = \frac{\overline{\beta}}{1 - \rho_0}, \quad i = 1, \dots, K.$$

The methods used in this section for the derivation of mean sojourn (cycle) times are based on mean value analysis. The derivation of queue length and sojourn time distributions requires a more detailed study of the model. This will be presented in the next section. It will also give us more insight into the results obtained above.

5.4 DISTRIBUTIONS OF QUEUE LENGTHS AND SOJOURN TIMES

In this section we derive expressions for the generating functions and Laplace-Stieltjes transforms of the distributions of the queue lengths and the sojourn (cycle) times of the different customers in the model. We shall start in Subsection 5.4.1 with the case of only one permanent customer (K=1); in Subsection 5.4.2 the general case $(K \ge 1)$ is considered. A close study of the behaviour of the system with K permanent customers will show that its queue length at an arbitrary epoch can be written as the sum of K independent random variables, which are related with queue lengths in the model with *one* permanent customer. The distributions of these random variables can be obtained from the results in Subsection 5.4.1. In Subsection 5.4.3 we shall consider the results for some special choices of the service time distributions. Finally, Subsection 5.4.4 is concerned with the analysis of the M/G/1 queue with Bernoulli feedback and additional permanent customers; it appears that this generalization can be analyzed completely analogously to the basic model.

5.4.1 The case K = 1

In this subsection we consider the M/G/1 queue with one permanent customer. As pointed out in Section 5.1 this special case behaves exactly like an M/G/1 queue with vacations and gated service. A service of the permanent customer corresponds to a server vacation in the vacation model. For the analysis we shall use the following *decomposition* result for the distribution of the queue length in a vacation queue. Define for the M/G/1 queue with vacations,

- $X(\cdot)$: generating function of the distribution of the queue length at an arbitrary epoch,
- $\chi_{V}(\cdot)$: generating function of the distribution of the queue length at an arbitrary epoch given that the server is on vacation,

and let

- $\pi(\cdot)$: generating function of the distribution of the queue length at an arbitrary epoch in the corresponding standard M/G/1 queue (i.e. the same model without vacations).

Then,

THEOREM 5.1 (Fuhrmann and Cooper [1985])

$$\chi(z) = \chi_V(z)\pi(z), \quad |z| \le 1.$$
(5.10)

Actually, this result does not only hold for the M/G/1 vacation queue with *gated* service; it is valid for a very general class of vacation models, see Fuhrmann and Cooper [1985].

Now, we return to the M/G/1 queue with one permanent customer. Define

- X_P : number of Poisson customers in the system at an arbitrary epoch given that the permanent customer is in service,

and let $\chi_P(\cdot)$ denote the generating function of the distribution of X_P . From (5.10) it follows immediately that the generating function $\chi_0(\cdot)$ of the distribution of the queue length at an arbitrary epoch can be written as

$$\chi_0(z) = \chi_P(z)\pi(z), |z| \le 1.$$
 (5.11)

The generating function of the queue length distribution in the standard M/G/1 queue, $\pi(\cdot)$, is given by the well-known formula (see e.g. Cohen [1982], p. 238):

$$\pi(z) = (1 - \rho_0) \frac{(1 - z)\beta_0\{\lambda(1 - z)\}}{\beta_0\{\lambda(1 - z)\} - z}, \quad |z| \le 1.$$
(5.12)

For the derivation of $\chi_P(z)$ we need the following definitions:

- X_{PB} : number of Poisson customers in the system just after the start of a service of the permanent customer;
- X_{PE} : number of Poisson customers in the system at a service completion epoch of the permanent customer.

 $\chi_{PB}(\cdot)$ and $\chi_{PE}(\cdot)$ will denote the generating functions of the distributions of X_{PB} and X_{PE} , respectively.

The number of Poisson customers present at an arbitrary epoch during the service of the permanent customer is equal to the number of Poisson customers present at the start of that service *plus* the number of Poisson customers that has arrived during the past service time. It is easily seen that these quantities are independent. Hence,

$$\chi_{P}(z) = \chi_{PB}(z) \frac{1 - \beta_{1}\{\lambda(1-z)\}}{\beta_{1}\lambda(1-z)}, \quad |z| \le 1.$$
(5.13)

(Remember that, for Re $\eta \ge 0$, $\frac{1-\beta_1(\eta)}{\beta_1\eta}$ is the LST of the distribution of the past part of the service time.)

Analogously, we have

$$\chi_{PE}(z) = \chi_{PB}(z)\beta_1\{\lambda(1-z)\}, \quad |z| \le 1.$$
(5.14)

Now, from (5.11)-(5.14) it follows that

$$\chi_0(z) = \frac{\chi_{PE}(z)}{\beta_1\{\lambda(1-z)\}} \frac{1 - \beta_1\{\lambda(1-z)\}}{\beta_1\lambda(1-z)} \, \pi(z), \quad |z| \le 1.$$
(5.15)

It remains to determine $\chi_{PE}(\cdot)$.

The discrete time stochastic process constituted by the number of Poisson customers present in the system at the successive service completion epochs of the permanent customer is easily seen to be a Markov chain with state space $\{0, 1, \ldots\}$ and stationary transition probabilities p_{ij} given by

$$p_{ij} = \int_{t=0}^{\infty} e^{-\lambda t} \frac{(\lambda t)^j}{j!} d[B_0^{i^*}(t)^* B_1(t)], \quad i,j \ge 0,$$
(5.16)

with * the convolution operator. Application of standard Markov chain theory leads to the following functional equation for $\chi_{PE}(\cdot)$:

$$\chi_{PE}(z) = \beta_1 \{ \lambda(1-z) \} \chi_{PE}(\beta_0 \{ \lambda(1-z) \}), \quad |z| \le 1.$$
(5.17)

From this equation the moments of X_{PE} can be obtained. For example, differentiating both sides once and taking z = 1 yields,

$$E\mathbf{X}_{PE} = \frac{\lambda\beta_1}{1-\rho_0}.$$
(5.18)

Analogously, the moments of the distribution of X_0 can be obtained from (5.15) and (5.17). It is easily found that, for the case K = 1,

$$E\mathbf{X}_{0} = \frac{\lambda\beta_{1}}{1-\rho_{0}} - \lambda\beta_{1} + \lambda\frac{\beta_{1}^{(2)}}{2\beta_{1}} + \frac{\lambda\rho_{0}}{1-\rho_{0}}\frac{\beta_{0}^{(2)}}{2\beta_{0}} + \rho_{0}.$$
 (5.19)

This is in agreement with (5.5) for K = 1 (noting that $EX_0 = \lambda ES_0$).

Remark 5.2

The generating function of the distribution of the queue length at the *departure* epochs of the Poisson customers is also given by (5.15). Indeed, an up-and-down-crossing argument shows that this distribution equals the queue length distribution seen by an *arriving* Poisson customer; the PASTA property implies that this is also the distribution of the queue length at an *arbitrary* epoch.

Remark 5.3

In Boxma and Cohen [1989] it is shown that the unique (non trivial) solution of (5.17) is given by:

$$\chi_{PE}(z) = \prod_{h=0}^{\infty} \beta_1 \{ \lambda (1 - \delta_h(z)) \}, \quad |z| \le 1,$$
(5.20)

with

$$\delta_0(z) := z,$$

$$\delta_{h+1}(z) := \beta_0 \{ \lambda(1 - \delta_h(z)) \}, \quad h = 0, 1, \dots.$$
(5.21)

The derivation is based on iteration of (5.17) and on some well-known results from branching theory; the condition $\rho_0 < 1$ guarantees the convergence of the infinite product in (5.20).

The LST's of the sojourn time of the Poisson customers and the cycle time of the permanent customer can be derived from the above queue length results. Noting that the cycle time of the permanent customer consists of his own service time plus the service times of the Poisson customers present in the system just after the end of his previous service, we have,

$$E\{e^{-\eta C_1}\} = \chi_{PE}(\beta_0(\eta))\beta_1(\eta), \quad \text{Re } \eta \ge 0, \tag{5.22}$$

with $\chi_{PE}(\cdot)$ determined by (5.17). The LST of the sojourn time distribution of the Poisson customers can be derived from (5.15). The classical observation for FCFS queues that the number of Poisson customers left behind by a departing customer equals the number of (Poisson) arrivals during the sojourn time of that departing customer, leads to (cf. Remark 5.2)

$$E\{e^{-\lambda(1-z)\mathbf{S}_0}\} = \boldsymbol{\chi}_0(z), \quad |z| \leq 1.$$

Hence,

$$E\{e^{-\eta \mathbf{S}_0}\} = \chi_0(1-\eta/\lambda), \quad \text{Re } \eta \ge 0.$$
(5.23)

From the above results the moments of the distributions of S_0 and C_1 can be obtained. Differentiating (5.22) and (5.23) once with respect to η and taking $\eta=0$ yields the mean cycle time and sojourn time (use (5.18) and (5.19)); it is easily verified that these results coincide with the results obtained in Section 5.3.

We conclude this subsection with a remark on notation. For the analysis in the next subsection it is convenient to have at our disposal a specific notation for some of the quantities in the model with one permanent customer. Therefore, for the M/G/1 queue with one permanent customer and service time distributions $B_0(\cdot)$ and $B_i(\cdot)$ for the Poisson customers and the permanent customer respectively we define, for i = 1, ..., K,

- $\chi_0^{(1,i)}(\cdot)$: generating function of the distribution of the number of Poisson customers in the system at an arbitrary epoch;
- $\chi_{PE}^{(1,i)}(\cdot)$: generating function of the distribution of the number of Poisson customers in the system at a service completion epoch of the permanent customer.

Obviously, these quantities can be obtained from (5.15) and (5.17) by taking $B_1(\cdot) \equiv B_i(\cdot), i = 1, ..., K$.

5.4.2 The case $K \ge 1$

We now turn to the case of an arbitrary number $K \ge 1$ of permanent customers. In Fig. 5.2(a) we have pictured the composition of the queue just after a service completion epoch of permanent customer i-1. It can be described as follows: counted from the head of the queue there is a group of Poisson customers followed by permanent customer i; subsequently there is a second group of Poisson customers followed by permanent customer i + 1, etc., until finally the queue is ended by a K-th group of Poisson customers followed by permanent customer i-1 (who has just returned from the head of the queue). The group of customers consisting of the Poisson customers at the head of the queue and permanent customer i will be called 'group i'; the group of customers consisting of permanent customer j and the Poisson customers ahead of jand behind j-1 will be called 'group j', j=1,...,K, $j\neq i$. (Note that during a service time the last permanent customer in the queue may be followed by one or more Poisson customers; in that case these Poisson customers are assumed to be members of the group at the head of the queue, cf. Fig. 5.2(b)). The (random) number of Poisson customers in a particular group will be referred to as the 'size' of that group. So, the total number of Poisson customers in the system is equal to the sum of the sizes of the K groups. To determine the distribution of the size of each group we shall first investigate how these groups develop. Let us start with the present situation, i.e. just after a service completion of permanent customer i-1. Now, the first customer of group i is taken into service; next the second one, etc, until finally permanent customer i










Fig. 5.2 Composition of the queue

- (a) just after the service completion of permanent customer i-1,
- (b) during the service of group i,
- (c) just after the service completion of permanent customer *i*.

completes his service. During these services group *i* consists of two parts, see Fig. 5.2(b): (i) permanent customer i together with the ('old') Poisson customers in front of him who have not yet been served to completion, and (ii) Poisson customers at the end of the queue (behind permanent customer i-1) who have newly arrived during the past service time of group *i*. Just after the service completion of permanent customer i this second part has developed into a complete, 'new', group i at the end of the queue, see Fig. 5.2(c); its size is equal to the number of Poisson customers who have arrived during the service of the 'old' group *i*. Note that the size of the other groups has not changed. After the service completion of group *i* the service of group i + 1 is started, next the service of group i+2, etc. During the service of groups $i+1, \ldots, i-1$ group i moves to the head of the queue, while its size remains unchanged. Just after the service of permanent customer i-1 the order of the different groups within the queue is again as in Fig. 5.2(a). Now, a second service of group i is started, and the whole procedure as described above is repeated, etc.

From the above discussion and from the memoryless property of the Poisson arrival process it is clear that during the service of group *i* this group behaves exactly like an M/G/1 queue with one permanent customer and service time distributions $B_0(\cdot)$ and $B_i(\cdot)$ for the Poisson customers and the permanent customer respectively, i = 1,...,K. It is also seen that at a service completion epoch of one of the permanent customers the sizes of the groups are independent; this independence property also holds at an arbitrary epoch given that group *i* is in service, i = 1,...,K. Now define, for i = 1,...,K,

- $\chi_0^{(i)}(\cdot)$: generating function of the distribution of the number of Poisson customers present in the system at an arbitrary epoch given that group *i* is in service;
- $\chi_{PE}^{(i)}(\cdot)$: generating function of the distribution of the number of Poisson customers present in the system at a service completion epoch of permanent customer *i*.

It is easily seen that

$$\chi_{PE}^{(i)}(z) = \prod_{j=1}^{K} \chi_{PE}^{(1,j)}(z) , \quad i = 1, ..., K, \quad |z| \le 1,$$
(5.24)

and

$$\chi_{0}^{(i)}(z) = \chi_{0}^{(1,i)}(z) \prod_{\substack{j=1\\j \neq i}}^{K} \chi_{PE}^{(1,j)}(z) , \quad i = 1, ..., K, \quad |z| \le 1,$$
(5.25)

102 '

where the generating functions $\chi_{PE}^{(1,j)}(\cdot)$ and $\chi_0^{(1,j)}(\cdot)$ of the queue length distributions in the model with one permanent customer are given by (5.20) and (5.15) respectively (see also the definitions at the end of Subsection 5.4.1). Substituting (5.15) into (5.25) yields

$$\chi_{0}^{(i)}(z) = \frac{1}{\beta_{i}\{\lambda(1-z)\}} \frac{1 - \beta_{i}\{\lambda(1-z)\}}{\beta_{i}\lambda(1-z)} \pi(z) \prod_{j=1}^{K} \chi_{PE}^{(1,j)}(z), \qquad (5.26)$$
$$i = 1, ..., K, \quad |z| \le 1.$$

Obviously, the probability p_i that group *i* is in service at an arbitrary epoch is equal to the long range fraction of time that group *i* is in service. Noting that the mean time spent on serving group *i* during a cycle is equal to $\beta_i/(1-\rho_0)$ (cf. the derivation of (5.24) and (5.18)) we have from (5.9),

$$p_{i} = \frac{\beta_{i}/(1-\rho_{0})}{E\mathbf{C}_{i}} = \frac{\beta_{i}/(1-\rho_{0})}{\overline{\beta}/(1-\rho_{0})} = \frac{\beta_{i}}{\overline{\beta}}, \quad i = 1, ..., K.$$
(5.27)

Hence, for $K \ge 1$,

$$\begin{aligned} \chi_{0}(z) &= \sum_{i=1}^{K} p_{i} \chi_{0}^{(i)}(z) = \\ \pi(z) \left\{ \prod_{j=1}^{K} \chi_{PE}^{(1,j)}(z) \right\} \sum_{i=1}^{K} \frac{\beta_{i}}{\overline{\beta}} \frac{1}{\beta_{i} \{\lambda(1-z)\}} \frac{1 - \beta_{i} \{\lambda(1-z)\}}{\beta_{i} \lambda(1-z)}, \quad i = 1, ..., K, \quad |z| \le 1. \end{aligned}$$
(5.28)

Remember that $\pi(z)$ represents the generating function of the queue length distribution in the M/G/1 queue without permanent customers, cf. (5.12). Note that it is not allowed to take K=0 in (5.28); in its derivation K is explicitly assumed to be positive.

The LST's of the sojourn time and cycle time distributions can be easily obtained from the above queue length results. Analogously to the derivations of the sojourn time and cycle time in the model with one permanent customer we have, from (5.24) (cf. (5.22)),

$$E\{e^{-\eta \mathbf{C}_{i}}\} = \chi_{PE}^{(i)}(\beta_{0}(\eta)) \prod_{j=1}^{K} \beta_{j}(\eta) = \prod_{j=1}^{K} \{\beta_{j}(\eta) \chi_{PE}^{(1,j)}(\beta_{0}(\eta))\},$$
(5.29)

i=1,...,K, Re $\eta \ge 0$,

and, from (5.28) (cf. (5.23) and Remark 5.2),

$$E\{e^{-\eta \mathbf{S}_0}\} = \chi_0(1-\eta/\lambda) =$$

$$\pi(1-\eta/\lambda)(\prod_{j=1}^K \chi_{PE}^{(1,j)}(1-\eta/\lambda))\sum_{i=1}^K \frac{\beta_i}{\overline{\beta}} \frac{1}{\beta_i(\eta)} \frac{1-\beta_i(\eta)}{\beta_i \eta}, \quad \text{Re } \eta \ge 0.$$
(5.30)

Differentiating these expressions once with respect to η and taking $\eta=0$ yields the mean cycle time of the permanent customers and the mean sojourn time of the Poisson customers; it is easily verified that these results coincide with (5.9) and (5.5).

5.4.3 Results for some special choices of the service time distributions For some special choices of the different service time distributions in the model the queue length and sojourn time formulas (5.24) and (5.28)-(5.30) reduce to much simpler expressions. The following two cases are worth mentioning.

(i) Equal service time distributions for the permanent customers If $B_1(\cdot) \equiv \cdots \equiv B_K(\cdot)$, then, from (5.24),

$$\chi_{PE}^{(i)}(z) = (\chi_{PE}(z))^{K}, \quad i = 1, ..., K, \quad |z| \le 1,$$
(5.31)

and, from (5.28),

$$\chi_0(z) = \pi(z) \frac{(\chi_{PE}(z))^K}{\beta_1\{\lambda(1-z)\}} \frac{1 - \beta_1\{\lambda(1-z)\}}{\beta_1\lambda(1-z)}, \quad |z| \le 1,$$
(5.32)

with $\chi_{PE}(\cdot)$ determined by (5.20).

(ii) All service time distributions equal and negative exponential If $B_i(t) = 1 - e^{-t/\beta_0}$, i = 0, 1, ..., K, then the solution of the functional equation (5.17) has an explicit form (cf. Remark 5.3, for the solution of the general case). Defining, for $|z| \le 1$, n = 1, 2, ...,

$$f^{(1)}(z) := f(z) := \beta_0 \{ \lambda(1-z) \},$$

$$f^{(n+1)}(z) := f(f^{(n)}(z)),$$
(5.33)

one can iterate (5.17) in the following way:

$$\chi_{PE}(z) = f^{(1)}(z)\chi_{PE}(f^{(1)}(z)) = f^{(1)}(z)f^{(2)}(z)\chi_{PE}(f^{(2)}(z)) = \dots = (5.34)$$
$$(\prod_{j=1}^{m} f^{(j)}(z))\chi_{PE}(f^{(m)}(z)).$$

For our case of exponentially distributed service times, i.e. (cf. (5.33))

$$f(z) = \frac{1}{1+\beta_0\lambda(1-z)} = \frac{1}{1+\rho_0(1-z)},$$

it is easily found that

. •

$$\prod_{j=1}^{m} f^{(j)}(z) = \frac{1}{1 + (1-z) \sum_{h=1}^{m} \rho_0^h} = \frac{1}{1 + (1-z)(\frac{1-\rho_0^{m+1}}{1-\rho_0} - 1)},$$
(5.35)

and hence,

$$f^{(m)}(z) \to 1, \quad \text{for } m \to \infty.$$
 (5.36)

Now, from (5.34)-(5.36) we obtain (cf. (5.20))

$$\chi_{PE}(z) = \lim_{m \to \infty} (\prod_{j=1}^{m} f^{(j)}(z)) \chi_{PE}(f^{(m)}(z)) = \frac{1 - \rho_0}{1 - \rho_0 z},$$
(5.37)

which equals the generating function of the queue length distribution in an ordinary M/M/1 queue. Substituting (5.37) into (5.31) and (5.32) we find, for $|z| \leq 1$,

$$\chi_{PE}^{(i)}(z) = \left(\frac{1-\rho_0}{1-\rho_0 z}\right)^K, \quad i = 1, ..., K,$$
(5.38)

$$\chi_0(z) = \left(\frac{1-\rho_0}{1-\rho_0 z}\right)^{K+1}.$$
(5.39)

Formula (5.39) exposes a remarkable effect of the presence of permanent customers on the queue length of the Poisson customers in an M/M/1 queue: their queue length distribution becomes the (K+1)-fold convolution of the queue length distribution in the system without those permanent customers.

From (5.29), (5.30), (5.38) and (5.39) the LST's of the cycle time and sojourn time distribution are obtained: for Re $\eta \ge 0$,

$$E\{e^{-\eta C_i}\} = \left(\frac{1-\rho_0}{1-\rho_0+\beta_0\eta}\right)^K, \quad i=1,...,K,$$
(5.40)

$$E\{e^{-\eta \mathbf{S}_0}\} = \left[\frac{1-\rho_0}{1-\rho_0+\beta_0\eta}\right]^{K+1}.$$
 (5.41)

So, the presence of K permanent customers also leads to a (K+1)-fold increase of the sojourn times of the Poisson customers.

Remark 5.4

The above mentioned phenomenon (expressed by (5.41)) for the case of identical, exponential service times can be explained as follows. Consider the model with one permanent customer, i.e. the case K=1 (the general case follows immediately from this special one). It is seen from Formula (5.40) that the successive cycle times of the permanent customer are exponentially distributed with mean $\beta_0/(1-\rho_0)$. A newly arriving (tagged) Poisson customer enters the queue during one of these cycles; it is clear that his sojourn time, S_0 , is equal to the residual cycle time $(\mathbf{C}_1^{(R)})$ plus the sum of the service times of the customers who have arrived during the past cycle time (V) plus his own service time (τ_0) : $\mathbf{S}_0 = \mathbf{C}_1^{(R)} + \mathbf{V} + \tau_0$. From standard probabilistic arguments it follows that the residual and past cycle time are exponentially distributed with mean $\beta_0/(1-\rho_0)$, and it is found that the number of customers who have arrived during the past cycle time has a geometric distribution with mean $\rho_0/(1-\rho_0)$. Moreover, $\mathbf{C}_1^{(R)}$, \mathbf{V} and $\boldsymbol{\tau}_0$ are mutually independent. Now it is easily seen that $V + \tau_0$ is exponentially distributed with mean $\beta_0 / (1 - \rho_0)$ and, hence, S_0 has a 2-stage Erlang distribution (E_2) with mean $2\beta_0/(1-\rho_0)$, cf. (5.41).

Remark 5.5

Note that in the present exponential (product form) case a departing (and hence again arriving) permanent customer sees the system in equilibrium with one less customer of his own type (see e.g. Walrand [1988, Section 3.4]), which confirms the relation between (5.38) and (5.39), and between (5.40) and (5.41).

In Section 5.6 it will be shown that generalizations of (5.38)-(5.41) hold for the M/M/1 queue with general feedback and additional permanent customers.

5.4.4 Generalizations; the M/G/1 queue with Bernoulli feedback and additional permanent customers

The results obtained in the previous subsections can be generalized in several

ways. Firstly one may allow different arrival rates during the service of the permanent customers. Another interesting possibility is the inclusion of a (Bernoulli) feedback mechanism for the Poisson customers. For both generalizations the basic decomposition formula (5.11), which is implied by Theorem 5.1, remains valid (see Fuhrmann and Cooper [1985] and Shanthikumar [1988]), and the analysis can be carried out in almost exactly the same way as in Subsection 5.4.1 and Subsection 5.4.2. We shall consider the feedback case in some more detail.

The M/G/1 queue with Bernoulli feedback and additional permanent customers Consider the M/G/1 queue with K permanent customers described in Section 5.2 and assume in addition that the Poisson customers, after having received a service, are fed back to the end of the queue with probability p and leave the system with probability 1-p.

As for the model without feedback we first analyze the case K=1. Clearly, this case behaves exactly like an M/G/1 Bernoulli feedback queue with vacations and gated service. In Fuhrmann and Cooper [1985] it is pointed out that for this generalization of the 'standard' M/G/1 vacation queue (without feedback) Theorem 5.1 remains valid with, in (5.10), $\pi(\cdot)$ denoting the generating function of the queue length distribution in the corresponding M/G/1 queue with Bernoulli feedback but without vacations. It is well-known that $\pi(\cdot)$ is given by: (see e.g. Takács [1963])

$$\pi(z) = (1 - \hat{\rho}_0) \frac{(1 - z)(1 - p)\beta_0\{\lambda(1 - z)\}}{(1 - p + pz)\beta_0\{\lambda(1 - z)\} - z},$$
(5.42)

provided that

$$\hat{\rho}_0 := \frac{\rho_0}{1-p} < 1. \tag{5.43}$$

(Note that $\pi(\cdot)$ equals the generating function of the queue length distribution in the standard M/G/1 queue with service time distribution $\hat{B}_0(t) = \sum_{j=1}^{\infty} (1-p)p^{j-1}B_0^{i^*}(t)$, cf. (5.12)). Now, it is easily seen that for the present model with Bernoulli feedback the rest of the analysis in Subsection 5.4.1 does not change with the exception that the functional equation (5.17) for the generating function $\chi_{PE}(\cdot)$ of the queue length distribution just after a service completion of the permanent customer becomes

$$\chi_{PE}(z) = \beta_1\{\lambda(1-z)\}\chi_{PE}((1-p+pz)\beta_0\{\lambda(1-z)\}), \quad |z| \le 1.$$
 (5.44)

With this adaptation Formula (5.15) for the generating function $\chi_0(\cdot)$ of the queue length distribution at an arbitrary epoch remains valid. Now, from

equation (5.44), (cf. (5.18))

$$E\mathbf{X}_{PE} = \frac{\lambda\beta_1/(1-p)}{1-\hat{\rho}_0}, \qquad (5.45)$$

and hence, from (5.15), (5.42) and (5.45) we have, for the case K = 1, (cf. (5.19))

$$E\mathbf{X}_{0} = \frac{\lambda\beta_{1}/(1-p)}{1-\hat{\rho}_{0}} - \lambda\beta_{1} + \lambda\frac{\beta_{1}^{(2)}}{2\beta_{1}} + \frac{\lambda\hat{\rho}_{0}}{1-\hat{\rho}_{0}}\left[p\frac{\beta_{0}}{1-p} + \frac{\beta_{0}^{(2)}}{2\beta_{0}}\right] + \hat{\rho}_{0}.$$
 (5.46)

Remark 5.6

It follows from (5.45) that the mean cycle time, EC_1 , of the permanent customer is given by $EC_1 = \beta_0 E \mathbf{X}_{PE} + \beta_1 = \beta_1 / (1 - \hat{\rho}_0)$. As might be expected, for $\beta_0 = \beta_1 = \beta$ this result coincides with (2.66) which gives the mean *k*-th sojourn time of a tagged customer in the M/G/1 queue with Bernoulli feedback for $k \rightarrow \infty$.

It is easily seen that, using the above formulas for the case K=1, the analysis for the case $K \ge 1$ can be carried out in exactly the same way as for the model without feedback and that all queue length formulas in Subsection 5.4.2 remain unchanged. For example, from the generating function of the queue length distribution given by (5.28) and from (5.45), (5.46) we obtain for general $K \ge 1$,

$$E\mathbf{X}_{0} = \frac{\lambda \overline{\beta} / (1-p)}{1-\hat{\rho}_{0}} + \lambda \sum_{i=1}^{K} \frac{\beta_{i}}{\overline{\beta}} (\frac{\beta_{i}^{(2)}}{2\beta_{i}} - \beta_{i}) + \frac{\lambda \hat{\rho}_{0}}{1-\hat{\rho}_{0}} [p \frac{\beta_{0}}{1-p} + \frac{\beta_{0}^{(2)}}{2\beta_{0}}] + \hat{\rho}_{0}.$$
(5.47)

Hence, using Little's formula, the mean *total* sojourn time of the Poisson customers is given by: (cf. (5.5))

$$ES_{0} = \frac{\overline{\beta}/(1-p)}{1-\hat{\rho}_{0}} + \sum_{i=1}^{K} \frac{\beta_{i}}{\overline{\beta}} (\frac{\beta_{i}^{(2)}}{2\beta_{i}} - \beta_{i}) + \frac{\hat{\rho}_{0}}{1-\hat{\rho}_{0}} [p\frac{\beta_{0}}{1-p} + \frac{\beta_{0}^{(2)}}{2\beta_{0}}] + \frac{\beta_{0}}{1-p}.$$
(5.48)

Remark 5.7

Having obtained the generating function of the distribution of the queue length at an arbitrary epoch we can also derive the LST of the joint distribution of the successive sojourn times of a tagged Poisson customer. The analysis can be done almost completely analogously to the case without permanent customers as treated by Doshi and Kaufmann [1988]. We shall only consider the case of negative exponential service times in some detail; in Section 5.6 it is shown how the LST of the joint sojourn time distribution in the M/M/1 queue with general feedback and additional permanent customers can be obtained from the corresponding result derived in Chapter 2 for the same model without permanent customers.

5.5 Relation with cyclic service models

In Section 5.1 it has been pointed out that the M/G/1 queue with *one* permanent customer behaves exactly like an M/G/1 queue with vacations and gated service. In this section we shall show that the M/G/1 queue with $K \ge 1$ permanent customers can be viewed as a variant of a *cyclic service model*.

A cyclic service model (also called a polling model) is a single server multi queue model in which the server attends to the queues in cyclic order, see e.g. Takagi [1986] and Groenendijk [1990]. From the analysis in Subsection 5.4.2 it follows that the M/G/1 queue with K permanent customers can also be viewed as a cyclic service model with K queues-albeit a rather special one, see Fig. 5.3.



Fig. 5.3 The M/G/1 queue with permanent customers viewed as a cyclic service model.

The service times of the permanent customers correspond to the switch-over times of the server between successive queues; the customers in these queues represent the (Poisson) customers in the different groups in the single queue M/G/1 model (cf. Fig. 5.2). To take into account that during the service of a particular group all new arrivals are attached to that group (cf. Fig. 5.2(b)), we have to assume that arrivals at a queue can only take place during its service and during the subsequent switch-over time.

The resulting polling model is non-standard, but it is well-known that under very general conditions the mean cycle time in a polling model is given by the sum of the mean switch-over times, divided by one minus the load of the system (see e.g. Takagi [1986]). Indeed this result holds here; it coincides with Formula (5.9) for the mean cycle time of a permanent customer.

5.6 The M/M/1 FEEDBACK QUEUE WITH ADDITIONAL PERMANENT CUSTOMERS 5.6.1 Introduction

In this section we consider the same M/M/1 queue with general feedback as in Chapter 2 but with $K \ge 1$ additional permanent customers. This model is pictured in Fig. 5.4.



Fig. 5.4 The M/M/1 feedback queue with K additional permanent customers.

It is assumed that the service times of the Poisson customers and the permanent customers are independent, negative exponentially distributed random variables, all with mean β . For the Poisson customers the assumptions about the feedback mechanism, notations, terminology, etc. are the same as for the model without permanent customers, see Section 2.2. Our main goal is to study the influence of the presence of the permanent customers on the joint distribution of the successive sojourn times of a (tagged) Poisson customer and to use the results for the analysis of the sojourn time in the M/G/1 PS queue with additional permanent customers. The results for this latter model are obtained by applying the same limiting procedure as used for the case without permanent customers, see Chapter 3.

Because the Poisson customers and the permanent customers have the same

exponential service time distribution, the joint stationary distribution of the number of type-*i* (Poisson) customers, X_i , i = 1, ..., N, in the system at an arbitrary epoch is of product form type. From the queue length results for general product form networks (see Baskett et al. [1975]) it is found that for our model, cf. (2.3),

$$P(x_{1},...,x_{N}) := Pr\{\mathbf{X}_{1} = x_{1},...,\mathbf{X}_{N} = x_{N}\} = (5.49)$$

$$(1-\rho)^{K+1} \frac{(x_{1} + \cdots + x_{N} + K)!}{x_{1}! \cdots x_{N}!K!} \prod_{i=1}^{N} (\lambda \beta q(i))^{x_{i}}, \quad x_{1},...,x_{N} = 0,1,....$$

(Remember that q(i) represents the relative arrival rate of type-*i* (Poisson) customers, i = 1, ..., N (cf. (2.1)), and that ρ denotes the total offered load to the system per unit of time due to the Poisson customers: $\rho = \sum_{i=1}^{N} \lambda \beta q(i)$).

The generating function of the joint queue length distribution is given by: (cf. the derivation of (2.4))

$$E\{z_{1}^{\mathbf{X}_{1}}\cdots z_{N}^{\mathbf{X}_{N}}\} = \sum_{x_{1}=0}^{\infty}\cdots\sum_{x_{N}=0}^{\infty} z_{1}^{x_{1}}\cdots z_{N}^{x_{N}}P(x_{1},\cdots,x_{N}) = (5.50)$$

$$(1-\rho)^{K+1}\frac{1}{K!}\sum_{m=0}^{\infty}\sum_{x_{1}}\cdots\sum_{x_{N}}\frac{(m+K)!}{x_{1}+\cdots+x_{N}!}\prod_{i=1}^{N}(\lambda\beta q(i)z_{i})^{x_{i}} = (1-\rho)^{K+1}\sum_{m=0}^{\infty}\binom{m+K}{K}\left[\sum_{i=1}^{N}\lambda\beta q(i)z_{i}\right]^{m} = \left[\frac{1-\rho}{1-\sum_{i=1}^{N}z_{i}\lambda\beta q(i)}\right]^{K+1},$$

$$|z_{i}| \leq 1, i = 1,...,N.$$

Comparing this result with (2.4) we observe a similar phenomenon as for the standard M/M/1 queue (cf. (5.39)): the presence of the K permanent customers in the M/M/1 feedback queue leads to a joint queue length distribution which is the (K+1)-fold convolution of the joint queue length distribution in the same model without permanent customers. In the next subsection we shall use (5.50) for the analysis of the sojourn time distribution.

5.6.2 Sojourn time distribution

In this section we present, in the form of Laplace-Stieltjes transforms and generating functions, an expression for the joint steady state distribution of the successive sojourn times S_j , j = 1, ..., k, and the number of type-*i* customers,

 $\mathbf{X}_{i}^{(j)}$, i = 1, ..., N, present at the *j*-th service completion of a customer who is fed back at least k - 1 times, k = 1, 2, It will appear that for the derivation of this quantity we can largely rely on the analysis of the sojourn time in the model without permanent customers given in Section 2.3.

Consider a newly arriving (tagged) customer, say C, and suppose that he finds $\mathbf{X}_{i}^{(0)} = x_{i}$ type-*i* (Poisson) customers in the system, i = 1, ..., N, together with the K permanent customers. It is easily seen that the determination of the (conditional) joint sojourn time distribution of C can be performed in almost exactly the same way as for the original M/M/1 feedback queue without permanent customers, see Appendix 2.1, the only difference being that for the present model one has to take into account that after *each* of his services C finds K additional permanent customers). Realizing this it follows immediately from the analysis in Appendix 2.1 that, for Re $\omega_{j} \ge 0$, $|z_{i,j}| \le 1$, i = 1, ..., N, j = 0, ..., k, (cf. (2.8))

$$E\{e^{-(\omega_{1}\mathbf{S}_{1}+...+\omega_{k}\mathbf{S}_{k})}(z_{1,0}^{\mathbf{X}_{1}^{(0)}}\cdots z_{N,0}^{\mathbf{X}_{N}^{(0)}})\cdots (z_{1,k}^{\mathbf{X}_{1}^{(k)}}\cdots z_{N,k}^{\mathbf{X}_{N}^{(k)}}) | \mathbf{X}_{1}^{(0)} = x_{1},\ldots,\mathbf{X}_{N}^{(0)} = x_{N}\}$$

$$= \left[\prod_{j=1}^{k}A_{k}^{N}(j,\omega,z)\right]^{K+1}\prod_{i=1}^{N}(z_{i,0}f_{k}^{N}(i,\omega,z))^{x_{i}},$$
(5.51)

with $\omega := (\omega_1, \ldots, \omega_k)$, $z := ((z_{1,0}, \ldots, z_{N,0}), \ldots, (z_{1,k}, \ldots, z_{N,k}))$, and with $A_k^N(\cdot, \cdot, \cdot)$ and $f_k^N(\cdot, \cdot, \cdot)$ defined by (2.9) and (2.10). Comparing (5.51) with the result ((2.8)) for the corresponding quantity in the model without permanent customers it appears that these results differ only by a factor $(\prod_{j=1}^k A_k^N(j, \omega, z))^K$; note that this could have been obtained directly from the discussion in Remark 2.1.

Using the PASTA property and deconditioning we obtain from (5.50) and (5.51) our main result:

THEOREM 5.2

The joint distribution of the successive sojourn times and the number of Poisson customers of each type present in the system at the service completion epochs of a tagged Poisson customer is the (K+1)-fold convolution of the corresponding (joint) distribution in the model without permanent customers, cf. (2.11):

$$E\{e^{-(\omega_{1}\mathbf{S}_{1}+...+\omega_{k}\mathbf{S}_{k})}(z_{1,0}^{\mathbf{X}_{1}^{(0)}}\cdots z_{N,0}^{\mathbf{X}_{N}^{(0)}})\cdots (z_{1,k}^{\mathbf{X}_{1}^{(k)}}\cdots z_{N,k}^{\mathbf{X}_{N}^{(k)}})\} =$$
(5.52)

$$\left[\frac{(1-\rho)\prod_{j=1}^{k}A_{k}^{N}(j,\omega,z)}{1-\lambda\beta\sum_{i=1}^{N}q(i)z_{i,0}f_{k}^{N}(i,\omega,z)}\right]^{K+1}, \quad \text{Re }\omega_{j} \ge 0, \quad |z_{i,j}| \le 1, \quad i=1,...,N, \quad j=0,...,k$$

Using Theorem 5.2 most of the sojourn time characteristics can be immediately obtained from the results given in Section 2.4. Here we shall restrict ourself to a summary of the most important characteristics.

- The *j*-th sojourn time S_j of a Poisson customer has a (K+1)-stage Erlang distribution (E_{K+1}) with mean $(K+1)\beta/(1-\rho)$: (cf. (2.20))

$$E\{e^{-\omega_{j}\mathbf{S}_{j}}\} = \left(\frac{1-\rho}{1-\rho+\beta\omega_{j}}\right)^{K+1}, \quad j=1,...,k.$$
(5.53)

- The correlation coefficient, $corr(\mathbf{S}_i, \mathbf{S}_j)$, of the *i*-th and the *j*-th sojourn time of a Poisson customer is independent of the number of permanent customers in the system: (cf. (2.24))

$$corr(\mathbf{S}_i, \mathbf{S}_j) = 1 - (1 - \rho)C_{j-i}, \quad 1 \le i < j \le k,$$
(5.54)

with C_n , $n = 1, \dots, k - 1$, determined by (2.22).

- The variance of the total sojourn time after k services, $var(\mathbf{S}^{(k)})$, is given by: (cf. (2.26))

$$var(\mathbf{S}^{(k)}) = (K+1) \left[\frac{\beta}{1-\rho} \right]^2 \left[k^2 - 2(1-\rho) \sum_{j=1}^{k-1} C_{k-j} \right], \quad k = 1, 2, \dots. \quad (5.55)$$

Remark 5.8

Noting that in the present product form model a departing (and hence arriving) permanent customer sees the system in equilibrium with one less customer of his own type (see e.g. Walrand [1988, Section 3.4]) the characteristics of the successive cycle times of a particular permanent customer can be immediately obtained from the above sojourn time results for the Poisson customers. For example, the cycle times have a K-stage Erlang distribution (E_K) with mean $K\beta/(1-\rho)$, cf. (5.53).

5.6.3 The M/G/1 PS queue with additional permanent customers

In Section 3.3 it has been shown how queue length and sojourn time results for the M/G/1 processor sharing queue can be obtained from queue length and sojourn time results for the M/M/1 queue with general feedback. We applied a limiting procedure in which the mean service time $\beta \rightarrow 0$ while the feedback probabilities approach one in such a way that a customer's total required service time remains constant, see Subsection 3.3.1. It is easily seen that application of the same limiting procedure to the present M/M/1 feedback model with K permanent customers leads to the M/G/1 PS queue with K permanent customers. Note that the behaviour of the latter model is independent of the service time distribution(s) of the permanent customers (the permanent customers are always in service). From (5.50) it follows immediately that for the M/G/1 PS queue with K permanent customers, the distribution of the queue length X^{PS} at an arbitrary epoch is the (K+1)-fold convolution of the queue length distribution in the same model without permanent customers: (cf. (3.1))

$$E\{z^{\mathbf{X}^{p_{s}}}\} = \left(\frac{1-\rho}{1-\rho z}\right)^{K+1}, \quad |z| \le 1,$$
(5.56)

i.e.

$$Pr\{\mathbf{X}^{PS}=n\} = (1-\rho)^{K+1} {n+K \choose K} \rho^n, \quad n=0,1,...,$$
(5.57)

with ρ the offered load to the system per unit of time due to the Poisson customers. From Theorem 5.2 we obtain the following remarkable sojourn time result:

THEOREM 5.3

For the M/G/1 PS queue with K permanent customers the distribution of the conditional sojourn time $\mathbf{S}^{PS}(x)$ of a Poisson customer with given service demand x is the (K+1)-fold convolution of the conditional sojourn time in the same model without permanent customers. This also holds for the unconditional sojourn time \mathbf{S}^{PS} of an arbitrary Poisson customer.

Theorem 5.3 implies: (cf. (3.19))

$$E\{\mathbf{S}^{PS}(x)\} = (K+1)\frac{x}{1-\rho}, \quad x \ge 0.$$
(5.58)

Remark 5.9

For the present PS model it is interesting to study the influence of the presence of the Poisson customers on the 'speed' with which the permanent customers are served. For $x \ge 0$ let $\mathbf{C}^{PS}(x)$ be the time required to give the permanent customers an amount x of service. From the discussion in Remark 5.8 and application of the limiting procedure it follows that $\mathbf{C}^{PS}(x)$ is distributed as the conditional sojourn time of a tagged Poisson customer with service demand x in the same model but with one less permanent customer. For example, from (5.58),

$$E\{\mathbf{C}^{PS}(x)\} = K \frac{x}{1-\rho}, \quad x \ge 0.$$
(5.59)

This formula shows that the influence of the Poisson customer stream on

 $E\{\mathbf{C}^{PS}(x)\}$ is simply a reduction of the capacity of the server by an amount ρ . Moreover, (5.59) implies that the mean *total* amount of service obtained by the permanent customers per unit of time (given by $Kx/E\{\mathbf{C}^{PS}(x)\}$) is independent of K.

Remark 5.10

In Remark 3.7 we concluded that for the M/G/1 PS queue (without permanent customers) the queue length distribution just after the departure of a tagged customer who has received an amount x of service is the same as at an arbitrary epoch, *independent* of x. From (5.56) it follows that for the M/G/1PS queue with one permanent customer the queue length distribution at an arbitrary epoch is the two-fold convolution of the queue length distribution in the PS queue without permanent customers. Since one would expect that, when the required service time x of a tagged customer becomes very large, the behaviour of the M/G/1 PS queue approaches that of the corresponding PS queue with one permanent customer, it seems paradoxical that both statements are true. However, viewing the M/G/1 PS queue as the limiting case of the M/M/1 queue with general feedback this is immediately clear (the departure of a tagged customer in the PS model corresponds to the (last) service completion of a tagged customer in the feedback model which is more likely to occur when there are fewer customers in the system). A similar 'paradox' for queue lengths in PS queues is discussed in Foley and Klutke [1989].

Remark 5.11

It should be noted that the M/G/1 PS queue with K permanent customers can also be viewed as a special case of generalized processor sharing (GPS), cf. Section 3.4: if there are j customers present in the system then the service rate for each of these customers is f(j)=1/(j+K), j=1,2,... It is easily verified that (5.57) and (5.58) coincide with the results for the queue length distribution and the mean sojourn time in the M/G/1 GPS queue given by (3.59) and (3.60) respectively, that have already been obtained by Cohen [1979]. Theorem 5.3 is a new result.

Remark 5.12

Using Theorem 5.3 the approximations for second moment characteristics of the sojourn time in the ordinary M/G/1 PS queue (developed in Chapter 4) can be easily extended to approximations for the corresponding quantities in the PS model with permanent customers.

The above results for the queue length and the sojourn time in the M/G/1 PS queue with permanent customers are interesting both from a theoretical and a practical point of view. One example where this queueing model may arise is provided by a 'Stored Program Controlled' (SPC) telephone exchange that is offered two types of jobs: (i) call requests, and (ii) operator tasks (see De Waal [1989]). To guarantee a certain quality of service of the call requests only a limited number (K) of operator tasks is allowed to be in service at the

same time. It is clear that under heavy traffic conditions of the operator tasks and for appropriate assumptions about the system parameters the above formulas (5.56)-(5.58) (approximately) reflect the influence of the choice of the control parameter K on the queue length and the delay of the call requests. From the discussion in Remark 5.9 it follows that under certain conditions the maximum throughput of the operator tasks is independent of K. So, if the objective is to minimize the delay of the call requests and to maximize the throughput of the operator tasks one should take K as small as possible, i.e. K=1.

References

- Adiri, I. (1972). Queueing models for multiprogrammed computers. In: Proc. Symp. on Computer-Communications Networks and Teletraffic, Polytechnic Institute of Brooklyn, 441-448.
- Adiri, I., Avi-Itzhak, B. (1969A). A time-sharing queue. Management Science 15, 639-657.
- Adiri, I., Avi-Itzhak, B. (1969B). A time-sharing model with many queues. Oper. Res. 17, 1077-1089.
- Ali, O.M.E., Neuts, M.F. (1984). A service system with two stages of waiting and feedback of customers. J. Appl. Prob. 21, 404-413.
- Asare, B.K., Foster, F.G. (1983). Conditional response times in the M/G/1 processor-sharing system. J. Appl. Prob. 20, 910-915.
- Avi-Itzhak, B., Halfin, S. (1988). Response times in M/M/1 time sharing schemes with a limited number of service positions. J. Appl. Prob. 25, 579-595.
- Avi-Itzhak, B., Halfin, S. (1989A). Expected response times in a non-symmetric time sharing queue with limited number of service positions. In: *Teletraffic Science for New Cost-Effective Systems, Networks and Services,* ed. M. Bonatti. North-Holland Publ. Cy., Amsterdam, 1485-1493.
- Avi-Itzhak, B., Halfin, S. (1989B). Response times in gated M/G/1 queues: the processor-sharing case. *Queueing Systems* 4, 263-279.
- Baskett, F., Chandy, K.M., Muntz, R.R., Palacios, F.G. (1975). Open, closed, and mixed networks of queues with different classes of customers. J. ACM 22, 248-260.
- Van den Berg, J.L. (1988). Unpublished work.
- Van den Berg, J.L. (1989). Simple approximations for second moment characteristics of the sojourn time in the M/G/1 processor sharing queue. In: *Messung, Modellierung und Bewertung von Rechensystemen und Netzen*, eds. G. Stiege, J.S. Lie. Springer-Verlag, Berlin, 105-120.
- Van den Berg, J.L., Boxma, O.J. (1989A). Sojourn times in feedback queues. In: Operations Research Proceedings 1988, eds. D. Pressmar et al.. Springer-Verlag, Berlin, 247-257.
- Van den Berg, J.L., Boxma, O.J. (1989B). Sojourn times in feedback and processor sharing queues. In: *Teletraffic Science for New Cost-Effective Systems, Networks and Services*, ed. M. Bonatti. North-Holland Publ. Cy., Amsterdam, 1467-1475.
- Van den Berg, J.L., Boxma, O.J., Groenendijk, W.P. (1989). Sojourn times in the M/G/1 queue with deterministic feedback, *Stochastic Models* 5, 115-129.

Van den Berg, J.L., Boxma, O.J. (1991). The M/G/1 queue with processor

sharing and its relation to a feedback queue, Queueing Systems 9, 365-401.

- Boxma, O.J. (1986A). A queueing model of finite and infinite source interaction. Oper. Res. Letters 5, 245-254.
- Boxma, O.J. (1986B). Models of two queues a few new views. In: *Teletraffic* Analysis and Computer Performance Evaluation, eds. O.J. Boxma, J.W. Cohen, H.C. Tijms. North-Holland Publ. Cy., Amsterdam.
- Boxma, O.J., Cohen, J.W. (1989). The M/G/1 queue with permanent customers. Report BS-R8919, Centre for Mathematics and Computer Science, Amsterdam.
- Boxma, O.J., Daduna, H. (1989). Sojourn times in queueing networks. Report BS-R8916, Centre for Mathematics and Computer Science, Amsterdam. To appear in: Stochastic Analysis of Computer and Communication Systems, ed. H. Takagi. North-Holland Publ. Cy., Amsterdam, 1990.
- Çinlar, E. (1975). Introduction to Stochastic Processes. Prentice-Hall, Englewood Cliffs (NJ).
- Coffman, E.G., Kleinrock, L. (1968). Feedback queueing models for timeshared systems. J. ACM 15, 549-576.
- Coffman, E.G., Muntz, R.R., Trotter, H. (1970). Waiting time distributions for processor-sharing systems. J. ACM 17, 123-130.
- Cohen, J.W. (1976). On Regenerative Processes in Queueing Theory. Springer-Verlag, Berlin.
- Cohen, J.W. (1979). The multiple phase service network with generalized processor sharing. Acta Informatica 12, 245-284.
- Cohen, J.W. (1982). *The Single Server Queue*, 2nd ed. North-Holland Publ. Cy., Amsterdam.
- Disney, R.L. (1981). A note on sojourn times in M/G/1 queues with instantaneous Bernoulli feedback. Nav. Res. Log. Quart. 28, 679-684.
- Disney, R.L., König, D. (1985). Queueing networks: a survey of their random processes. SIAM Review 27, 335-403.
- Disney, R.L., König, D., Schmidt, V. (1984). Stationary queue-length and waiting-time distributions in single-server feedback queues. Adv. Appl. Prob. 16, 437-446.
- Disney, R.L., McNickle, D.C., Simon, B. (1980). The M/G/1 queue with instantaneous Bernoulli feedback. Nav. Res. Log. Quart. 27, 635-644.
- Doshi, B.T. (1986). Queueing systems with vacations a survey. Queueing Systems 1, 29-66.
- Doshi, B.T., Kaufman, J.S. (1988). Sojourn time in an M/G/1 queue with Bernoulli feedback. In: Queueing Theory and its Applications - Liber Amicorum for J.W. Cohen, eds. O.J. Boxma and R. Syski. North-Holland Publ. Cy., Amsterdam, 207-233.
- Doshi, B.T., Rege, K.M. (1985). Analysis of a multistage queue. AT&T Tech.

J. **64**, 369-390.

- Doshi, B.T., Wong, W.S. (1987). Exact solution of a simple finite infinite source interaction model. *Queueing Systems* 2, 67-82.
- Feller, W. (1950). An Introduction to Probability Theory and Its Applications, Vol. I. Wiley, New York.
- Foley, R.D., Klutke, G.A. (1989). Stationary increments in the accumulated work process in processor-sharing queues. J. Appl. Prob. 26, 671-677.
- Fontana, B., Diaz Berzosa, C. (1984). Stationary queue-length distributions in an M/G/1 queue with two non-preemptive priorities and general feedback. In: *Performance of Computer-Communication Systems*, eds. H. Rudin and W. Bux. North-Holland Publ. Cy., Amsterdam, 333-347.
- Fontana, B., Diaz Berzosa, C. (1985). M/G/1 queue with N-priorities and feedback: joint queue-length distributions and response time distribution for any particular sequence. In: *Teletraffic Issues in an Advanced Information Society*, ITC-11, ed. M. Akiyama. North-Holland Publ. Cy., Amsterdam, 452-458.
- Fuhrmann, S.W., Cooper, R.B. (1985). Stochastic decomposition in the M/G/1 queue with generalized vacations. *Oper. Res.* 33, 1117-1129.
- Groenendijk, W.P. (1990). Conservation Laws in Polling Systems, Ph.D. Thesis. Centre for Mathematics and Computer Science, Amsterdam.
- Hunter, J.J. (1989). Sojourn time problems in feedback queues. Queueing Systems 5, 55-76.
- Jackson, J.R. (1957). Networks of waiting lines. Oper. Res. 5, 518-521.
- Jackson, J.R. (1963). Jobshop-like queueing systems. *Management Science* 10, 131-142.
- Jaiswal, N.K. (1982). Performance evaluation studies for time-sharing computer systems. *Perfermance Evaluation* 2, 223-236.
- Kaufman, J.S. (1985). Finite and infinite source interactions. In: *Performance* '84, ed. E. Gelenbe. North-Holland Publ. Cy., Amsterdam.
- Kelly, F.P. (1979). Reversibility and Stochastic Networks. Wiley, New York.
- Kleinrock, L. (1964). Analysis of a time-shared processor. Nav. Res. Log. Quart. 11, 59-73.
- Kleinrock, L. (1967). Time-shared systems: a theoretical treatment. J. ACM 14, 242-261.
- Kleinrock, L. (1975). Queueing Systems, Vol. I. Wiley, New York.
- Kleinrock, L. (1976). Queueing Systems, Vol. II. Wiley, New York.
- Klutke, G.A., Kiessler, P.C., Disney, R.L. (1988). Interoutput times in processor sharing queues with feedback. *Queueing Systems* 3, 363-376.
- Lam, S.S., Shankar, A.U. (1981). A derivation of response time distributions for a multi-class feedback queueing system. *Performance Evaluation* 1, 48-61.

- Lemoine, A.J. (1979). Total sojourn time in networks of queues. Report TR 79-020-1, Systems Control, Inc., Palo Alto, CA.
- Muntz, R.R. (1972). Waiting time distribution for round-robin queueing systems. In: Proc. Symp. on Computer-Communications Networks and Teletraffic, Polytechnic Institute of Brooklyn, 429-439.
- Nelson, R.D. (1987). Expected response time for a FCFS feedback queue with multiple classes. Research Report RC 13221, IBM Research Division, T.J. Watson Research Center, Yorktown Heights, N.Y.
- O'Donovan, T.M. (1974). Direct solutions of M/G/1 processor-sharing models. Oper. Res. 22, 1232-1235.
- Ott, T.J. (1984). The sojourn-time distribution in the M/G/1 queue with processor sharing. J. Appl. Prob. 21, 360-378.
- Rege, K.M., Sengupta, B. (1989). A single server queue with gated processorsharing discipline. *Queueing Systems* 4, 249-261.
- Resing, J.A.C., Hooghiemstra, G., Keane, M.S. (1989). The M/G/1 processor sharing queue as the almost sure limit of feedback queues. *Report no. 89-32, Faculty of Technical Mathematics and Informatics, Delft University of Technology.* To appear in: J. Appl. Prob. 27.
- Sakata, M., Noguchi, S., Oizumi, J. (1969). Analysis of a processor shared queueing model for time sharing systems. *Proc. 2nd Hawaii Int. Conf. on System Sciences*, 625-628.
- Sakata, M., Noguchi, S., Oizumi, J. (1971). An analysis of the M/G/1 queue under round-robin scheduling. Oper. Res. 19, 371-385.
- Schassberger, R. (1981). On the response time distribution in a discrete round-robin queue. Acta Informatica 16, 57-62.
- Schassberger, R. (1984). A new approach to the M/G/1 processor-sharing queue. Adv. Appl. Prob. 16, 202-213.
- Schrage, L.E. (1967). The queue M/G/1 with feedback to lower priority queues. *Management Science* 13, 466-474.
- Shanthikumar, J.G. (1988). On stochastic decomposition in M/G/1 type queues with generalized server vacations. *Oper. Res.* 36, 566-569.
- Simon, B. (1984). Priority queues with feedback. J. ACM 31, 134-149.
- Stoyan, D. (1983). Comparison Methods for Queues and Other Stochastic Models. Wiley, New York.
- Takács, L. (1963). A single-server queue with feedback. *Bell System Tech. J.* **42**, 505-519.
- Takagi, H. (1986). Analysis of Polling Systems. The MIT Press, Cambridge, Massachusetts.
- Tijms, H.C. (1986). Stochasic Modeling and Analysis. Wiley, New York.
- Titchmarsh, E.C. (1968). *The Theory of Functions*, 2nd ed. Oxford University Press, London.

Voelker, D., Doetsch, G. (1950). Die Zweidimensionale Laplace-Transformation. Verlag Birkhäuser, Basel.

- De Waal, P.R. (1989). An approximation method for a processor sharing queue with controlled arrivals and a waitbuffer. In: *Proc. of the 28th IEEE Conf. on Decision and Control, Tampa, FL.*
- Walrand, J. (1988). An Introduction to Queueing Networks. Prentice-Hall, Englewood Cliffs (NJ).
- Walrand, J., Varaiya, P. (1980). Sojourn times and the overtaking condition in Jacksonian networks. Adv. Appl. Prob. 12, 1000-1018.
- Wang, Y.T. (1981). An analysis of a round-robin schedule with preemptive priorities. In: *Performance '81*, ed. F.J. Kylstra. North-Holland Publ. Cy., Amsterdam, 147-157.
- Whitt, W. (1982). Approximating a point process by a renewal process, I: Two basic methods. Oper. Res. 30, 125-147.
- Whitt, W. (1984). On approximations for queues, III: Mixtures of exponential distributions. AT&T Bell Lab. Tech. J. 63, 163-175.
- Wolff, R.W. (1982). Poisson arrivals see time averages. Oper. Res. 30, 223-231.
- Yashkov, S.F. (1983). A derivation of response time distribution for a M/G/1 processor sharing queue. *Problems Contr. & Info. Theory* 12, 133-148.
- Yashkov, S.F. (1986). A note on asymptotic estimates of the sojourn time variance in the M/G/1 queue with processor-sharing. Syst. Anal. Model. Simul. 3, 267-269.
- Yashkov, S.F. (1987). Processor-sharing queues: some progress in analysis. *Queueing Systems* 2, 1-17.

SUBJECT INDEX

Arrival theorem	23
BCMP networks	3
Bernoulli feedback, M/M/1 queue with	3, 28
- sojourn time	30
Coxian distribution	46, 75
Cyclic service models	91, 109
Deterministic feedback, M/M/1 queue with	28, 31
- sojourn time	32
Decomposition theorem for queues with server vacations	96, 97
Erlang distribution	46, 75
Feedback queues	3,17
Feedforward networks	2, 3
General feedback, M/M/1 queue with	17
- queue length distribution	19, 20
- sojourn time distribution	22, 24
- sojourn time variance	27
General feedback, M/G/1 queue with	33
- mean sojourn time	35
General feedback, M/G/1 Processor Sharing queue with	65
- sojourn time	68, 69, 70
Generalized processor sharing, M/G/1 queue with	8, 61
- queue length distribution	61
- sojourn time	62, 64
Hyper exponential distribution	46, 75
- balanced means	46, 75
- gamma normalization	46, 75
Markov renewal process	5, 21
PASTA property	20
Permanent customers, M/G/1 queue with	8, 90
- queue length distribution	103
- sojourn time distribution	104
Permanent customers, M/M/1 feedback queue with	110
- queue length distribution	111
- sojourn time distribution	112
Permanent customers, M/G/1 Processor Sharing queue with	113
- queue length distribution	114
- sojourn time distribution	114

	123
Polling system	109
Processor sharing, M/G/1 queue with	6, 11, 42
- queue length distribution	42
- sojourn time variance	51
- sojourn time distribution	59
- approximations for second moment of sojourn time	71, 76, 77, 80
Processor sharing queues with feedback	65
- sojourn time	68, 69
Product form distribution	19
Quality control in production process	1
Round robin service discipline	6
SPC telephone exchange	115, 116
Squared coefficient of variation	75
State dependent service rates, M/M/1 queue with	61
Time sharing	5, 11
Vacation queues	91

.

.

MATHEMATICAL CENTRE TRACTS

1 T. van der Walt. Fixed and almost fixed points. 1963.

A.R. Bloemena. Sampling from a graph. 1964.
 G. de Leve. Generalized Markovian decision processes, part I: model and method. 1964.

4 G. de Leve. Generalized Markovian decision processes, part II: probabilistic background. 1964.

5 G. de Leve, H.C. Tijms, P.J. Weeda. Generalized Markovian decision processes, applications. 1970.

6 M.A. Maurice. Compact ordered spaces. 1964. 7 W.R. van Zwet. Convex transformations of random variables. 1964

8 J.A. Zonneveld. Automatic numerical integration. 1964.

9 P.C. Baayen. Universal morphisms. 1964.

10 E.M. de Jager. Applications of distributions in mathematical physics. 1964.

11 A.B. Paalman-de Miranda. Topological semigroups. 1964. 12 J.A.Th.M. van Berckel, H. Brandt Corstius, R.J. Mokken, A. van Wijngaarden. Formal properties of newspaper Dutch. 1965.

13 H.A. Lauwerier. Asymptotic expansions. 1966, out of print; replaced by MCT 54.

14 H.A. Lauwerier. Calculus of variations in mathematical physics. 1966.

15 R. Doornbos. Slippage tests. 1966.

16 J.W. de Bakker. Formal definition of programming languages with an application to the definition of ALGOL 60. 1967.

17 R.P. van de Riet. Formula manipulation in ALGOL 60, part 1. 1968.

18 R.P. van de Riet. Formula manipulation in ALGOL 60, part 2. 1968.

19 J. van der Slot. Some properties related to compactness. 1968.

20 P.J. van der Houwen. Finite difference methods for solving partial differential equations. 1968.

21 E. Wattel. The compactness operator in set theory and topology. 1968.

22 T.J. Dekker. ALGOL 60 procedures in numerical algebra, part 1, 1968.

23 T.J. Dekker, W. Hoffmann. ALGOL 60 procedures in numerical algebra, part 2. 1968.

24 J.W. de Bakker. Recursive procedures. 1971.

25 E.R. Paërl. Representations of the Lorentz group and projec-tive geometry. 1969.

26 European Meeting 1968. Selected statistical papers, part 1. 1968.

27 European Meeting 1968. Selected statistical papers, part II. 1968

28 J. Oosterhoff. Combination of one-sided statistical tests. 1969

29 J. Verhoeff. Error detecting decimal codes. 1969.

30 H. Brandt Corstius. Exercises in computational linguistics. 1970.

31 W. Molenaar. Approximations to the Poisson, binomial and hypergeometric distribution functions. 1970.

32 L. de Haan. On regular variation and its application to the weak convergence of sample extremes. 1970.

33 F.W. Steutel. Preservation of infinite divisibility under mix-ing and related topics. 1970.

34 I. Juhász, A. Verbeek, N.S. Kroonenberg. Cardinal func-tions in topology. 1971.

35 M.H. van Emden. An analysis of complexity. 1971.

36 J. Grasman. On the birth of boundary layers. 1971. 37 J.W. de Bakker, G.A. Blaauw, A.J.W. Duijvestijn, E.W. Dijkstra, P.J. van der Houwen, G.A.M. Kamsteeg-Kemper, F.E.J. Kruseman Aretz, W.L. van der Poel, J.P. Schaap-Kruseman, M.V. Wilkes, G.Zoutendijk. MC-25 Informatica Symposium, 1971.

38 W.A. Verloren van Themaat. Automatic analysis of Dutch compound words. 1972.

39 H. Bavinck. Jacobi series and approximation. 1972.

40 H.C. Tijms. Analysis of (s,S) inventory models. 1972.

41 A. Verbeek. Superextensions of topological spaces. 1972.

42 W. Vervaat. Success epochs in Bernoulli trials (with applica-tions in number theory). 1972.

43 F.H. Ruymgaart. Asymptotic theory of rank tests for independence. 1973.

44 H. Bart. Meromorphic operator valued functions. 1973. 45 A.A. Balkema. Monotone transformations and limit laws. 1973.

1913.
46 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 1: the language. 1973.
47 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 2: the compiler. 1973.
48 F.E.J. Kruseman Aretz, P.J.W. ten Hagen, H.L. Oudshoorn. An ALGOL 60 compiler in ALGOL 60, text of the MC-compiler for the EL-X8. 1973. 49 H. Kok. Connected orderable spaces. 1974.

5 A. van Wijngaarden, B.J. Mailloux, J.E.L. Peck, C.H.A. Koster, M. Sintzoff, C.H. Lindsey, L.G.L.T. Meertens, R.G. Fisker (eds.). *Revised report on the algorithmic language* ALGOL 68, 1976.

51 A. Hordijk. Dynamic programming and Markov potential theory. 1974.

52 P.C. Baayen (ed.). Topological structures. 1974. 53 M.J. Faber. Metrizability in generalized ordered spaces. 1974.

54 H.A. Lauwerier. Asymptotic analysis, part 1. 1974.

55 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 1: theory of designs, finite geometry and coding theory. 1974. 56 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 2: graph theory. foundations, partitions and combinatorial geometry. 1974.

57 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 3: combinatorial group theory. 1974.

58 W. Albers. Asymptotic expansions and the deficiency concept in statistics. 1975.

59 J.L. Mijnheer. Sample path properties of stable processes. 1074

60 F. Göbel. Queueing models involving buffers. 1975. 63 J.W. de Bakker (ed.). Foundations of computer science. 1975

64 W.J. de Schipper. Symmetric closed categories. 1975. 65 J. de Vries. Topological transformation groups, 1: a categorical approach. 1975.

66 H.G.J. Pijls. Logically convex algebras in spectral theory and eigenfunction expansions. 1976.

of 8 P.P.N. de Groen. Singularly perturbed differential operators of second order. 1976.

69 J.K. Lenstra. Sequencing by enumerative methods. 1977

70 W.P. de Roever, Jr. Recursive program schemes: semantics and proof theory. 1976.

71 J.A.E.E. van Nunen. Contracting Markov decision processes. 1976.

72 J.K.M. Jansen. Simple periodic and non-periodic Lamé functions and their applications in the theory of conical waveguides. 1977.

73 D.M.R. Leivant. Absoluteness of intuitionistic logic. 1979. 74 H.J.J. te Riele. A theoretical and computational study of generalized aliquot sequences. 1976.

75 A.E. Brouwer. Treelike spaces and related connected topo-logical spaces. 1977.

76 M. Rem. Associons and the closure statement. 1976.

77 W.C.M. Kallenberg. Asymptotic optimality of likelihood ratio tests in exponential families. 1978. 78 E. de Jonge, A.C.M. van Rooij. Introduction to Riesz spaces. 1977.
79 M.C.A. van Zuijlen. Emperical distributions and rank statistics. 1977.

80 P.W. Hemker. A numerical study of stiff two-point boundary problems, 1977

81 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 1. 1976. 82 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 2. 1976.

83 L.S. van Benthem Jutting. Checking Landau's "Grundlagen" in the AUTOMATH system. 1979.

84 H.L.L. Busard. The translation of the elements of Euclid from the Arabic into Latin by Hermann of Carinthia (?), books vii-xii. 1977.

85 J. van Mill. Supercompactness and Wallman spaces. 1977. 86 S.G. van der Meulen, M. Veldhorst. Torrix I, a program-ming system for operations on vectors and matrices over arbi-trary fields and of variable size. 1978.

88 A. Schrijver. Matroids and linking systems. 1977. 89 J.W. de Roever. Complex Fourier transformation and analytic functionals with unbounded carriers. 1978.

90 L.P.J. Groenewegen, Characterization of optimal strategies in dynamic games. 1981.

91 J.M. Geysel. Transcendence in fields of positive characteris-tic. 1979.

92 P.J. Weeda. Finite generalized Markov programming. 1979. 93 H.C. Tijms, J. Wessels (eds.). Markov decision theory. 1977.

94 A. Bijlsma. Simultaneous approximations in transcendental number theory. 1978.

95 K.M. van Hee. Bayesian control of Markov chains. 1978. 96 P.M.B. Vitányi. Lindenmayer systems: structure, languages, and growth functions. 1980.

97 A. Federgruen. Markovian control problems; functional equations and algorithms. 1984.

98 R. Geel. Singular perturbations of hyperbolic type. 1978. 99 J.K. Lenstra, A.H.G. Rinnooy Kan, P. van Emde Boas (eds.). Interfaces between computer science and operations research. 1978.

100 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part 1. 1979.

101 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part ings bice 2. 1979.

102 D. van Dulst. Reflexive and superreflexive Banach spaces. 1978

103 K. van Harn. Classifying infinitely divisible distributions by functional equations. 1978.

104 J.M. van Wouwe. Go-spaces and generalizations of metri-zability. 1979.

105 R. Helmers. Edgeworth expansions for linear combinations of order statistics. 1982.

106 A. Schrijver (ed.). Packing and covering in combinatorics. 1979

107 C. den Heijer. The numerical solution of nonlinear opera-tor equations by imbedding methods. 1979.

108 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 1. 1979.

109 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 2. 1979.

110 J.C. van Vliet. ALGOL 68 transput, part 1: historical review and discussion of the implementation model. 1979.

111 J.C. van Vliet. ALGOL 68 transput, part II: an implemen-tation model. 1979.

112 H.C.P. Berbee. Random walks with stationary increments and renewal theory. 1979.

113 T.A.B. Snijders. Asymptotic optimality theory for testing problems with restricted alternatives. 1979.

114 A.J.E.M. Janssen. Application of the Wigner distribution to harmonic analysis of generalized stochastic processes. 1979.

115 P.C. Baayen, J. van Mill (eds.). Topological structures II, part 1. 1979.

116 P.C. Baayen, J. van Mill (eds.). Topological structures 11, part 2. 1979.

117 P.J.M. Kallenberg. Branching processes with continuous state space. 1979.

118 P. Groeneboom. Large deviations and asymptotic efficiencies. 1980

119 F.J. Peters. Sparse matrices and substructures, with a novel implementation of finite element algorithms. 1980.

120 W.P.M. de Ruyter. On the asymptotic analysis of large-scale ocean circulation. 1980.

121 W.H. Haemers. Eigenvalue techniques in design and graph theory, 1980. 122 J.C.P. Bus. Numerical solution of systems of nonlinear equations, 1980.

123 I. Yuhász. Cardinal functions in topology - ten years later. 1980.

124 R.D. Gill. Censoring and stochastic integrals. 1980.

125 R. Eising. 2-D systems, an algebraic approach. 1980. 126 G. van der Hoek. Reduction methods in nonlinear pro-gramming. 1980.

127 J.W. Klop. Combinatory reduction systems. 1980.

128 A.J.J. Talman. Variable dimension fixed point algorithms and triangulations. 1980.

129 G. van der Laan. Simplicial fixed point algorithms. 1980. 130 P.J.W. ten Hagen, T. Hagen, P. Klint, H. Noot, H.J. Sint, A.H. Veen. *ILP: intermediate language for pictures.* 1980

131 R.J.R. Back. Correctness preserving program refinements: proof theory and applications. 1980.

132 H.M. Mulder. The interval function of a graph. 1980. 133 C.A.J. Klaassen. Statistical performance of location esti-mators. 1981.

134 J.C. van Vliet, H. Wupper (eds.). Proceedings interna-tional conference on ALGOL 68. 1981.

135 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.). Formal methods in the study of language, part I. 1981.

136 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.). Formal methods in the study of language, part 11. 1981. 137 J. Telgen. Redundancy and linear programs. 1981.

138 H.A. Lauwerier. Mathematical models of epidemics. 1981.

139 J. van der Wal. Stochastic dynamic programming, succes-sive approximations and nearly optimal strategies for Markov decision processes and Markov games. 1981.

140 J.H. van Geldrop. A mathematical theory of pure exchange economies without the no-critical-point hypothesis. 1981.

141 G.E. Welters. Abel-Jacobi isogenies for certain types of Fano threefolds. 1981.

142 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 1. 1981.

143 J.M. Schumacher. Dynamic feedback in finite- and infinite-dimensional linear systems. 1981.

144 P. Eijgenraam. The solution of initial value problems using interval arithmetic; formulation and analysis of an algorithm. 1981.

145 A.J. Brentjes. Multi-dimensional continued fraction algo-rithms, 1981.

146 C.V.M. van der Mee. Semigroup and factorization methods in transport theory. 1981.

147 H.H. Tigelaar. Identification and informative sample size. 1982

148 L.C.M. Kallenberg. Linear programming and finite Mar-kovian control problems. 1983.

149 C.B. Huijsmans, M.A. Kaashoek, W.A.J. Luxemburg, W.K. Vietsch (eds.). From A to Z, proceedings of a symposium in honour of A.C. Zaanen. 1982.

150 M. Veldhorst. An analysis of sparse matrix storage schemes. 1982.

151 R.J.M.M. Does. Higher order asymptotics for simple linear rank statistics. 1982.

152 G.F. van der Hoeven. Projections of lawless sequences. 1982

153 J.P.C. Blanc. Application of the theory of boundary value problems in the analysis of a queueing model with paired services. 1982.

Vices. 1962.
154 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part I. 1982.
155 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part II. 1982.
156 D.M.C. Appen Computer and Actional Methods.

156 P.M.G. Apers. Query processing and data allocation in distributed database systems. 1983.

157 H.A.W.M. Kneppers. The covariant classification of two-dimensional smooth commutative formal groups over an alge-braically closed field of positive characteristic. 1983.

158 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 1. 1983.

159 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 2. 1983.

160 A. Rezus. Abstract AUTOMATH. 1983.

161 G.F. Helminck. Eisenstein series on the metaplectic group, an algebraic approach. 1983.

162 J.J. Dik. Tests for preference. 1983.

163 H. Schippers. Multiple grid methods for equations of the second kind with applications in fluid mechanics. 1983.

164 F.A. van der Duyn Schouten. Markov decision processes with continuous time parameter. 1983.

165 P.C.T. van der Hoeven. On point processes. 1983.

166 H.B.M. Jonkers. Abstraction, specification and implemen tation techniques, with an application to garbage collection. 1983.

167 W.H.M. Zijm. Nonnegative matrices in dynamic program-ming. 1983.

168 J.H. Evertse. Upper bounds for the numbers of solutions of diophantine equations. 1983.

169 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 2. 1983.

CWI TRACTS

1 D.H.J. Epema. Surfaces with canonical hyperplane sections. 1984

2 J.J. Dijkstra. Fake topological Hilbert spaces and characteri-zations of dimension in terms of negligibility. 1984.

3 A.J. van der Schaft. System theoretic descriptions of physical systems. 1984.

4 J. Koene. Minimal cost flow in processing networks, a primal approach. 1984.

5 B. Hoogenboom. Intertwining functions on compact Lie groups. 1984.

6 A.P.W. Böhm. Dataflow computation. 1984.

7 A. Blokhuis. Few-distance sets. 1984.

8 M.H. van Hoorn. Algorithms and approximations for queue-ing systems. 1984.

9 C.P.J. Kovmans. Models of the lambda calculus, 1984.

10 C.G. van der Laan, N.M. Temme. Calculation of special functions: the gamma function, the exponential integrals and error-like functions. 1984.

11 N.M. van Dijk. Controlled Markov processes; time-discretization. 1984.

12 W.H. Hundsdorfer. The numerical solution of nonlinear stiff initial value problems: an analysis of one step methods. 1985.

13 D. Grune. On the design of ALEPH. 1985.

14 J.G.F. Thiemann. Analytic spaces and dynamic program-ming: a measure theoretic approach. 1985.

15 F.J. van der Linden. Euclidean rings with two infinite primes. 1985.

16 R.J.P. Groothuizen. Mixed elliptic-hyperbolic partial dif-ferential operators: a case-study in Fourier integral operators. 1985.

17 H.M.M. ten Eikelder. Symmetries for dynamical and Hamiltonian systems. 1985.

18 A.D.M. Kester. Some large deviation results in statistics.

19 T.M.V. Janssen. Foundations and applications of Montague grammar, part 1: Philosophy, framework, computer science. gram 1986.

20 B.F. Schriever. Order dependence. 1986.

21 D.P. van der Vecht. Inequalities for stopped Brownian motion. 1986.

22 J.C.S.P. van der Woude. Topological dynamix. 1986. 23 A.F. Monna. Methods, concepts and ideas in mathematics: aspects of an evolution. 1986.

24 J.C.M. Baeten. Filters and ultrafilters over definable sub-sets of admissible ordinals. 1986.

25 A.W.J. Kolen. Tree network and planar rectilinear location theory. 1986.

26 A.H. Veen. The misconstrued semicolon: Reconciling imperative languages and dataflow machines. 1986. 27 A.J.M. van Engelen. Homogeneous zero-dimensional abso-lute Borel sets. 1986.

28 T.M.V. Janssen. Foundations and applications of Montague grammar, part 2: Applications to natural language. 1986.

29 H.L. Trentelman. Almost invariant subspaces and high gain feedback. 1986.

30 A.G. de Kok. Production-inventory control models: approximations and algorithms. 1987.

31 E.E.M. van Berkum. Optimal paired comparison designs for factorial experiments. 1987. 32 J.H.J. Einmahl. Multivariate empirical processes. 1987.

33 O.J. Vrieze. Stochastic games with finite state and action

spaces. 1987. 34 P.H.M. Kersten. Infinitesimal symmetries: a computational approach. 1987.

35 M.L. Eaton. Lectures on topics in probability inequalities. 1987.

36 A.H.P. van der Burgh, R.M.M. Mattheij (eds.). Proceedings of the first international conference on industrial and applied mathematics (ICIAM 87). 1987.

37 L. Stougie. Design and analysis of algorithms for stochastic integer programming. 1987.

38 J.B.G. Frenk. On Banach algebras, renewal measures and

regenerative processes. 1987.

39 H.J.M. Peters, O.J. Vrieze (eds.). Surveys in game theory and related topics. 1987.

40 J.L. Geluk, L. de Haan. Regular variation, extensions and Tauberian theorems. 1987.

41 Sape J. Mullender (ed.). The Amoeba distributed operating system: Selected papers 1984-1987. 1987.

42 P.R.J. Asveld, A. Nijholt (eds.). Essays on concepts, for-malisms, and tools. 1987.

43 H.L. Bodlaender. Distributed computing: structure and nplexity. 1987. 44 A.W. van der Vaart. Statistical estimation in large parame-

ter spaces. 1988.

45 S.A. van de Geer. Regression analysis and empirical processes. 1988.

46 S.P. Spekreijse. Multigrid solution of the steady Euler equa-tions. 1988.

47 J.B. Dijkstra. Analysis of means in some non-standard situations. 1988.

48 F.C. Drost. Asymptotics for generalized chi-square goodness-of-fit tests. 1988. 49 F.W. Wubs. Numerical solution of the shallow-water equa-

tions. 1988. 50 F. de Kerf. Asymptotic analysis of a class of perturbed Korteweg-de Vries initial value problems. 1988.

51 P.J.M. van Laarhoven. Theoretical and computational aspects of simulated annealing. 1988.

52 P.M. van Loon. Continuous decoupling transformations for linear boundary value problems. 1988.

53 K.C.P. Machielsen. Numerical solution of optimal control problems with state constraints by sequential quadratic pro-gramming in function space. 1988.

54 L.C.R.J. Willenborg. Computational aspects of survey data processing. 1988.

55 G.J. van der Steen. A program generator for recognition, parsing and transduction with syntactic patterns. 1988.

56 J.C. Ebergen. Translating programs into delay-insensitive circuits. 1989.

57 S.M. Verduyn Lunel. Exponential type calculus for linear delay equations. 1989.

58 M.C.M. de Gunst. A random model for plant cell popula-tion growth. 1989.

59 D. van Dulst. Characterizations of Banach spaces not con-taining l^1 . 1989.

60 H.E. de Swart. Vacillation and predictability properties of low-order atmospheric spectral models. 1989.

61 P. de Jong. Central limit theorems for generalized multil-inear forms. 1989.

62 V.J. de Jong. A specification system for statistical software. 1989.

63 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part I. 1989.

64 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part II. 1989. 65 B.M.M. de Weger. Algorithms for diophantine equations. 1989.

66 A. Jung. Cartesian closed categories of domains. 1989.

67 J.W. Polderman. Adaptive control & identification: Conflict or conflux?. 1989.

68 H.J. Woerdeman. Matrix and operator extensions. 1989. 69 B.G. Hansen. Monotonicity properties of infinitely divisible distributions. 1989.

70 J.K. Lenstra, H.C. Tijms, A. Volgenant (eds.). Twenty-five years of operations research in the Netherlands: Papers dedi-cated to Gijs de Leve. 1990.

71 P.J.C. Spreij. Counting process systems. Identification and stochastic realization. 1990.

72 J.F. Kaashoek. Modeling one dimensional pattern formation by anti-diffusion. 1990.

73 A.M.H. Gerards. Graphs and polyhedra. Binary spaces and cutting planes. 1990.

74 B. Koren. Multigrid and defect correction for the steady Navier-Stokes equations. Application to aerodynamics. 1991. 75 M.W.P. Savelsbergh. Computer aided routing. 1992.

76 O.E. Flippo. Stability, duality and decomposition in general mathematical programming. 1991. 77 A.J. van Es. Aspects of nonparametric density estimation. 1991.

78 G.A.P. Kindervater. Exercises in parallel combinatorial computing. 1992.

79 J.J. Lodder. Towards a symmetrical theory of generalized functions. 1991.

80 S.A. Smulders. Control of freeway traffic flow. 1993.

.

81 P.H.M. America, J.J.M.M. Rutten. A parallel object-oriented language: design and semantic foundations. 1992. 82 F. Thuijsman. Optimality and equilibria in stochastic games. 1992.

R.J. Kooman. Convergence properties of recurrence sequences. 1992.
 A.M. Cohen (ed.). Computational aspects of Lie group representations and related topics. Proceedings of the 1990 Computational Algebra Seminar at CWI, Amsterdam. 1991.

V. de Valk. One-dependent processes. 1993.
 A. J.A. Baars, J.A.M. de Groot. On topological and linear equivalence of certain function spaces. 1992.

87 A.F. Monna. *The way of mathematics and mathematicians*. 1992.

88 E.D. de Goede. Numerical methods for the three-dimensional shallow water equations. 1993.

89 M. Zwaan. Moment problems in Hilbert space with applica-tions to magnetic resonance imaging. 1993.

90 C. Vuik. The solution of a one-dimensional Stefan problem. 1993.

91. E.R. Verheul. Multimedians in metric and normed spaces. 1993.

92. J.L.M. Maubach. Iterative methods for non-linear partial differential equations. 1993.

93. A.W. Ambergen. Statistical uncertainties in posterior pro-babilities. 1993.

P.A. Zegeling, Moving-grid methods for time-dependent partial differential equations. 1993.
 M.J.C. van Pul. Statistical analysis of software reliability models. 1993.

96. J.K. Scholma, A Lie algebraic study of some integrable systems associated with root systems. 1993.
97. J.L. van den Berg. Sojourn times in feedback and processor sharing queues. 1993.